Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 1 of 130

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF OHIO EASTERN DIVISION

In re: : Case No. 18-55889

•

Ed Map, Inc., : Chapter 11

:

Debtor. : Judge Hoffman

MOTION OF DEBTOR FOR INTERIM AND FINAL ORDERS AUTHORIZING USE OF CASH COLLATERAL AND PROVIDING ADEQUATE PROTECTION TO KEY BANK

Now comes Ed Map, Inc., the Debtor and Debtor-in-Possession (the "Debtor"), and moves the Court (the "Motion") pursuant to 11 U.S.C. §§361, 362, 363, 364 and 507(b), Rules 2002, 4001 and 9014 of the Federal Rules of Bankruptcy Procedure, and Local Bankruptcy Rule 4001-2 for entry and approval of an Order authorizing the use of cash collateral of Key Bank (the "Bank") and authorizing Debtor further to provide adequate protection to the Bank. As a result of the Debtor seeking an expedited hearing on this matter, no 21-day notice is attached. A Memorandum in Support follows.

Respectfully submitted,

/s/ John W. Kennedy

Myron N. Terlecky (0018628) John W. Kennedy (0042672)

Strip, Hoppers, Leithart, McGrath & Terlecky Co., LPA

575 South Third Street

Columbus, Ohio 43215-5759 Telephone: (614) 228-6345 Facsimile: (614) 228-6369

Email: mnt@columbuslawyer.net
Proposed Attorneys for Ed Map, Inc.,
Debtor and Debtor-in-Possession

MEMORANUM IN SUPPORT

I. BACKGROUND INFORMATION

- 1. The Debtor commenced this case by filing a voluntary petition under Chapter 11 of the Bankruptcy Code on September 17, 2018 (the "Petition Date").
- 2. The Court has jurisdiction over this matter pursuant to 28 U.S.C. §§ 157(b)(1) and 1334. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(A). Venue of this proceeding is proper in this District pursuant to 28 U.S.C. §§ 1408 and 1409 because the Debtor's assets and principal place of business are located in this District.
- 3. Pursuant to 11 U.S.C. §§ 1107(a) and 1108, the Debtor is operating its business and managing its affairs as debtor in possession. No trustee, examiner or statutory committee has been appointed in this case.
- 4. Debtor is a Delaware Corporation licensed to do business in the State of Ohio. The Debtor has been in business since October 2003.
- 5. Debtor is in the business of providing print and digital textbooks primarily to forprofit colleges, while also serving as a content strategy and logistics company. Debtor has proprietary software known as OPENVUE, which is a registered trademark of the Debtor. OPENVUE is a cloud-based platform that students may use to obtain course materials.
- 6. Over the last eight years, for-profit colleges have been closing with some regularity. As a result, Debtor's gross revenue has decreased in the last three years from \$66 million during fiscal year 2015, to \$42 million in fiscal year 2017.
- 7. Debtor has reduced its employees from a high of 80 to 44 in an effort to address its reduced revenue. These efforts resulted in cost-savings that permitted Debtor to have sufficient cash flow to fund operations. Unfortunately, some of its customers have failed to pay for services

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 3 of 130

rendered. Debtor has filed suit to collect these unpaid receivables, seeking to recover \$2,426,991.24, together with interest and attorney fees in one case, and \$675,951.07, together with interest and attorney fees in another case.

- 8. Debtor had been in negotiations to sell its assets so as to continue to provide product to its customers, to preserve jobs, and to pay in full the secured claim and to provide a return to unsecured creditors. These negotiations were unilaterally terminated on September 17, 2018.
- 9. Debtor entered into a Distribution Agreement (the "Distribution Agreement") with Vital Source Technologies, Inc. ("Vital Source") to serve as a distributor of the digital content of Ed Map.
- 10. On September 13, 2018, Vital Source abruptly terminated the Distribution Agreement based on the insolvency of the Debtor, thereby closing off the ability of Debtor to deliver electronic textbooks and related products to its customers. This unilateral action adversely impacted the potential sale of Debtor's assets. In an effort to preserve ongoing value and to seek a sale of its assets, Debtor filed this Chapter 11 proceeding.
- 11. Prior to the Petition Date, the Debtor entered into a line of credit (the "Loan Agreement") with the Bank. As of the Petition Date, Debtor owes approximately \$3.9 million to the Bank, exclusive of fees and other charges.
- 12. As a result of the pre-petition loan with the Bank, all cash in the Debtor's possession or in which the Debtor has an interest on and after the Petition Date constitutes cash collateral (the "Cash Collateral") in which the Bank asserts an interest within the meaning of 11 U.S.C. §363(a).
- 13. By this Motion, pursuant to 11 U.S.C. §§105, 361, 362, 363 and 364, Bankruptcy Rules 2002, 4001 and 9014, and LBR 4001-2, the Debtor requests that this Court enter an order (i) authorizing the Debtor's interim use of Cash Collateral, (ii) approving adequate protection to

the Bank, and (iii) scheduling a final hearing (the "Final Hearing") for this Court to consider entry of a final order authorizing and approving the relief requested in this Motion. LBR Form 4001-2 is attached hereto as Exhibit A. A proposed order authorizing the interim use of Cash Collateral is attached hereto as Exhibit B. The loan documents for the Loan Agreement are attached hereto as Exhibit C.

II. BASIS FOR RELIEF

- 14. The Debtor seeks authority from this Court in accordance with 11 U.S.C. § 363(c)(2) to use the income generated from its operations which is subject to the security interest of the Bank. The Debtor seeks to use cash collateral to make adequate protection payments to the Bank, to pay the obligations owed to the Bank and to pay its ordinary-course post-petition operating expenses.
- 15. The Debtor will suffer immediate and irreparable harm if not authorized to use cash collateral or obtain financing. It is important for the Debtor to be able to use the income generated from its operations which is subject to the security interests of the Bank to continue its operations, maintain its customer base, pay operating obligations, pay adequate protection payments and protect the value of the Debtor's operations.
- 16. The Debtor intends to provide adequate protection to the Bank for the use of the Cash Collateral by offering to maintain the going concern value of the collateral by using the Cash Collateral to continue to operate the business and by providing post-petition replacement liens pursuant to 11 U.S.C. §361(2) in accounts receivable. The priority of any such post-petition replacement liens shall be the same as existed as of the Petition Date.
- 17. The continuation of the Debtor's operations likely presents the best opportunity for the Bank and other creditors to receive the greatest recovery on account of their claims.

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 5 of 130

Accordingly, the use of the Cash Collateral will allow the Debtor to continue operations and thereby protect the creditors' interests. Courts have consistently recognized that the preservation of the value of secured lenders' collateral constitutes adequate protection of such creditors' interest in the collateral. *See*, *In re Pursuit Athletic Footwear*, *Inc.*, 193 B.R. 713, 716 (Bankr. D. Del. 1996) (if there is no actual diminution of value of collateral and the debtor can operate profitably post-petition, then the secured creditor is adequately protected); *In re Willowood E. Apartments of Indianapolis II*, *Ltd.*, 114 B.R. 138, 143 (Bankr. S.D. Ohio 1990) (interest is collateral is adequately protected when cash collateral applied to normal operating and maintenance expenditures on the collateral property).

- 18. Furthermore, in the present case, the value of Debtor's inventory and accounts receivable far exceeds the balance due to the Bank; therefore, the Bank is adequately protected by the equity cushion in its collateral. A substantial equity cushion can obviate the need for any other adequate protection. *In re Mellor*, 734 F.2d 1396 (9th Cir. 1984); *In re Adams*, 218 B.R. 597 (Bankr. D. Kan. 1998); *In re Curtis*, 9 B.R. 110 (Bankr. E.D. Pa. 1981) ("Although the existence of an equity cushion as a method of adequate protection is not specifically delineated in §361, it is the classic form of protection for a secured debt..."); *In re Mr. D Realty Co.*, 27 B.R. 359 (Bankr. S.D. Ohio 1983); *In re Brown*, 78 B.R. 499 (Bankr. S.D. Ohio 1987).
- 19. Concurrently with this Motion, Debtor is requesting an expedited hearing to consider the use of Cash Collateral. An immediate need exists for the Debtor to obtain approval of the use of Cash Collateral in order to meet key expenses such as payroll, rent, utilities, insurance, and taxes. Without the immediate use of Cash Collateral, the Debtor will essentially be forced to shut down, precluding any possibility of preserving the value of the business, generating new value and successfully reorganizing.

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 6 of 130

20. Pursuant to Bankruptcy Rule 4001, the Debtor requests that this Court set a preliminary hearing on the use of Cash Collateral, and that at such preliminary hearing the Court authorize the temporary use of Cash Collateral in order to prevent immediate and irreparable harm to Debtor pending a final hearing. The Debtor further requests a final hearing as soon as the Court's schedule permits.

Based upon the foregoing, it is respectfully requested that the Court enter an Order granting the Motion and that the Court grant such other and further relief as is appropriate.

Respectfully submitted,

/s/ John W. Kennedy

Myron N. Terlecky (0018628)
John W. Kennedy (0042672)
Strip, Hoppers, Leithart, McGrath & Terlecky Co., LPA
575 South Third Street
Columbus, Ohio 43215-5759
Telephone (614) 238 6345

Telephone: (614) 228-6345 Facsimile: (614) 228-6369

Email: mnt@columbuslawyer.net
Proposed Attorneys for Ed Map, Inc.,
Debtor and Debtor-in-Possession

EXHIBIT A

LBR Form 4001–2

Cash Collateral / Postpetition Financing Provisions

The Debtor, through a separately filed motion, agreed order or stipulation has requested the approval of the use of cash collateral, which contains the following provisions:

Page No.	Line No. (If Applica	Description of Provision able)
3-4	□4,5	(A) Provisions providing for adequate protection or priority for a claim that arose before
		the commencement of the case, including:
		(i) Cross-collateralization clauses (i.e., clauses that secure the repayment of prepetition
		debt with postpetition assets in which the secured creditor would not otherwise have a
		security interest by virtue of its prepetition security agreement or applicable law); or
		(ii) Roll-up clauses (i.e., clauses that provide for the use of property of the estate or the
		proceeds of a postpetition loan to make cash payments on prepetition debt).
		(B) Provisions or findings of fact that release, waive, or limit any claim or other cause of
		action belonging to the estate or the trustee, including but not limited to:
		(i) the release, waiver, or limitation of claims or other causes of action against any
		secured creditor without first giving parties in interest at least 75 days from the entry
		of the order and the creditors' committee, if appointed, at least 60 days from the date
		of its appointment to investigate such matters.;
		(ii) the release, waiver, or limitation of claims or other causes of action against any
		secured creditor for alleged prepetition torts or breaches of contract;
		(iii) the waiver of avoidance actions under the Code; or
		(iv) any modification of the statute of limitations or other deadline to commence an
		action.
		(C) Provisions or findings of fact that determine the validity, enforceability, priority, or
		amount of a claim that arose before the commencement of the case, or of any lien securing
		the claim.
		(D) Provisions that grant a lien on property of the estate that is not otherwise subject to a
		lien, grant a junior lien on property of the estate that is subject to a lien, or create a lien
		senior or equal to any existing lien without the consent of that lienholder.
		(E) Provisions that release, waive, or limit any right under § 506(c) of the Code.

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 8 of 130

	(F) Provisions that grant a lien on any claim or causes of action arising under §§ 544, 545, 547, 548, 549, 553(b), 723(a), or 724(a) of the Code.
	(G) Provisions that provide disparate treatment with regard to professional fee carveouts for the professionals retained by a creditors' committee from that provided for the professionals retained by the debtor.
	(H) Provisions that prime administrative expenses of the kind specified in §§ 503(b) or 507(a) of the Code.
	(I) Provisions that waive or modify any entity's authority or right to file a plan or seek an extension of time in which the debtor has the exclusive right to file a plan or otherwise operate to divest the debtor of any discretion in the administration of the estate or limit access to the court to seek any relief under other applicable provisions of law.
	(J) Provisions that establish deadlines for filing a plan of reorganization, for approval of a disclosure statement, for a hearing on confirmation, or entry of a confirmation order.
	(K) Provisions providing for the indemnification of any entity.
	(L) Provisions waiving or modifying provisions of the Code or applicable rules relating to the automatic stay.
	(M) Provisions that waive or modify the applicability of nonbankruptcy law relating to the perfection of a lien on property of the estate, or on the foreclosure or other enforcement of the lien.
	(N) Provisions that waive or modify the debtor's right to move for a court order pursuant to § 363(c)(2)(B) of the Code authorizing the use of cash collateral or § 364 of the Code to obtain credit.
	(O) Provisions that grant a lien in an amount in excess of the dollar amount of cash collateral authorized under the applicable cash collateral order.
□	(P) Findings of fact on matters extraneous to the approval process.
	(Q) Provisions that bar the debtor from future bankruptcy filings.

Exhibit B

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF OHIO EASTERN DIVISION

In re: : Case No. 18-55889

.

Ed Map, Inc. : Chapter 11

:

Debtor. : Judge

<u>INTERIM ORDER AUTHORIZING USE OF CASH COLLATERAL AND,</u> PROVIDING ADEQUATE PROTECTION TO SECURED CREDITOR (DOC. --)

This cause came before the Court on an expedited hearing on September ____, 2018 to consider the *Motion of Debtor for Interim and Final Orders (1) Authorizing Use of Cash Collateral*, (2) *Providing Adequate Protection to Secured Creditor*, and (3) *Authorizing Secured Post-Petition Financing with Key Bank* (Doc. ___) (the "Motion") filed by Ed Map, Inc., the Debtor and Debtor-in-Possession (the "Debtor") pursuant to 11 U.S.C. §§361, 362, 363, 364 and 507(b), Rules 2002, 4001 and 9014 of the Federal Rules of Bankruptcy Procedure, and Local Bankruptcy Rule 4001-2. Upon the consideration of the testimony and presentation at the expedited hearing, and after due deliberation and sufficient cause appearing therefor,

THE COURT FINDS THAT:

- A. The Debtor filed a petition for relief under Chapter 11 of the United States Bankruptcy Code on September 17, 2018 (the "Petition Date").
- B. Pursuant to 11 U.S.C. §§ 1107(a) and 1108, the Debtor is operating its business and managing its affairs as debtor in possession. No trustee, examiner or statutory committee has been appointed in this case.
- C. The Court has jurisdiction over this matter pursuant to 28 U.S.C. §§ 157(b)(1) and 1334. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(A). Venue of this proceeding is proper in this District pursuant to 28 U.S.C. §§ 1408 and 1409 because the Debtor's assets and principal place of business are located in this District.
- D. The form of the Motion meets the requirements of Fed. R. Bankr. P. 4001(b)(1) and LBR 4001-2. Service thereof has been made in accordance with Fed. R. Bank. P. 4001(b)(1)(C) and LBR 4001-2.
- E. Prior to the Petition Date, the Debtor entered into a loan agreement with KeyBank, N.A. (the "Bank"), secured by a lien (the "Prepetition Lien") on substantially all of Debtor's assets (the "Prepetition Collateral"), including deposit accounts, accounts receivable and the products and proceeds of the Prepetition Collateral (the "Cash Collateral") as more particularly set forth in the Motion.
- F. The Bank asserts that the Prepetition Lien is a valid, perfected and non-avoidable first priority lien on the Prepetition Collateral.
- G. The Debtor has not yet completed its review of the Prepetition Lien, or the value of the Prepetition Collateral, and the Court has not been asked to find, and it does not find, that the Prepetition Lien is valid, perfected or non-avoidable.

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 11 of 130

H. The Court has reviewed the Motion and has considered the representations of the parties regarding the relief requested therein. The relief requested in the Motion and granted hereby is in the best interest of the Debtor's bankruptcy estate, its creditors and other parties in interest.

NOW, THEREFORE, IT IS ORDERED, ADJUDGED AND DECREED:

- 1. <u>Granting of Motion</u>. The Motion is granted on an interim basis as set forth herein. Any and all objections to the Motion are hereby overruled.
- 2. <u>Authorization to Use Cash Collateral</u>. The Debtor is authorized to use Cash Collateral in the ordinary course of the Debtor's business and solely for the purposes set forth in the Budget annexed hereto as Exhibit A (the "Budget") during the period from entry of this Order through and including the earlier of November 30, 2018 and the date of occurrence of an Event of Default as defined below (the "End Date").
- Cash Collateral solely at the times and in the amounts specified in the Budget, provided, however, that (a) the total funds expended by the Debtor may not at any time exceed by more than 10% the cumulative aggregate level of expenses authorized in the Budget through the end of the week, (b) the Debtor may apply any unused portion in one week to any subsequent weekly period; and (c) the Debtor may amend the Budget with the written consent of the Bank (which the Bank may give or withhold in its sole discretion), whereupon such amended budget shall be the "Budget" for purposes of this Order.
- 4. <u>Replacement Lien</u>. As adequate protection for any post-petition diminution in the value of the Prepetition Collateral from the Petition Date to the End Date (as such End Date may be extended by agreement of the parties or further order of the Court) (in the aggregate, the

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 12 of 130

"Diminution Amount"), the Bank is hereby granted a post-petition replacement lien (the "Post-petition Lien") in and to all property presently securing the Claim, together with any post-petition proceeds thereof. The Post-petition Lien shall attach to and be enforceable against the same property and any post-petition proceeds thereof, to the same extent, and in the same order of priority as the Prepetition Lien. For the avoidance of doubt, (a) the Post-petition Lien shall not attach to causes of action or rights of recovery under chapter 5 of the Bankruptcy Code, or the proceeds of such claims, and (b) the Bank may not foreclose or exercise any other remedy in respect of the Prepetition or Post-petition Liens without further order of this Court.

- 5. <u>Monthly Payment of Interest.</u> As further adequate protection, the Debtor shall pay the Bank interest only payments in the approximate amount of \$25,000.00 (the "Monthly Payment") on or before the 5th day of each month. The Bank shall apply each Monthly Payment in accordance with the loan documents on which the Claim is based, provided, however, that if the Bank's claim is ultimately allowed as a secured claim (hereinafter, the "Allowed Secured Claim") in an amount less than the full amount of the Claim, as determined by final order of this Court, the Monthly Payments shall be reallocated so as to be applied solely to the Allowed Secured Claim, and the Bank shall refund to the Debtor the amount (if any) by which the total Monthly Payments exceed the Allowed Secured Claim.
- 6. Reservation of Rights (Future Use of Cash Collateral). This Order shall be without prejudice to the right of the Debtor to seek, and the right of the Bank or any other party in interest to oppose, the Debtor's use of Cash Collateral beyond the End Date.
- 7. Reservation of Rights (Challenge to Bank's Claim and Lien). Subject to the following paragraph, this Order shall be without prejudice to the right of the Debtor or any other party in interest to object to the Claim or to seek avoidance of the Prepetition Lien pursuant to

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 13 of 130

Chapter 5 of the Bankruptcy Code or on any other basis whatsoever (the "Objection Rights"). The Post-petition Liens may be revoked or reduced to the extent that any such avoidance, challenge or objection is successful.

- 8. Deadline to Challenge Bank's Claim or Lien. Any exercise of Objection Rights by the Debtor or any other party in interest shall be filed with this Court not later than 75 days from the entry of this Order or 60 days from the appointment of a creditors' committee, whichever is later, and if not timely filed, TIME BEING OF THE ESSENCE FOR THIS PURPOSE, any such exercise of Objection Rights shall be forever barred notwithstanding any subsequent events in this case, including but not limited to conversion, dismissal, approval or disapproval of such sale, occurrence or non-occurrence of the closing thereof, or confirmation of a plan. Notwithstanding the foregoing, the Debtor's rights concerning the proper classification or treatment of the Claim, the value of the PrePetition Collateral, and (if the Claim is fully secured) the correct amount of postpetition interest, fees, costs or charges to be included in the Allowed Secured Claim pursuant to section 506(b) of the Bankruptcy Code, shall be preserved without the need for timely exercise of Objection Rights.
- 9. <u>Financial Reports and Information</u>. The Debtor shall supply the following reports and information to the Bank:
 - a. At the end of every two-week period, the Debtor shall cause to be served on the Bank a report, in a format reasonably acceptable to the Bank, comparing the Debtor's actual cash receipts and disbursements through the end of the previous week to the receipts and disbursements forecast by the Budget. As soon as feasible, the Debtor shall supply the Bank with a report on which receipts and disbursements since the Petition Date are attributable to the Cash Collateral,

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 14 of 130

and thereafter the bi-weekly reports supplied by the Debtor shall break out receipts constituting Cash Collateral and disbursements of the Debtor attributable to the Cash Collateral.

- b. At the same time the Debtor files a financial report with the United States Trustee, the Debtor shall supply a copy of such report or information to the Bank.
- c. If the Bank requests financial or other information from the Debtor, the Debtor shall supply such information, to the extent the request therefor is reasonable, within a reasonable time.
- 10. <u>Default</u>. The Debtor's authority to use Cash Collateral shall terminate upon the occurrence of any of the following events, unless waived by the Bank in writing (each, an "Event of Default"):
 - a. the Debtor's failure to perform any of his obligations under this Order, provided, however, that on no more than one occasion during the term of this Order, the Debtor shall have a three-day grace period for delivery to the Bank of any report required by the preceding paragraph of this Order.
 - b. modification (without the express written consent of the Bank in its sole discretion), reversal or vacatur of this Order; or
 - c. dismissal of the case, conversion of the case to a chapter 7 case, or the appointment of a Chapter 11 Trustee.
- 11. <u>Final Hearing</u>. The Court shall hold a final hearing on the Motion and the Debtor's use of Cash Collateral on _______, 2018 at _____a.m./p.m. (the "Final Hearing"). Any objections to the further use of Cash Collateral shall be filed with this Court and served upon the

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 15 of 130

Debtor, the United States Trustee, and the Bank by	_, 2018 (the "Objection
Deadline"). The Debtor shall serve notice of the entry of this Order, the	e Final Hearing, and the
Objection Deadline on all persons entitled to receive notice under Fed. R	. Bankr. P. 4001(b)(3).
IT IS SO ORDERED.	

Submitted by:

/s/ Myron N. Terlecky

Myron N. Terlecky (0018628)
John W. Kennedy (0042672)
Strip, Hoppers, Leithart, McGrath & Terlecky Co., LPA
575 South Third Street
Columbus, Ohio 43215-5759
Telephone (614) 228-6345

Telephone: (614) 228-6345 Facsimile: (614) 228-6369

Email: mnt@columbuslawyer.net
Proposed Attorneys for Ed Map, Inc.,
Debtor and Debtor-in-Possession

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 16 of 130

Exhibit C - Loan Agreement

CREDIT AND SECURITY AGREEMENT

among

KEYBANK NATIONAL ASSOCIATION (as Lender and Issuer)

and

ED MAP, INC. (as a Borrower)

November 22, 2016

TABLE OF CONTENTS

	DEFINITIONS	
1.1	Accounting Terms.	
1.2	General Terms.	
1.3	Uniform Commercial Code Terms.	
1.4	General Matters of Construction.	
1.5	Time References.	23
ARTICLE 2	ADVANCES, PAYMENTS	23
2.1	Revolving Loans.	23
2.2	Procedure for Borrowing Advances.	
2.3	Disbursement of Loan Proceeds.	
2.4	Maximum Advances.	25
2.5	Repayment of Loans.	25
2.6	Statement of Account.	
2.7	Letters of Credit.	
2.8	Issuance of Letters of Credit.	
2.9	Requirements For Issuance of Letters of Credit.	
2.10	Reserved	
2.11	Additional Payments.	
2.12	Use of Proceeds	27
ARTICLE 3	INTEREST AND FEES	27
3.1	Interest	
3.2	Letter of Credit Fees.	
3.3	Unused Facility Fee.	
3.4	Administration Fee.	
3.5	Upfront Fee.	
3.6	Computation of Interest and Fees.	
3.7	Maximum Charges.	29
3.8	Increased Costs.	29
3.9	Basis For Determining Interest Rate Inadequate or Unfair; Indemnity;	
	Libor Rate Unlawful.	
3.10	Capital Adequacy	31
ARTICLE 4	COLLATERAL: GENERAL TERMS	31
4.1	Security Interest in the Collateral	
4.2	Perfection of Security Interest.	
4.3	Disposition of Collateral.	
4.4	Preservation of Collateral	
4.5	Ownership of Collateral	
4.6	Defense of the Lender's Interests	
4.7	Books and Records.	33
4.8	Financial Disclosure	34
4.9	Compliance with Laws	34
4.10	Inspection of Premises; Appraisals.	34
4.11	Insurance.	34
4.12	Failure to Pay Insurance	35
4.13	Payment of Taxes.	
4.14	Payment of Leasehold Obligations.	
4.15	Accounts.	36
4.16	Maintenance of Equipment.	
4.17	Exculpation of Liability.	
4.18	Reseved	38

i

4	4.19	Financing Statements.	38
4	4.20	Pledged Securities.	
4	4.21	Cash Management System.	
ARTIC	LE 5 1	REPRESENTATIONS AND WARRANTIES	4 1
	5.1	Authority	
	5.2	Formation and Qualification; Subsidiaries	41
	5.3	Officers, Directors, Shareholders, Capitalization.	
	5.4	Governmental Approvals; No Conflicts.	
	5.5	Tax Returns.	
	5.6	Financial Statements.	
	5.7	Corporate Name.	
	5.8	O.S.H.A. and Environmental Compliance.	
	5.9	Solvency; No Litigation, No Violation, ERISA	
	5.10	Patents, Trademarks, Copyrights and Licenses	
	5.11	Licenses and Permits.	
	5.12	Default of Indebtedness.	
	5.13	No Burdensome Restrictions; No Default.	
	5.14	No Labor Disputes.	
	5.15	Margin Regulations.	
	5.16	Investment Company Act.	
	5.17	Disclosure.	
	5.18	Hedging Contracts.	
	5.19	Material Business Agreements	
	5.20	Application of Certain Laws and Regulations.	
	5.21	Anti-Terrorism Laws	
	5.22	Anti-Corruption Laws and Sanctions.	46
		_	
		AFFIRMATIVE COVENANTS	
	6.1	Conduct of Business and Maintenance of Existence and Assets.	
	6.2	Violations.	
	6.3	Fixed Charge Coverage Ratio.	ر 4
	6.4	Execution of Supplemental Instruments.	
	6.5	Payment of Indebtedness.	
	6.6	Standards of Financial Statements.	
	6.7	Taxes	
	6.8	Deposit Accounts.	
(6.9	Interest Rate Protection.	48
ARTIC	LE 7	NEGATIVE COVENANTS	
,	7.1	Merger, Consolidation, Acquisition and Sale of Assets.	
,	7.2	Creation of Liens	
,	7.3	Guarantees	
,	7.4	Investments.	
,	7.5	Loans	
,	7.6	Capital Expenditures.	49
,	7.7	Dividends and Distributions	49
,	7.8	Indebtedness	
,	7.9	Nature of Business.	
,	7.10	Transactions with Affiliates.	50
,	7.11	Leases	50
,	7.12	Subsidiaries; Partnerships.	
,	7.13	Management Fees.	
,	7.14	Fiscal Year and Accounting Changes.	
,	7.15	Pledge of Credit.	

7.16	Amendment of Charter Documents.	
7.17	ERISA.	50
7.18	Prepayment of Indebtedness.	
7.19	Modification of Material Business Agreements.	
7.20	Anti-Terrorism Laws	51
ARTICLE 8	CONDITIONS PRECEDENT	51
8.1	Conditions to Initial Loans.	
8.2	Conditions to Each Advance.	52
ARTICLE 9	INFORMATION AS TO THE LOAN PARTIES	54
9.1	Disclosure of Material Matters.	52
9.2	Collateral Reporting and Information.	55
9.3	Litigation	
9.4	Material Occurrences.	5 6
9.5	Annual Financial Statements.	
9.6	Monthly Financial Statements.	57
9.7	Additional Information.	57
9.8	Projected Operating Budget, Availability Forecast.	57
9.9	Notice of Suits, Adverse Events.	58
ARTICLE 1	0 EVENTS OF DEFAULT	58
10.1	Payment of Obligations	58
10.2	Misrepresentations.	
10.3	Failure to Furnish Information.	
10.4	Liens Against Assets.	
10.5	Breach of Covenants.	
10.6	Judgment.	
10.7	Insolvency and Related Proceedings	
10.8	Material Adverse Effect	
10.9	Loss of Priority Lien.	
10.10	$\boldsymbol{\mathcal{C}}$	59
10.11		
10.12		
10.13		
10.14		
10.15 10.16		
10.16		
	1	
	1 LENDER'S RIGHTS AND REMEDIES AFTER DEFAULT	
11.1	Rights and Remedies.	
11.2	Lender Discretion	
11.3	Setoff	
11.4 11.5	Rights and Remedies not Exclusive.	
11.5	Appointment of Receiver. Shareholder Cure. Error! Bookm	
	2 WAIVERS AND JUDICIAL PROCEEDINGS	
12.1	Waiver of Notice.	
12.2	Delay	
12.3	Jury Waiver	
	3 EFFECTIVE DATE AND TERMINATION	
13.1	Term	
13.2	Termination	62

{6369722:7} iii

ARTICLE 14 THE BORROWER REPRESENTATIVE62		
62		
63		
64		
64		
64		
65		
65		
65		
65		
66		
66		
67		
68		
68		
68		
68		
69		
69		
69		
69		
69		
70		

{6369722:7} iv

LIST OF SCHEDULES AND EXHIBITS

All Schedules to the Credit and Security Agreement:

Schedule 1.2(a)	Owned Real Property
Schedule 1.2(b)	Liens
Schedule 4.5	Inventory
Schedule 4.15(c)	Loan Parties' States of Organization and Chief Executive Offices
Schedule 5.2(a)	Incorporation/Organization/Foreign Qualification
Schedule 5.2(b)	Subsidiaries
Schedule 5.3	Officers, Directors, Shareholders, Capitalization
Schedule 5.9(b)	Litigation
Schedule 5.10	Patents, Trademarks, Copyrights and Licenses
Schedule 5.19	Material Business Agreements
Schedule 6.8	Accounts
Schedule 7.4	Investments
Schedule 7.5	Loans
Schedule 7.8	Indebtedness

All Exhibits to the Credit and Security Agreement:

Exhibit A Form of Compliance Certificate Exhibit B Form of Revolving Note

Exhibit C Form of Notice of Loan

CREDIT AND SECURITY AGREEMENT

This **CREDIT AND SECURITY AGREEMENT** (this "<u>Agreement</u>"), has been executed and is dated as of November 22, 2016, by and among **ED MAP, INC.**, as a Borrower, and **KEYBANK NATIONAL ASSOCIATION**, as the Lender and the Issuer.

IN CONSIDERATION of the mutual covenants and undertakings herein contained, the receipt and sufficiency of which are hereby acknowledged, the Borrowers, the Lender and the Issuer hereby agree as follows:

ARTICLE 1 DEFINITIONS

- Accounting Terms. As used in the Loan Documents, accounting terms not defined in Section 1.2 or elsewhere in this Agreement and accounting terms partly defined in Section 1.2 to the extent not defined shall have the respective meanings given to them under GAAP; provided, however, whenever such accounting terms are used for the purposes of determining compliance with financial covenants in this Agreement, such accounting terms shall be defined in accordance with GAAP. All financial computations to be made under this Agreement shall, unless otherwise specifically provided herein, be made in accordance with GAAP applied on a basis consistent in all material respects with the financial statements delivered to the Lender on or prior to the Closing Date. Notwithstanding any other provision contained herein, all terms of an accounting or financial nature used herein shall be construed, and all computations of amounts and ratios referred to herein shall be made without giving effect to any change in GAAP occurring after the Closing Date as a result of the adoption of any proposals set forth in the Proposed Accounting Standards Update, Leases (Topic 840), issued by the Financial Accounting Standards Board on August 17, 2010, or any other proposals issued by the Financial Accounting Standards Board in connection therewith, in each case if such change would require treating any lease (or similar arrangement conveying the right to use) as a capital lease where such lease (or similar arrangement) was not required to be so treated under GAAP as in effect on the Closing Date.
- 1.2 <u>General Terms</u>. For purposes of this Agreement, the following terms shall have the following meanings:
 - "Accommodation Payment" shall have the meaning set forth in Section 14.2.
- "Acquisition" shall mean any transaction or series of related transactions for the purpose of or resulting, directly or indirectly, in (a) the acquisition of all or substantially all of the assets of a Person, or of all or substantially all of any business or division of a Person, (b) the acquisition of in excess of 50% of the Equity Interests of any Person, or otherwise causing any Person to become a Subsidiary, or (c) a merger or consolidation or any other combination with another Person (other than a Person that is already a Subsidiary).
 - "Advances" shall mean and include the Revolving Loans and Letters of Credit.
- "Affiliate" of any Person shall mean (a) any Person which, directly or indirectly, is in control of, is controlled by, or is under common control with such Person, or (b) any Person who is a director or officer (i) of such Person, (ii) of any Subsidiary of such Person or (iii) of any Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the power, directly or indirectly, (x) to vote fifteen percent (15%) or more of the securities having ordinary voting power for the election of directors of such Person, or (y) to direct or cause the direction of the management and policies of such Person whether by contract or otherwise.

- "Agreement" shall have the meaning set forth in the preamble.
- "Allocable Amount" shall have the meaning set forth in Section 14.2.
- "<u>Anti-Corruption Laws</u>" shall mean all laws, rules, and regulations of any jurisdiction applicable to the Borrower or any of its Subsidiaries from time to time concerning or relating to bribery or corruption.
- "Anti-Terrorism Laws" shall mean any laws relating to terrorism or money laundering, including Executive Order No. 13224, the USA Patriot Act, the laws comprising or implementing the Bank Secrecy Act, and the laws administered by the United States Treasury Department's Office of Foreign Asset Control.
 - "Applicable Base Rate Margin" shall mean 1.00%.
 - "Applicable Letter of Credit Fee Percentage" shall mean 3.00%.
 - "Applicable LIBOR Rate Margin" shall mean 3.00%.
 - "Applicable Unused Facility Fee Percentage" shall mean 0.375%.
- "Approved Electronic Communication System" shall mean the StuckyNet System or any other equivalent electronic service, whether owned, operated or hosted by Lender, any affiliate of Lender or any other Person.
- "<u>Authorized Officer</u>" shall mean a Financial Officer or other individual authorized by a Financial Officer in writing (with a copy to Lender), or any other officer approved by the Lender in its Permitted Discretion, in each case to handle certain administrative matters in connection with this Agreement.
- "Availability Block" shall mean, at any time that (a) the trailing twelve-12 month EBITDA does not exceed \$1,000,000, as determined by the Lender in its Permitted Discretion, and (b) an Advance is outstanding, an amount equal to the aggregate amount previously paid by the Borrowers in respect of Permitted Acquisitions and Permitted Stock Repurchases up to an aggregate amount equal to \$1,250,000.
- "Base Rate" shall mean, for any day, a fluctuating interest rate per annum equal to the higher of: (a) the rate of interest which is established from time to time by the Lender at its principal office in Cleveland, Ohio as its "prime rate" or "base rate" in effect, such rate to be adjusted automatically, without notice, as of the opening of business on the effective date of any change in such rate (it being agreed that: (i) such rate is not necessarily the lowest rate of interest then available from the Lender on fluctuating rate loans and (ii) such rate may be established by the Lender by public announcement or otherwise); (b) the Federal Funds Effective Rate in effect on such day <u>plus</u> 0.50%; and (c) the Libor Rate for one (1) month interest periods <u>plus</u> 1.00%.
 - "Base Rate Loan" shall mean any Loan that bears interest based upon the Base Rate.
 - "Blocked Person" shall have the meaning assigned to such term in Section 5.21(b).
- "<u>Borrower</u>" and "<u>Borrowers</u>" shall mean, individually or collectively, as the context may require, Ed Map, Inc., a Delaware corporation, and any other Person who may hereafter become a party hereto.
 - "Borrowing Base" shall mean, at any time, the sum of:
 - (a) up to 85% of each Borrower's Eligible Accounts at such time, *plus*

- (b) the least of (i) up to 50% of each Borrower's Eligible Inventory, (ii) the product of 85% <u>multiplied by</u> the Net Orderly Liquidation Value <u>multiplied by</u> each Borrower's Eligible Inventory, in each case, valued at the lower of cost or market, determined on a first in first out basis, at such time, and (iii) \$5,000,000, minus
 - (c) the Availability Block, *minus*
 - (d) Reserves.

The Lender may, in its Permitted Discretion, reduce the advance rates set forth above, adjust Reserves or reduce one or more of the other elements used in computing the Borrowing Base; provided, however, the Lender shall endeavor to provide notice to the Borrowers of any such Reserve or reduction, but the failure to give such notice shall not affect its ability to implement any such Reserve or reduction.

"Borrowing Base Certificate" shall mean a certificate duly executed by a Financial Officer of the Borrower Representative appropriately completed and in form and substance satisfactory to the Lender.

"Borrower Representative" shall mean Ed Map, Inc.

"Business Day" shall mean any day other than Saturday or Sunday or a legal holiday on which commercial banks are authorized or required by law to be closed for business in Cleveland, Ohio and, if the applicable Business Day relates to any Libor Rate Loans, such day must also be a day on which dealings in deposits are carried on in the London interbank market.

"Capital Expenditures" shall mean any expenditure made or liability incurred which is, determined in accordance with GAAP, treated as a capital expenditure and not as an expense item for the year in which it was made or incurred, as the case may be.

"Cash Concentration Account" shall mean, with respect to the Borrowers, that certain commercial deposit account maintained at KeyBank National Association, the funds within which: (a) shall be the sole and exclusive property of the Lender and (b) the Lender shall have the irrevocable and exclusive right to withdraw until all of the Obligations are paid, performed, satisfied and enforced in full and the commitments of the Lender to make Advances hereunder and all Letters of Credit have terminated.

"<u>CERCLA</u>" shall mean the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, 42 U.S.C. Sections 9601 et seq.

"Change of Control" shall mean (a) the Permitted Holders shall cease to own, free and clear of all Liens or other encumbrances, directly or indirectly, 50% of the outstanding voting Equity Interests of each Borrower on a fully diluted basis; (b) any merger or consolidation of or with any or sale of all or substantially all of the property or assets of any Loan Party (other than any Loan Party into a Borrower); or (c) each Borrower shall cease to own, free and clear of all Liens or other encumbrances, at least 100% of the outstanding voting Equity Interests of any existing or future Subsidiary.

"Charges" shall mean all taxes, charges, fees, imposts, levies or other assessments, including all net income, gross income, gross receipts, sales, use, ad valorem, value added, transfer, franchise, profits, inventory, capital stock, license, withholding, payroll, employment, social security, unemployment, excise, severance, stamp, occupation and property taxes, custom duties, fees, assessments, liens, claims and charges, together with any interest and any penalties, additions to tax or additional amounts, imposed by any taxing or other similar Governmental Body, domestic or foreign (including the Pension Benefit Guaranty Corporation or any environmental agency or superfund), upon the Collateral or any Loan Party.

"Charter Documents" shall mean, as to any Person (other than a natural person), the charter, certificate or articles of incorporation or organization, by-laws, regulations, general or limited partnership agreement, certificate of limited partnership, certificate of formation, operating agreement, and other similar organizational or governing documents of such Person.

"Closing Date" shall mean November 22, 2016.

"Code" shall mean the Internal Revenue Code of 1986 and the regulations promulgated thereunder.

"Collateral" shall mean and include all real and personal property other than any Excluded Asset owned by the Loan Parties, whether now owned or existing, or hereafter arising or acquired or received by the Loan Parties, wherever located, including:

- (a) all Accounts;
- (b) all Inventory;
- (c) all Equipment and Fixtures;
- (d) all General Intangibles, Payment Intangibles and Intellectual Property;
- (e) all Investment Property;
- (f) all Deposit Accounts and any and all monies credited by or due from any financial institution or any other depository;
 - (g) all Chattel Paper, Instruments and Documents;
- all of the Loan Parties' right, title and interest in and to (i) its respective goods and other personal property including all merchandise returned or rejected by Account Debtors, relating to or securing any of the Accounts; (ii) all of the Loan Parties' rights as a consignor, a consignee, an unpaid vendor, mechanic, artisan, or other lien or, including stoppage in transit, setoff, detinue, replevin, reclamation and repurchase; (iii) all additional amounts due to the Loan Parties from any Account Debtors relating to the Accounts; (iv) other property, including warranty claims, relating to any goods securing this Agreement; (v) all of the Loan Parties' contract rights, rights of payment which have been earned under a contract right, Instruments (including promissory notes), Documents, Chattel Paper (including electronic chattel paper), warehouse receipts, Deposit Accounts, letters of credit, and money; (vi) all Commercial Tort Claims (whether now existing or hereafter arising); (vii) if and when obtained by the Loan Parties, all real and personal property of third parties in which the Loan Parties have been granted a Lien or security interest as security for the payment or enforcement of Accounts; (viii) all Letter of Credit Rights (whether or not the respective letter of credit is evidenced by a writing); (ix) all Supporting Obligations; and (x) any other goods or personal property, if any, in which the Loan Parties may hereafter in writing grant a security interest to the Lender hereunder, or in any amendment or supplement hereto or thereto, or under any other agreement between the Lender and the Loan Parties;
- (i) all of the Loan Parties' ledger sheets, ledger cards, files, correspondence, records, books of account, business papers, computer software (owned by the Loan Parties or in which they have an interest), computer programs, tapes, disks and documents relating to (a), (b), (c), (d), (e), (f), (g) or (h) of this Paragraph; and

- (j) all proceeds and products of (a), (b), (c), (d), (e), (f), (g) and (h) in whatever form, including: cash, deposit accounts (whether or not comprised solely of proceeds), certificates of deposit, insurance proceeds (including hazard, flood and credit insurance), negotiable instruments and other instruments for the payment of money, chattel paper, security agreements, documents, eminent domain proceeds, condemnation proceeds and tort claim proceeds.
- "Commodity Exchange Act" shall mean the Commodity Exchange Act (7 U.S.C. § 1 et seq.), as amended from time to time, and any successor statute.
- "Compliance Certificate" shall mean a certificate of the Loan Parties signed by a Financial Officer of the Borrower Representative on behalf of each Loan Party appropriately completed and in substantially the form of $\underline{\text{Exhibit A}}$ hereto.
- "Consents" shall mean all filings and all licenses, permits, consents, approvals, authorizations, qualifications and orders of Governmental Bodies and other third parties, domestic or foreign, necessary to carry on any Loan Party's business, including any Consents required under all applicable federal, state or other applicable law.
- "Controlled Disbursement Account" shall mean a commercial Deposit Account designated "controlled disbursement account" and maintained by one or more of the Loan Parties with the Lender, without liability by Lender to pay interest thereon.
- "Controlled Group" shall mean all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with any Person, are treated as a single employer under Section 414 of the Code.
- "<u>Default</u>" shall mean an event which, with the giving of notice or passage of time or both, would constitute an Event of Default.
- "<u>Dodd-Frank Act</u>" shall mean the Dodd-Frank Wall Street Reform and Consumer Protection Act (Pub.L. 111-203, H.R. 4173) signed into law on July 21, 2010, as amended from time to time.
 - "Dollar" and the sign "\sums" shall mean lawful money of the United States of America.
 - "Domestic Subsidiary" shall mean a Subsidiary that is not a Foreign Subsidiary.
- "Earnings Before Interest and Taxes" shall mean, for any fiscal period, the sum of (a) net income (or loss) for such period calculated based on a valuation of Inventory using first-in first-out (FIFO) basis and valued at the lower of cost or market in accordance with GAAP (excluding other non-cash extraordinary and/or non-recurring expenses/income approved by the Lender), <u>plus</u> (b) all interest expense for such period <u>plus</u> (c) all charges against (or <u>minus</u> credits to) income for federal, state and local taxes for such period, in each case, calculated on a consolidated basis for the Borrowers and their Subsidiaries.
- "EBITDA" shall mean, for any fiscal period, the sum of (a) Earnings Before Interest and Taxes for such period, <u>plus</u> (b) depreciation expenses for such period, <u>plus</u> (c) amortization expenses for such period, in each case, calculated on a consolidated basis for the Borrowers and their Subsidiaries.
- "Eligible Accounts" shall mean, at any time, Accounts of the Borrowers that the Lender determines in its Permitted Discretion are eligible as the basis for the extension of Revolving Loans and the issuance of Letters of Credit. Without limiting the Lender's discretion provided herein, Eligible Accounts shall not include any Account:

- (a) which is not subject to a first priority perfected security interest in favor of the Lender;
- (b) which is subject to any Lien other than a Permitted Encumbrance that does not have priority over the Lien in favor of the Lender;
- (c) (i) which is unpaid more than 90 (120 with respect to Educational Corporation of America and Herzing Corporation) days after the date of the original invoice therefor, (ii) which is unpaid more than 60 days after the original due date, or (iii) which has been written off the books of the Borrowers or otherwise designated as uncollectible;
- (d) which is owing by an Account Debtor if more than 50% of the Accounts owing from such Account Debtor and its Affiliates are ineligible;
- (e) with respect to which any covenant, representation, or warranty contained in this Agreement has been breached or is not true in all material respects;
- (f) which (i) does not arise from the sale of goods or performance of services in the ordinary course of business, (ii) is not evidenced by an invoice or other documentation reasonably satisfactory to the Lender which has been sent to the Account Debtor, (iii) represents a progress billing, (iv) is contingent upon a Borrower's completion of any further performance, (v) represents a sale on a bill-and-hold, guaranteed sale, sale-and-return, sale on approval, consignment, cash-on-delivery or any other repurchase or return basis, or (vi) relates to payments of interest;
- (g) for which the goods giving rise to such Account have not been shipped to the Account Debtor or for which the services giving rise to such Account have not been performed by a Borrower or if such Account was invoiced more than once;
- (h) with respect to which any check or other instrument of payment has been returned uncollected for any reason;
- (i) which is owed by an Account Debtor which has (i) applied for, suffered, or consented to the appointment of any receiver, custodian, trustee, or liquidator of its assets, (ii) has had possession of all or a material part of its property taken by any receiver, custodian, trustee or liquidator, (iii) filed, or had filed against it, any request or petition for liquidation, reorganization, arrangement, adjustment of debts, adjudication as bankrupt, winding-up, or voluntary or involuntary case under any state or federal bankruptcy laws, (iv) has admitted in writing its inability, or is generally unable to, pay its debts as they become due, (v) become insolvent, or (vi) ceased operation of its business;
- (j) which is owed by any Account Debtor which has sold all or substantially all of its assets;
- (k) which is owed by an Account Debtor which (i) does not maintain its chief executive office in the U.S. or (ii) is not organized under applicable law of the U.S. or any state of the U.S.;
 - (l) which is owed in any currency other than U.S. dollars;
- (m) which is owed by (i) any Federal, state or local government (or any department, agency, public corporation, or instrumentality thereof) unless such Account is backed by a Letter

of Credit acceptable to the Lender which is in the possession of the Lender, or (ii) with respect to the government of the U.S., or any department, agency, public corporation, or instrumentality thereof, the Federal Assignment of Claims Act of 1940, as amended (31 U.S.C. § 3727 et seq. and 41 U.S.C. § 15 et seq.), and any other steps necessary to perfect the Lien of the Lender in such Account have been complied with to the Lender's satisfaction;

- (n) which is owed by any Affiliate, employee, officer, director, agent or stockholder of any Loan Party;
- (o) which, for any Account Debtor, the aggregate amount of Accounts owed to such Account Debtor exceeds 25% (30% with respect of Delta Career Education and Education Corporation of America) of all Eligible Accounts to the extent of such excess;
- (p) which is owed by an Account Debtor or any Affiliate of such Account Debtor to which any Loan Party is indebted, or is subject to any security, deposit, progress payment, retainage or other similar advance made by or for the benefit of an Account Debtor;
- (q) which is subject to any asserted counterclaim, deduction, defense, setoff or dispute;
 - (r) which is evidenced by any promissory note, chattel paper, or instrument;
- (s) which is owed by an Account Debtor located in any jurisdiction which requires filing of a "Notice of Business Activities Report" or other similar report in order to permit a Borrower to seek judicial enforcement in such jurisdiction of payment of such Account, unless such Borrower has filed such report or qualified to do business in such jurisdiction within 45 days after the Closing Date;
- (t) with respect to which a Borrower has made any agreement with the Account Debtor for any reduction thereof, other than discounts and adjustments given in the ordinary course of business, or any Account which was partially paid and such Borrower created a new receivable for the unpaid portion of such Account;
- (u) which does not comply in all material respects with the requirements of all applicable laws and regulations, whether Federal, state or local, including the Federal Consumer Credit Protection Act, the Federal Truth in Lending Act and Regulation Z of the Board of Governors of the Federal Reserve System;
- (v) which is for goods that have been sold under a purchase order or pursuant to the terms of a contract or other agreement or understanding (written or oral) that indicates or purports that any Person other than a Borrower has or has had an ownership interest in such goods, or which indicates any party other than a Borrower as payee or remittance party;
 - (w) which was created on cash on delivery terms; or
- (x) which the Lender determines may not be paid by reason of the Account Debtor's inability to pay or which the Lender otherwise determines, in each case in its Permitted Discretion is unacceptable for any reason whatsoever.

In the event that an Account in excess of \$100,000 that was previously an Eligible Account ceases to be an Eligible Account hereunder, the Borrower Representative shall notify the Lender thereof within five (5) Business Days. In determining the amount of an Eligible Account, the face amount of an

Account may, in the Lender's Permitted Discretion, be reduced by, without duplication, to the extent not reflected in such face amount, (i) the amount of all accrued and actual discounts, claims, credits or credits pending, promotional program allowances, price adjustments, finance charges or other allowances (including any amount that the Borrowers may be obligated to rebate to an Account Debtor pursuant to the terms of any agreement or understanding (written or oral)) and (ii) the aggregate amount of all cash received in respect of such Account but not yet applied by the Borrowers to reduce the amount of such Account.

"Eligible Inventory" shall mean, at any time, Inventory of the Borrowers that the Lender determines in its Permitted Discretion is eligible as the basis for the extension of Revolving Loans and the issuance of Letters of Credit. Without limiting the Lender's discretion provided herein, Eligible Inventory shall not include any Inventory:

- (a) which is not subject to a first priority perfected Lien in favor of the Lender;
- (b) which is subject to any Lien other than a Permitted Encumbrance that does not have priority over the Lien in favor of the Lender;
- (c) which, the Lender determines, in its Permitted Discretion, is slow moving, obsolete, unmerchantable, defective, unfit for sale, not salable at prices approximating at least the cost of such Inventory in the ordinary course of business or unacceptable due to age, type, category and/or quantity;
- (d) with respect to which any covenant, representation, or warranty contained in this Agreement has been breached or is not true and which does not conform to all standards imposed by any Governmental Body;
- (e) in which any Person other than a Borrower shall (i) have any direct or indirect ownership, interest or title to such Inventory or (ii) be indicated on any purchase order or invoice with respect to such Inventory as having or purporting to have an interest therein;
- (f) which is not finished goods or which constitutes work-in-process, spare or replacement parts, subassemblies, packaging and shipping material, manufacturing supplies, samples, prototypes, displays or display items, bill-and-hold goods, goods that are returned or marked for return, repossessed goods, defective or damaged goods, goods held on consignment, or goods which are not of a type held for sale in the ordinary course of business;
- (g) which is not located in the U.S. or is in transit with a common carrier from vendors and suppliers;
- (h) which is located in any location leased by a Borrower unless the lessor has delivered to the Lender a Waiver;
- (i) which is located in any third party warehouse or is in the possession of a bailee (other than a third party processor) and is not evidenced by a Document unless such warehouseman or bailee has delivered to the Lender a Waiver and such other documentation as the Lender may require;
- (j) which is being processed offsite at a third party location or outside processor, or is in transit to or from said third party location or outside processor;
 - (k) which is a discontinued product or component thereof;

- (l) which is the subject of a consignment by a Borrower as consignor;
- (m) which is perishable;
- (n) which contains or bears any Intellectual Property rights licensed to a Borrower unless the Lender is satisfied that it may sell or otherwise dispose of such Inventory without (i) infringing the rights of such licensor, (ii) violating any contract with such licensor, or (iii) incurring any liability with respect to payment of royalties other than royalties incurred pursuant to sale of such Inventory under the current licensing agreement;
 - (o) which is not reflected in a current perpetual inventory report of the Borrowers;
 - (p) for which reclamation rights have been asserted by the seller; or
- (q) which the Lender otherwise determines in its Permitted Discretion is unacceptable for any reason whatsoever.

In the event that Inventory in excess of \$100,000 that was previously Eligible Inventory ceases to be Eligible Inventory hereunder, the Borrower Representative shall notify the Lender thereof within five (5) Business Days.

"Environmental Laws" shall mean all federal, state and local environmental, land use, zoning, health, chemical use, safety and sanitation laws, statutes, ordinances and codes relating to the protection of the environment and/or governing the use, storage, treatment, generation, transportation, processing, handling, production or disposal of Hazardous Substances and the rules, regulations, policies, guidelines, interpretations, decisions, orders and directives of Governmental Bodies with respect thereto.

"Equity Interests" shall mean shares of capital stock, partnership interests, membership interests in a limited liability company, beneficial interests in a trust or other equity ownership interests in a Person, and any warrants, options or other rights entitling the holder thereof to purchase or acquire any such equity interest.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, and the rules and regulations promulgated thereunder.

"Eurocurrency Reserve Percentage" shall mean, for any Interest Period in respect of any Libor Rate Loan, as of any date of determination, the aggregate of the then stated maximum reserve percentages (including any marginal, special, emergency or supplemental reserves), expressed as a decimal, applicable to such Interest Period (if more than one such percentage is applicable, the daily average of such percentages for those days in such Interest Period during which any such percentages shall be so applicable) by the Board of Governors of the Federal Reserve System, any successor thereto, or any other banking authority, domestic or foreign, to which the Lender may be subject in respect to eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of the Federal Reserve Board) or in respect of any other category of liabilities including deposits by reference to which the interest rate on Libor Rate Loans is determined or any category of extension of credit or other assets that include the Libor Rate Loans. For purposes hereof, such reserve requirements shall include those imposed under Regulation D of the Federal Reserve Board and the Libor Rate Loans shall be deemed to constitute Eurocurrency Liabilities subject to such reserve requirements without benefit of credits for proration, exceptions or offsets which may be available from time to time to the Lender under said Regulation D.

"Event of Default" shall have the meaning set forth in Article 10.

"Excess Availability" shall mean, as of any time of determination, (a) the Maximum Borrowing Amount <u>minus</u> (b) the Revolving Exposure, <u>minus</u> (c) all amounts due and owing to the Borrowers' trade creditors which are outstanding sixty (60) days or more beyond the due date, <u>minus</u> (d) fees and expenses for which the Borrowers are liable to the Lender but which have not been paid.

"Excluded Asset" shall mean any of the following:

- (a) a leasehold mortgage on the Borrowers' leased location located at 294/296 Harper Street, Nelsonville, Ohio 45764;
- (b) depository accounts and security accounts maintained solely for one or more of the following: payroll, medical expenses and payments, pension benefits, employee wages, withholding taxes, social security taxes; sales taxes, or accounts otherwise held exclusively for the benefit of third parties, other than the Loan Parties or their Affiliates;
- (c) any intent-to-use trademark application to the extent and for so long as creation by a Borrower of a Lien therein would result in the loss by such Person of any material rights therein:
- (d) any Borrower's rights or interests in or under any license, contract, permit, instrument, security or franchise or any of its rights thereunder, in each case to the extent such a grant would, under the terms of such contract, result in a breach of, or constitute a default under, such contract; provided that rights to payment and other rights under any such contract shall be included in the Collateral to the extent permitted by, or to the extent such terms are deemed ineffective by, Section 9-406, 9-407, 9-408 or 9-409 of the UCC, and provided, however, such property shall be included in Collateral at such time as the restriction terminates; and
- (e) Investment Property to the extent a security interest in such assets would result in adverse tax consequences (including as a result of the operation of Section 956 of the IRC or any similar law or regulation in any applicable jurisdiction) to the Borrowers or any Subsidiary as reasonably determined in good faith by the Borrower Representative in writing delivered to the Lender;

provided, however, that "Excluded Assets" shall not include any Proceeds, products, substitutions or replacements of any Excluded Asset (unless such Proceeds, products, substitutions or replacements would themselves constitute an Excluded Asset under paragraphs (a) through (d); and provided further that any such property or asset that at any time ceases to satisfy the criteria for an Excluded Asset and constitutes Collateral (whether as a result of the applicable Loan Party obtaining any necessary consent, any change in applicable law, or otherwise), shall no longer be an Excluded Asset.

"Excluded Swap Obligation" shall mean, with respect to any Guarantor, any Swap Obligation if, and to the extent that, all or a portion of the guarantee of such Guarantor of, or the grant by such Guarantor of a security interest to secure, such Swap Obligation (or any guarantee thereof) is or becomes illegal under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) (a) by virtue of such Guarantor's failure for any reason to constitute an "eligible contract participant" as defined in the Commodity Exchange Act and the regulations thereunder at the time the guarantee of such Guarantor or the grant of such security interest becomes or would become effective with respect to such Swap Obligation or (b) in the case of a Swap Obligation subject to a clearing requirement pursuant to Section 2(h) of the Commodity Exchange Act (or any successor provision thereto), because such Guarantor is a "financial entity," as defined in Section 2(h)(7)(C)(i) the Commodity Exchange Act (or any successor

provision thereto), at the time the guarantee of such Guarantor becomes or would become effective with respect to such related Swap Obligation. If a Swap Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Swap Obligation that is attributable to swaps for which such guarantee or security interest is or becomes illegal.

"Executive Order No. 13224" shall mean Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"Facility Termination Date" shall mean November 22, 2019.

"Federal Funds Effective Rate" shall mean, for any day, the rate per annum (rounded upwards, if necessary, to the nearest one hundredth of one percent (1/100 of 1%)) equal to the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided, however, that: (a) if the day for which such rate is to be determined is not a Business Day, the Federal Funds Effective Rate for such day shall be such a rate on such transactions on the immediately preceding Business Day as so published on the next succeeding Business Day and (b) if such rate is not so published for any Business Day, the Federal Funds Effective Rate for such Business Day shall be the average of quotations for such day on such transactions received by the Lender from three federal funds brokers of recognized standing selected by the Lender.

"<u>Financial Officer</u>" shall mean any of the following officers: chief executive officer, president, chief financial officer, treasurer, or accounting manager. Unless otherwise qualified, all references to a Financial Officer in this Agreement shall refer to a Financial Officer of a Borrower.

"Fixed Charge Coverage Ratio" shall mean and include, with respect to any fiscal period, the ratio of (a) EBITDA <u>less</u> (i) Capital Expenditures that were not specifically funded by Indebtedness (provided, that, Capital Expenditures that were funded by a Revolving Loan shall not be netted from Capital Expenditures in calculating the Fixed Charge Coverage Ratio, unless such any Revolving Loan is reimbursed pursuant to Indebtedness incurred pursuant to clause (c) of Section 7.8 within the time period set forth in Section 7.8), <u>less</u> (ii) taxes paid in cash, <u>less</u> (iii) dividends and distributions paid in cash, excluding Permitted Stock Repurchases paid in cash, in each case, of the Borrowers and their Subsidiaries calculated on a consolidated basis with respect to such period, <u>plus</u> the amount of any Shareholder Payment made with respect to fiscal quarters included in such period to (b) Fixed Charges.

"Fixed Charges" shall mean, with respect to any fiscal period, the sum of (a) interest expense paid in cash <u>plus</u> (b) principal payments on Indebtedness (other than the Revolving Loan), including, without limitation, capital lease payments, in each case, of the Borrowers and their Subsidiaries calculated on a consolidated basis and with respect to such period.

"<u>Foreign Subsidiary</u>" shall mean a Subsidiary that is organized under the laws of any jurisdiction other than the United States, any State thereof or the District of Columbia.

"GAAP" shall mean generally accepted accounting principles in the United States of America in effect from time to time.

"Governmental Body" shall mean any nation or government, any state or other political subdivision thereof or any entity exercising the legislative, judicial, regulatory or administrative functions of or pertaining to a government.

"<u>Guarantor</u>" shall mean, individually or collectively, as the context may require, any Person that guaranties all or any portion of the Obligations, and, in each case, their respective successors and assigns.

"Hazardous Substance" shall mean, without limitation, any flammable explosives, radon, radioactive materials, asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, petroleum and petroleum products, methane, hazardous materials, Hazardous Wastes, hazardous or Toxic Substances or related materials as defined in CERCLA, the Hazardous Materials Transportation Act, as amended (49 U.S.C. Sections 1801, et seq.), RCRA or any other applicable Environmental Law and in the regulations adopted pursuant thereto.

"<u>Hazardous Wastes</u>" shall mean all waste materials subject to regulation under CERCLA, RCRA or applicable state law, and any other applicable Federal and state laws now in force or hereafter enacted relating to hazardous waste disposal.

"<u>Hedging Contracts</u>" shall mean any foreign exchange contract, currency swap agreement, futures contract, commodities hedge agreement, interest rate protection agreement, interest rate future agreement, interest rate swap agreement, interest rate cap agreement, interest rate collar agreement, option agreement or any other similar hedging agreement or arrangement entered into by a Person in the ordinary course of business and not for speculative purposes.

"Hedging Obligations" shall mean all liabilities of a Person under Hedging Contracts.

"Indebtedness" shall mean, with respect to a Person at any date of determination, any and all indebtedness, obligations or liabilities (but not including trade payables and accrued expenses incurred in the ordinary course of business which are not represented by a promissory note or other evidence of indebtedness and which are not more than sixty (60) days past due) (whether matured or unmatured, liquidated or unliquidated, direct or indirect, absolute or contingent, or joint or several) of such Person, in each case, for or in respect of: (a) borrowed money, (b) amounts raised under or liabilities in respect of any note purchase or acceptance credit facility, (c) reimbursement obligations (contingent or otherwise) under any letter of credit, (d) Hedging Obligations, (e) any other transaction (including forward sale or purchase agreements, capitalized leases and conditional sales agreements) having the commercial effect of a borrowing of money entered into by such Person to finance its operations or capital requirements, (f) any guaranty of Indebtedness for borrowed money, and (g) all indebtedness secured by a Lien on assets owned by such Person, whether or not such indebtedness actually shall have been created, assumed or incurred by such Person, other the Indebtedness of the type contained in clauses (b), (c), (d), (e), (g), (h), (j), (k), (m), and (n) of the definition of Permitted Encumbrances.

"Intellectual Property" shall mean patents, patent rights, patent applications, copyrights, works which are the subject matter of copyrights, copyright registrations, trademarks, trade names, trade styles, trademark and service mark applications, and licenses and rights to use any of the foregoing; all extensions, renewals, reissues, divisions, continuations, and continuations-in-part of any of the foregoing; all rights to sue for past, present and future infringement of any of the foregoing; inventions, trade secrets, formulae, processes, compounds, drawings, designs, blueprints, surveys, reports, manuals, and operating standards; goodwill; customer and other lists in whatever form maintained; and trade secret rights, copyright rights, rights in works of authorship, and contract rights relating to computer software programs, in whatever form created or maintained.

"Interest Period" shall mean the period provided for any Libor Rate Loan pursuant to Section 2.2(b).

"<u>Issuer</u>" shall mean, with respect to any Letter of Credit, the issuer of such Letter of Credit and shall be, with respect to any Letter of Credit hereunder, KeyBank National Association, and each of its successors and assigns.

"<u>Lender</u>" shall mean, initially, KeyBank National Association, and shall include each Person which becomes a transferee, successor or assign of the Lender.

"<u>Letter of Credit Exposure</u>" shall mean, at any time, the aggregate undrawn amount of all outstanding Letters of Credit <u>plus</u> the aggregate amount of all disbursements relating to Letters of Credit that have not been reimbursed by the Borrowers.

"Letter of Credit Fees" shall have the meaning set forth in Section 3.2.

"Letters of Credit" shall have the meaning set forth in Section 2.7.

"Libor Rate" shall mean, for any Interest Period with respect to a Libor Rate Loan, the quotient (rounded upwards, if necessary, to the nearest one sixteenth of one percent (1/16 of 1%)) of: (a) the per annum rate of interest, determined by the Lender in accordance with its usual procedures (which determination shall be conclusive absent manifest error) as of approximately 12:00 noon (London time) two (2) Business Days prior to the beginning of such Interest Period pertaining to such Libor Rate Loan, as provided by Reuters (or any other similar company or service that provides rate quotations comparable to those currently provided by such companies as the rate in the London interbank market), as determined by the Lender from time to time for purposes of providing quotations of interest rates applicable to deposits in Dollars or in the London interbank market, as the rate in the London interbank market for deposits in Dollars in immediately available funds with a maturity comparable to such Interest Period divided by (b) a number equal to 1.00 minus the Eurocurrency Reserve Percentage. In the event that such rate quotation is not available for any reason, then the rate (for purposes of clause (a) hereof) shall be the rate, determined by the Lender as of approximately 12:00 noon (London time) two (2) Business Days prior to the beginning of such Interest Period pertaining to such Libor Rate Loan, to be the average (rounded upwards, if necessary, to the nearest one sixteenth of one percent (1/16 of 1%)) of the per annum rates at which deposits in Dollars in immediately available funds in an amount comparable to such Libor borrowing and with a maturity comparable to such Interest Period are offered to the prime banks by leading banks in the London interbank market. The Libor Rate shall be adjusted automatically on and as of the effective date of any change in the Eurocurrency Reserve Percentage.

"Libor Rate Loan" shall mean any Loan that bears interest based on the Libor Rate.

"<u>Lien</u>" shall mean any mortgage, deed of trust, pledge, hypothecation, assignment, security interest, lien (whether statutory or otherwise), Charge, claim or encumbrance, or preference, priority or other security agreement or preferential arrangement held or asserted in respect of any asset of any kind or nature whatsoever including any conditional sale or other title retention agreement, any lease having substantially the same economic effect as any of the foregoing, and the filing of, or agreement to give, any financing statement under the Uniform Commercial Code or comparable law of any jurisdiction.

"Loan" shall mean each Revolving Loan; and "Loans" shall collectively mean all of the Revolving Loans.

"Loan Account" shall have the meaning set forth in Section 2.6.

"<u>Loan Documents</u>" shall mean this Agreement, the Notes, the Perfection Certificate, the Letters of Credit, the Waivers, the Mortgages, if any, any Hedging Contracts with the Lender or its Affiliates, and any and all other agreements, instruments and documents, including guaranties, pledges, powers of

attorney, consents, and all other writings heretofore, now or hereafter executed by any Loan Party and/or delivered to the Issuer or the Lender in respect of the transactions contemplated by this Agreement.

"<u>Loan Party</u>" or "<u>Loan Parties</u>" shall mean, singularly or collectively, as the context may require, each Borrower, each additional Borrower that may join this Agreement after the Closing Date, and their respective successors and assigns.

"<u>Lockbox</u>" shall mean a post office box rented by and in the name of the Borrowers as required by this Agreement and as to which only the Lender has access pursuant to the requirements of this Agreement.

"Master Agreement" shall mean that Master Agreement entered into among the Borrowers and the Lender in connection with the cash management services undertaken by the Lender on behalf of Borrowers.

"Material Adverse Effect" shall mean a material adverse effect on (a) the financial condition, results of operations, or business of the Loan Parties, taken as a whole, (b) the Loan Parties' ability taken as a whole to pay the Obligations in accordance with the terms thereof, (c) the value of the Collateral taken as a whole, the Lender's Liens on the Collateral, or the priority of any such Lien, or (d) the practical realization of the benefits of the Lender's rights and remedies under the Loan Documents, taken as a whole.

"Material Business Agreement" shall mean any agreement that if terminated, rescinded or breached would have a Material Adverse Effect.

"Material Recovery Determination Notice" shall have the meaning set forth in Section 4.11.

"Material Recovery Event" shall mean (a) any casualty loss in respect of assets of a Loan Party covered by casualty insurance, and (b) any compulsory transfer or taking under threat of compulsory transfer of any asset of a Loan Party by any Governmental Authority; provided that, in the case of either subpart (a) or (b), the proceeds received from such loss, transfer or taking exceeds One Hundred Thousand Dollars (\$100,000).

"Maximum Borrowing Amount" shall mean, at any time, an amount equal to the lesser of (a) the Revolving Commitment *minus* all Reserves then in effect and (b) the Borrowing Base.

"Mortgages" shall mean each mortgage or other agreement that conveys or evidences a Lien to the Lender, for its benefit and the benefit of the Issuer, on the Owned Real Property.

"Multiemployer Plan" shall mean a "multiemployer plan" as defined in Sections 3(37) and 4001(a)(3) of ERISA.

"Net Orderly Liquidation Value" shall mean, the orderly liquidation value (net of costs and expenses estimated to be incurred in connection with such liquidation) of the Borrowers' Inventory or Equipment that is estimated to be recoverable in an orderly liquidation of such Inventory or Equipment expressed as a percentage of the net book value thereof, such percentage to be as determined from time to time by reference to the most recent Inventory or Equipment appraisal completed by a qualified third-party appraisal company (approved by the Lender in its Permitted Discretion) delivered to the Lender.

"Note" shall mean each Revolving Note; and "Notes" shall collectively mean all of the Revolving Notes.

"Obligations" shall mean and include any and all loans, advances, debts, liabilities, obligations, covenants and duties (absolute, contingent, matured or unmatured) owing by the Loan Parties to the Lender or the Issuer or to any other direct or indirect Subsidiary or Affiliate of the Issuer or the Lender of any kind or nature, present or future (including any interest accruing thereon after maturity, or after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding relating to any Loan Party, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding), whether or not evidenced by any note, guaranty or other instrument, whether arising under any agreement, instrument or document (including the Loan Documents), whether or not for the payment of money, whether arising by reason of an extension of credit, opening of a letter of credit, loan, equipment lease or guarantee, under any Hedging Contract or in connection with any commercial credit cards, stored value cards, cash management or treasury administration services or in any other manner, whether arising out of overdrafts or deposit or other accounts or electronic funds transfers (whether through automated clearing houses or otherwise) or out of the Issuer's or the Lender's nonreceipt of or inability to collect funds or otherwise not being made whole in connection with depository transfer check or other similar arrangements, whether direct or indirect (including those acquired by assignment or participation), absolute or contingent, joint or several, due or to become due, now existing or hereafter arising, contractual or tortious, liquidated or unliquidated, regardless of how such indebtedness or liabilities arise or by what agreement or instrument they may be evidenced or whether evidenced by any agreement or instrument, including any and all of any Loan Party's Indebtedness and/or liabilities under the Loan Documents or under any other agreement between the Issuer or the Lender and any Loan Party and any amendments, extensions, renewals or increases and all reasonable and customary costs and expenses of the Lender and the Issuer for which the Loan Parties are liable under Section 15.9 incurred in the documentation, negotiation, modification, enforcement, collection or otherwise in connection with any of the foregoing, including reasonable and documented attorneys' fees and expenses and all obligations of any Loan Party to the Lender or the Issuer to perform acts or refrain from taking any action; provided, however, that the definition of "Obligations" shall not create any guarantee by any Guarantor of (or grant of security interest by any Guarantor to support, as applicable) any Excluded Swap Obligations of such Guarantor for purposes of determining any obligations of any Guarantor.

"Operating Account" shall mean a commercial Deposit Account designated "operating account" and maintained by the Borrower Representative with the Lender, without liability of the Lender to pay interest thereon, from which account the Borrowers shall have the right to withdraw funds until the Lender terminates such right after the occurrence and during the continuance of an Event of Default.

"Owned Real Property" shall mean all Real Property set forth on Schedule 1.2(a).

"Payment Office" shall mean initially KeyBank National Association, Mail Code: OH-01-49-0114, 4900 Tiedeman Road, Brooklyn, Ohio 44144-2302; and, thereafter, such other office of the Lender, if any, which it may designate by notice to the Borrower Representative.

"Permitted Acquisition" shall mean any Acquisition by any Borrower, to the extent that each of the following conditions shall have been satisfied:

- (a) the assets, business or Person being acquired is engaged in the same business of the Borrowers and their Subsidiaries;
- (b) (i) in the case of a stock acquisition, the target thereof shall be a Domestic Subsidiary of a Borrower and (ii) in the case of an asset acquisition, the assets being acquired shall be located in the United States;
- (c) the Borrower Representative shall have delivered to the Lender (i) as soon as available (but no later than five (5) Business Days after the consummation of such acquisition),

final executed counterparts of the material agreements, documents and instruments pursuant to which such acquisition is to be consummated (including any management, non-compete, employment and option agreements) and any schedules to such agreements, documents and instruments, (ii) to the extent required under the related acquisition agreement, all consents and approvals from applicable governmental authorities and other Persons required to consummate such acquisition and (iii) if reasonably requested by the Lender, environmental assessments reasonably satisfactory to the Lender;

- (d) the Borrowers and their Subsidiaries (including any new Subsidiary) shall execute and deliver the agreements, instruments and other documents required by Sections 4.2, 4.20, and 7.12, and upon the consummation of such acquisition to the extent required by Section 7.12, the target thereof shall be a Borrower hereunder;
- (e) there shall not exist on the date on which the acquisition is consummated, both before and after giving effect to the acquisition, a Default or Event of Default;
- (f) such acquired assets and/or Equity Interests shall be acquired free and clear of all Liens (except for Permitted Liens);
- (g) if the Accounts and Inventory acquired in connection with such acquisition are proposed to be included in the determination of the Borrowing Base, the Lender shall have conducted an audit and field examination of such Accounts and Inventory, the results of which shall be reasonably satisfactory to the Lender;
- (h) the total consideration paid or payable (including all transaction costs and all Indebtedness incurred, assumed and/or reflected on a consolidated balance sheet of the Borrowers and their Subsidiaries after giving effect to such Acquisition) shall not exceed for all Acquisitions consummated during the term of this Agreement, \$1,250,000 <u>less</u> the amount paid for Permitted Stock Repurchases during the term of this Agreement; and
- (i) as soon as available, but not less than five (5) days prior to such acquisition, the Borrower Representative shall have provided the Lender (i) notice of such acquisition and (ii) a copy of all business and financial information reasonably requested by the Lender including proforma financial statements, statements of cash flow, and availability projections.

"<u>Perfection Certificate</u>" shall mean the initial questionnaire and the responses thereto provided by the Borrowers to the Lender.

"<u>Permitted Discretion</u>" shall mean a determination made in good faith and in the exercise of reasonable business judgment (from the perspective of a secured asset based lender).

"Permitted Dispositions" shall mean any of the following:

- (a) non-exclusive licenses of intellectual property;
- (b) the sale or discount of not more than \$100,000 in the aggregate of overdue accounts receivable that are no longer Eligible Accounts in connection with collections of the same;
- (c) dispositions resulting from any casualty or other insured damage to, or any taking under power of eminent domain or by condemnation or similar proceeding of, any property or asset of any Loan Party;

- (d) sales, transfers and other dispositions of used, damaged, surplus, obsolete, or outmoded, or not useful machinery, equipment or worn out assets and scrap;
 - (e) Permitted Stock Repurchases; and
- (f) other sales, leases, assignments, transfers or other dispositions for consideration not in excess of \$25,000 in the aggregate in any fiscal year.

"Permitted Encumbrances" shall mean (a) Liens in favor of the Lender and the Issuer; (b) Liens for taxes, assessments or other Charges that (i) are not delinquent or (ii) are being contested in good faith by appropriate proceedings that stay the enforcement of such Liens and with respect to which proper reserves have been taken by the Loan Parties in accordance with GAAP; provided, that, such Liens shall have no effect on the priority of the Liens in favor of the Lender or the value of the assets in which the Lender has such a Lien and a stay of enforcement of any such Lien shall be in effect; (c) deposits or pledges to secure obligations under worker's compensation, social security or similar laws, or under unemployment insurance or general liability or product liability insurance; (d) deposits or pledges to secure bids, tenders, contracts (other than contracts for the payment of money), leases, statutory obligations, performance bonds, surety and appeal bonds and other obligations of like nature arising in the ordinary course of any Loan Party's business; (e) mechanics, workers, materialmen's, warehousemen's, common carriers, landlord's or other like Liens arising in the ordinary course of any Loan Party's business with respect to obligations which are not due or which are being contested in good faith by the applicable Loan Party; (f) Liens placed upon equipment and real estate assets acquired to secure a portion of the purchase price thereof, provided that (i) any such Lien shall not encumber any other property of the Loan Parties other than insurance and other proceeds of such equipment and real estate and (ii) the aggregate amount of Indebtedness secured by such Liens incurred as a result of such purchases during any calendar year shall not exceed the amount provided for in Section 7.68; (g) zoning restrictions, easements, encroachments, rights of way, restrictions, leases, licenses, restrictive covenants and other similar title exceptions or Liens affecting Real Property, none of which materially impairs the use of such Real Property or the value thereof, and none of which is violated in any material respect by existing or supporting structures or land use; (h) attachment and judgment liens which do not constitute an Event of Default under Section 10.6; (i) Liens disclosed on Schedule 1.2(b) provided that the principal amount secured thereby is not hereafter increased, and no additional assets become subject to such Lien; (j) bankers' liens, rights of setoff and other similar Liens in the ordinary course of business in favor of a bank or institution with which accounts or deposits are maintained, liens in favor of collecting banks arising under the UCC in the ordinary course of business, and other Liens that constitute contractual rights of setoff; (k) Liens arising from precautionary UCC financing statement filings (or similar filings under applicable law) regarding "true leases" or operating leases or, to the extent permitted under the Loan Documents, the consignment of goods to a Loan Party and Liens in connection with the financing of motor vehicles; (1) the replacement, extension or renewal of any Lien permitted by clause (f) above upon or in the same property subject thereto arising out of the extension, renewal or replacement of the Indebtedness secured thereby; (m) Liens arising in the ordinary course of business in favor of customs brokers, custom and forwarding agents and similar Persons in respect of imported goods and merchandise in the custody of such Persons; and (n) inchoate or statutory Liens that are related to obligations incurred in the ordinary course of business and not due or delinquent.

"Permitted Guaranties" shall mean any of the following:

(a) any guarantee obligation arising under the Loan Documents;

- (b) any indemnities by any Loan Party of any liability of its directors, officers and employees in their capacities as such as permitted by applicable law or in connection with real property; and
- (c) any guaranty of any Loan Party of Indebtedness of another Loan Party permitted by Section 7.8.

"<u>Permitted Holders</u>" shall mean (a) Michael Mark, Kerry Pigman and Gregory P. Smith and their spouses, children and grandchildren, (b) the respective estates of, and any trusts created for the benefit of, the foregoing Persons, or (c) any trust, corporation, partnership, limited liability company or other entity, the beneficiaries, stockholders, partners or owners, as applicable, of which consist of the foregoing Persons.

"Permitted Indebtedness" shall mean any of the following:

- (a) Indebtedness of a Borrower owed to another Loan Party and of a Loan Party owed to another Loan Party;
- (b) guarantees by any Borrower of Indebtedness of any other Borrower and by any Subsidiary of Indebtedness of any Borrower or any other Subsidiary, <u>provided</u> that (i) the Indebtedness so guaranteed is permitted by Section 7.8, (ii) guarantees by any Loan Party of Indebtedness of any Subsidiary that is not a Loan Party shall be subject to Section 7.4, and (iii) guarantees permitted under this clause (c) shall be subordinated to the Obligations on the same terms as the Indebtedness so guaranteed is subordinated to the Obligations, if applicable;
- (c) Indebtedness which represents amendments, restatements, supplements, extensions, renewals, refinancing or replacements (such Indebtedness being so amended, restated, supplemented, extended, renewed, refinanced or replaced being referred to herein as the "Refinance Indebtedness") of any of the Indebtedness described in Section 7.8 (such Indebtedness being referred to herein as the "Original Indebtedness"); provided that (i) such Refinance Indebtedness does not increase the principal amount of the Original Indebtedness, (ii) any Liens securing such Refinance Indebtedness are not extended to any additional property of any Loan Party, and (iii) if such Original Indebtedness was subordinated in right of payment to any of the Obligations, then the terms and conditions of such Refinance Indebtedness must include subordination terms and conditions that are at least as favorable to the Administrative Agent and the Lenders as those that were applicable to such Original Indebtedness;
- (d) Indebtedness of any Loan Party owed to any Person providing workers' compensation, retirement, health, disability or other employee benefits or property, casualty or liability insurance, pursuant to reimbursement or indemnification obligations to such Person, in each case incurred in the ordinary course of business;
- (e) Indebtedness of any Loan Party in respect of performance bonds, bid bonds, appeal bonds, surety bonds and similar obligations, in each case provided in the ordinary course of business;
- (f) Indebtedness of any Loan Party in respect of netting services, overdraft protections and otherwise in connection with deposit and checking accounts, in each case, in the ordinary course of business;

- (g) obligations under any agreement governing the provision of treasury or cash management services, including deposit accounts, overnight draft, credit cards, debit cards, p-cards (including purchasing cards and commercial cards), funds transfer, automated clearinghouse, zero balance accounts, returned check concentration, controlled disbursement, lockbox, account reconciliation and reporting and other cash management services;
- (h) Indebtedness with respect to the deferred purchase price for any Permitted Acquisition, provided that such Indebtedness is subordinated to the Obligations on terms reasonably acceptable to the Lender;
 - (i) Shareholder Subordinated Debt; and
- (j) other unsecured Indebtedness in an aggregate principal amount not to exceed \$25,000 at any time outstanding.

"Permitted Investments" shall mean any of the following:

- (a) Those items described in clauses (a) through (e) of Section 7.4;
- (b) loans or advances made by a Loan Party to its employees, officers, or directors on an arm's-length basis in the ordinary course of business consistent with past practices for travel and entertainment expenses, relocation costs and similar purposes in an aggregate outstanding amount not to exceed \$50,000 at any time;
- (c) accounts receivable, notes payable, or stock or other securities issued by Account Debtors to a Loan Party pursuant to negotiated agreements with respect to settlement of such Account Debtor's Accounts obligations in the ordinary course of business, consistent with past practices, and investments received in satisfaction or partial satisfaction thereof from financially troubled Account Debtors to the extent reasonably necessary in order to prevent or limit loss or received in connection with the bankruptcy or reorganization of customers or suppliers, or settlement of disputes with suppliers, in each case in the ordinary course of business;
 - (d) investments in the form of Hedging Contracts permitted by this Agreement;
- (e) investments of any Person existing at the time such Person becomes Subsidiary of a Borrower or consolidates or merges with a Borrower (including in connection with a Permitted Acquisition) so long as such investments were not made in contemplation of such Person becoming a Subsidiary or of such merger;
- (f) investments made as a result of receipt of non-cash consideration from a sale, transfer or other disposition of assets permitted under this Agreement;
- (g) investments constituting deposits described in the definition of the term "Permitted Encumbrances";
 - (h) Permitted Acquisitions; and
- (i) any other investments in an aggregate amount not to exceed \$50,000 at any time outstanding.

"<u>Permitted Stock Repurchases</u>" shall mean repurchases or redemptions by the Borrower Representative of its preferred stock, to the extent that each of the following conditions shall have been satisfied:

- (a) there shall not exist on the date on which the repurchase or redemption is consummated, both before and after giving effect to the repurchase or redemption, a Default or Event of Default;
- (b) the total consideration paid or payable (including all transaction costs) shall not exceed for all such repurchases and redemptions consummated during the term of this Agreement, \$1,250,000 <u>less</u> the amount paid for Permitted Acquisitions during the term of this Agreement; and
- (c) as soon as possible, but not less than five (5) days prior to such repurchase or redemption, the Borrower Representative shall have provided the Lender with notice of the same.

"Person" shall mean any individual, sole proprietorship, partnership, corporation, business trust, joint stock company, trust, unincorporated organization, association, limited liability company, institution, public benefit corporation, joint venture, entity or Governmental Body.

"Plan" shall mean any employee pension benefit plan (other than a Multiemployer Plan) that is subject to the provisions of Title IV of ERISA or Sections 412 and 430 of the Code or Section 302 of ERISA and in respect of which any Loan Party, or any member of a Controlled Group in which a Loan Party is a member, is (or, if such Plan were terminated, would under Section 4062 or Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"Pledged Securities" shall mean all of the Equity Interest of a Subsidiary of a Loan Party, whether now owned or hereafter acquired or created, and all proceeds thereof; provided that Pledged Securities shall exclude (a) Equity Interests of any Foreign Subsidiary that is not a first-tier Foreign Subsidiary, and (b) Equity Interests in any first-tier Foreign Subsidiary in excess of sixty-five percent (65%) of the total outstanding shares of voting capital stock or other voting equity interest of such first-tier Foreign Subsidiary. (Schedule 5.3 hereto lists, as of the Closing Date, all of the Pledged Securities.)

"Projections" shall have the meaning set forth in Section 5.6(a).

"<u>RCRA</u>" shall mean the Resource Conservation and Recovery Act of 1976, as amended, 42 U.S.C. Section 6901 et seq.

"Real Property" shall mean all real property, both owned and leased, of the Loan Parties.

"Register" shall have the meaning set forth in Section 15.3(d).

"Release" shall have the meaning set forth in Section 5.8(c).

"Reserves" shall mean any and all amounts the Lender deems necessary in it Permitted Discretion to block, withhold, reserve or maintain against the amount of the Loans that the Lender will make available to the Borrowers with respect to the Collateral, any Loan Party, or such other matters as the Lender deems necessary or appropriate.

"Revolving Commitment" shall mean the commitment of the Lender to make Revolving Loans and issue Letters of Credit, as such commitment may be reduced pursuant to the terms of this Agreement. The initial amount of the Lender's Revolving Commitment is \$10,000,000.

"Revolving Exposure" shall mean, at any time, the sum of the outstanding principal amount of Revolving Loans and Letter of Credit Exposure at such time.

"Revolving Loan" shall mean a loan made pursuant to Section 2.1.

"<u>Revolving Note</u>" or "<u>Revolving Notes</u>" shall mean, singularly or collectively, as the context may require, the promissory notes referred to in Section 2.1.

"Sanctioned Country" shall mean, at any time, a country or territory which is itself the subject or target of any Sanctions (at the time of this Agreement, Cuba, Iran, North Korea, Sudan and Syria).

"Sanctioned Person" shall mean, at any time, (a) any Person listed in any Sanctions-related list of designated Persons maintained by the Office of Foreign Assets Control of the U.S. Department of the Treasury, the U.S. Department of State, (b) any Person operating, organized or resident in a Sanctioned Country or (c) any Person owned or controlled by any such Person or Persons described in the foregoing clauses (a) or (b).

"Sanctions" shall mean economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by the U.S. government, including those administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury or the U.S. Department of State.

"<u>Shareholder Payment</u>" shall mean a payment made by one or more holders of the Borrower Representative's Equity Interests to the Borrower Representative consisting of Shareholder Subordinated Debt or contributions to the capital, in each case subject to terms reasonably acceptable to the Lender.

"<u>Shareholder Subordinated Debt</u>" shall mean unsecured loans to the Borrower Representative by any holder of its Equity Interests that is subject to a Subordination Agreement in form and substance reasonably satisfactory to the Lender.

"<u>StuckyNet System</u>" shall mean the Lender's StuckyNet-Link internet-based communication system utilized by the Lender.

"<u>Subordinated Debt</u>" shall mean (a) all obligations of any Borrower or any Subsidiary to any Guarantor, (b) all other Shareholder Subordinated Debt, and (c) any other unsecured Indebtedness of any Borrower or any Subsidiary that has subordination terms, covenants, pricing and other terms that have been approved in writing by the Lender.

"Subordination Agreements" shall mean all subordination agreements executed by a holder of Subordinated Debt in favor of the Bank from time to time after the Closing Date, in each case as amended, restated, supplemented or otherwise modified from time to time.

"Subsidiary" shall mean a corporation or other entity of whose shares of stock or other ownership interests having ordinary voting power (other than stock or other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the directors of such corporation, or other Persons performing similar functions for such entity, are owned, directly or indirectly, by such Person.

"Swap Obligation" shall mean, with respect to any Guarantor, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act.

"Threshold Amount" shall mean \$50,000.

"Toxic Substances" shall mean and include any material present on the Real Property which has been shown to have significant adverse effect on human health or which is subject to regulation under the Toxic Substances Control Act (TSCA), 15 U.S.C. Sections 2601 et seq., applicable state law, or any other applicable Federal or state laws now in force or hereafter enacted relating to toxic substances, and includes asbestos, polychlorinated biphenyls (PCBs) and lead-based paints.

"Transferee" shall have the meaning set forth in Section 15.14(a).

"UFCA" shall have the meaning set forth in Section 14.2.

"UFTA" shall have the meaning set forth in Section 14.2.

"<u>Uniform Commercial Code</u>" shall mean the Uniform Commercial Code as in effect from time to time in the State of Ohio; provided, however, that if by reason of mandatory provisions of law, the perfection or the effect of perfection or non-perfection of the security interest or Lien in any Collateral is governed by the Uniform Commercial Code as in effect in a jurisdiction other than Ohio, "Uniform Commercial Code" shall mean the Uniform Commercial Code as in effect in such other jurisdiction for purposes of the provisions hereof relating to such perfection or effect of perfection or non-perfection.

"<u>USA Patriot Act</u>" shall mean the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"<u>Waivers</u>" shall mean, collectively, any and all landlord's waivers, warehouseman's waivers, creditor's waivers, mortgagee waivers and processing facility and similar bailee's waivers, executed and delivered in connection with this Agreement, in form and substance reasonably satisfactory to the Lender.

"Week" shall mean the time period commencing on Monday and ending on Friday of each calendar week.

- 1.3 <u>Uniform Commercial Code Terms.</u> All terms used herein and defined in the Uniform Commercial Code as adopted in the State of Ohio from time to time shall have the meaning given therein unless otherwise defined herein. Such terms shall include: "<u>Account</u>", "<u>Account Debtor</u>", "<u>Certificated Security</u>", "<u>Chattel Paper</u>", "<u>Commercial Tort Claim</u>", "<u>Commodities Account</u>", "<u>Deposit Account</u>", "<u>Document</u>", "<u>Equipment</u>", "<u>Farm Products</u>", "<u>Financial Asset</u>", "<u>Fixture</u>", "<u>General Intangible</u>", "<u>Instrument</u>", "<u>Inventory</u>", "<u>Investment Property</u>", "<u>Lease</u>", "<u>Lessor</u>", "<u>Letter-of-Credit Rights</u>", "<u>money</u>", "<u>Payment Intangibles</u>", "<u>Proceeds</u>", "<u>Product</u>", "<u>Record</u>", "<u>Secured Party</u>", "<u>Securities Account</u>", "<u>Security Entitlement</u>", "<u>Security Interest</u>" and "<u>Supporting Obligation</u>". To the extent the definition of any category or type of Collateral is expanded by any amendment, modification or revision to the Uniform Commercial Code, such expanded definition will apply automatically as of the effective date of such amendment, modification or revision.
- 1.4 General Matters of Construction. For the purpose of computing periods of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each mean "to but excluding". Unless the context otherwise expressly requires, (a) all references to laws, statutes and regulations shall include any amendments, renewals, extensions, replacements, or successor laws, statutes or regulations, (b) any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement,

instrument, or other document as from time to time amended, supplemented or otherwise modified, substituted, amended and restated, or replaced, (c) any reference herein to any Person shall be construed to include such Person's successors and assigns, (d) the words "herein," "hereof," and "hereunder," and words of similar import, shall be construed to refer to this Agreement in its entirety and not any particular provision hereof, (e) any reference to payment, repayment, or prepayment shall be construed as referring to payment of immediately available funds in Dollars, (f) any pronoun used shall be deemed to cover all genders, (g) any reference to any Loan Document or other deliverable shall mean, unless the context expressly states otherwise, such Loan Document or deliverable in form and substance reasonably satisfactory to the Lender, (h) the words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation", (i) the word "will" shall be construed to have the same meaning and effect as the word "shall", (j) wherever appropriate in the context, terms used herein in the singular also include the plural and vice versa, and (k) captions used in this Agreement are intended for convenience only and do not constitute and shall not be interpreted as part of this Agreement.

1.5 <u>Time References.</u> All time references in the Loan Documents are to Cleveland, Ohio time.

ARTICLE 2 ADVANCES, PAYMENTS

2.1 <u>Revolving Loans</u>. Subject to the terms and conditions set forth in this Agreement, the Lender will make Revolving Loans to the Borrowers in aggregate amounts outstanding at any time prior to the Facility Termination Date equal to the Maximum Borrowing Amount <u>less</u> the Letter of Credit Exposure. The Revolving Loan shall initially bear interest as a Base Rate Loan and may be converted into a Libor Rate Loan in accordance with Section 2.2(d). Revolving Loans shall be evidenced by a secured promissory note substantially in the form attached hereto as <u>Exhibit B</u> hereto (the "<u>Revolving Note</u>").

2.2 **Procedure for Borrowing Advances.**

- (a) The Borrower Representative shall notify the Lender by providing the Lender with a notice of loan substantially in the form attached hereto as Exhibit C hereto (a "Notice of Loan") prior to 11:00 A.M. on a Business Day of a Borrower's request to incur, on that day, a Revolving Loan hereunder. Should any amount required to be paid as interest hereunder, or as fees or other charges under this Agreement or any other agreement with the Lender, or with respect to any other Obligation, become due, the same shall be deemed a request for a Revolving Loan charged to the Loan Account as of the date such payment is due, in the amount required to pay in full such interest, fee, charge or Obligation under this Agreement or any other agreement with the Lender, and such request shall be irrevocable.
- (b) Notwithstanding the provisions of subsection (a) above, in the event a Borrower desires to obtain a Libor Rate Loan, the Borrower Representative shall notify the Lender in writing by providing a Notice of Loan to the Lender no later than 11:00 A.M. at least three (3) Business Days' prior to the date of such proposed borrowing, specifying (i) the date of the proposed borrowing (which shall be a Business Day), (ii) the amount of such Revolving Loan to be borrowed, which amount shall be in a minimum amount of \$500,000 and in integral multiples of \$100,000 thereafter, and (iii) the duration of the first Interest Period therefor. Interest Periods for Libor Rate Loans shall be for one (1), two (2), three (3) or six (6) months; provided, if an Interest Period would end on a day that is not a Business Day, it shall end on the next succeeding Business Day unless such day falls in the next succeeding calendar month in which case the Interest Period shall end on the next preceding Business Day. No Libor Rate Loan shall be made available to the Borrowers during the continuance of a Default or an Event of Default.

- Rate Loan is made and shall end on such date as the Borrower Representative may elect as set forth in subsection (b)(iii) above, provided that the exact length of each Interest Period shall be determined in accordance with the practice of the interbank market for offshore Dollar deposits and no Interest Period shall end on or after the Facility Termination Date. The Borrower Representative shall elect the initial Interest Period applicable to a Libor Rate Loan by its Notice of Loan given to the Lender pursuant to Section 2.2(b) or by its Notice of Loan given to the Lender pursuant to Section 2.2(d), as the case may be. The Borrower Representative shall elect the duration of each succeeding Interest Period by giving irrevocable written notice by providing a Notice of Loan to the Lender of such duration not less than three (3) Business Days prior to the last day of the then current Interest Period applicable to such Libor Rate Loan. If the Lender does not receive timely notice of the Interest Period elected by the Borrower Representative, the Borrower Representative shall be deemed to have elected to convert such Loan to a Base Rate Loan subject to Section 2.2(d).
- Interest Period applicable to any outstanding Libor Rate Loan, or on any Business Day with respect to Base Rate Loans, convert any such Loan into a Loan of another type in the same aggregate principal amount provided that any conversion of a Libor Rate Loan shall be made only on the last Business Day of the then current Interest Period applicable to such Libor Rate Loan. If a Borrower desires to convert a Loan, the Borrower Representative shall give the Lender not less than three (3) Business Days' prior written notice by providing a Notice of Loan to the Lender to convert from a Base Rate Loan to a Libor Rate Loan or one (1) Business Day's prior written notice to convert from a Libor Rate Loan to a Base Rate Loan, specifying the date of such conversion, the Loans to be converted and if the conversion is from a Base Rate Loan to any other type of Loan, the duration of the Interest Period therefor; provided, however, a Borrower shall not be permitted to convert a Base Rate Loan to a Libor Rate Loan or continue to select a Libor Rate Loan during the continuance of a Default or an Event of Default. After giving effect to each such conversion, there shall not be outstanding more than six (6) Libor Rate Loans, in the aggregate.
- (e) At its option and upon three (3) Business Days' prior written notice, a Borrower may prepay the Libor Rate Loans in whole at any time or in part from time to time, without premium or penalty, but with accrued interest on the principal being prepaid to the date of such repayment and the payment of the internal processing fees of the Lender and any breakage fees in each case arising the prepay of such Libor Rate Loans. The Borrower Representative shall specify the date of prepayment of the Loans or Advances which are Libor Rate Loans and the amount of such prepayment. In the event that any prepayment of a Libor Rate Loan is required or permitted on a date other than the last Business Day of the then current Interest Period with respect thereto, the Borrowers shall indemnify the Lender therefor in accordance with Section 3.9.
- 2.3 <u>Disbursement of Loan Proceeds.</u> All Loans shall be disbursed from whichever office or other place the Lender may designate from time to time and, together with any and all other Obligations of the Borrowers to the Lender, shall be charged to the Loan Account on the Lender's books. During the term of this Agreement, the Borrower Representative may use the Revolving Loans by borrowing, prepaying and reborrowing, all in accordance with the terms and conditions hereof. The proceeds of each Revolving Loan requested by the Borrower Representative or deemed to have been requested by any Borrower under Section 2.2(a) shall, with respect to requested Revolving Loans to the extent the Lender makes such Revolving Loans, be made available to the Borrower Representative on the day so requested by way of credit to a Borrower's Operating Account at the Lender, in immediately

available federal funds or other immediately available funds or, with respect to Revolving Loans deemed to have been requested by a Borrower, be disbursed to the Lender to be applied to the outstanding Obligations giving rise to such deemed request.

2.4 <u>Maximum Advances</u>. Subject to Section 4.4, the Revolving Exposure outstanding at any time shall not exceed the Maximum Borrowing Amount. If the Revolving Exposure at any time exceeds the Maximum Borrowing Amount, subject to Section 4.4, such excess shall be immediately due and payable without the necessity of any demand, at the Payment Office, whether or not a Default or Event of Default has occurred.

2.5 **Repayment of Loans.**

- (a) The Loans shall be due and payable in full on the Facility Termination Date subject to earlier prepayment as herein provided.
- (b) Any Account Debtor payment with respect to Accounts which is evidenced by a check, note, draft or any other similar item of payment may not be immediately collectible. In calculating outstanding availability, the Lender agrees that any such item of payment will be deemed to have been received by the Lender and will be provisionally credited to the Loan Account by the Lender on the Business Day immediately following the day on which the Lender has actual possession of such item of payment for deposit to the Cash Concentration Account. In consideration of the Lender's agreement for provisional crediting of items of payment, the Borrowers agree that, in calculating interest and other charges on the Obligations, all Account Debtor payments will be treated as having been credited to the Loan Account on the Business Day immediately following the Business Day on which such payments are deemed to have been received by the Lender pursuant to this paragraph.
- (c) The Lender shall not be required to credit the Loan Account for the amount of any item of payment or other payment which is unsatisfactory to the Lender, until such time as such payment becomes good funds. All credits (other than federal wire transfers) shall be provisional, subject to verification and final settlement. The Lender may charge the Loan Account for the amount of any item of payment or other payment which is returned to the Lender unpaid or otherwise not collected. The Borrowers agree that any information and data reported to the Borrowers pursuant to any service which is received prior to final posting and confirmation is subject to correction and is not to be construed as final posting information. The Lender shall have no liability for the content of such preliminary service related information.
- (d) All payments of principal, interest and other amounts payable hereunder, or under any of the other Loan Documents shall be made to the Lender at the Payment Office not later than 11:00 A.M. on the due date in lawful money of the United States of America in federal funds or other funds immediately available to the Lender. The Lender shall have the right to effectuate payment on any and all Obligations due and owing hereunder by charging the Loan Account or by making Advances as provided in Section 2.2. The aggregate unpaid amount of Loans, types of Loans, Interest Periods and similar information with respect to the Loans and Letters of Credit set forth on the records of Lender shall be rebuttably presumptive evidence with respect to such information, including the amounts of principal, interest and fees owing to the Lender.
- (e) Whenever any payment to be made hereunder, including, without limitation, any payment to be made on any Loan, shall be stated to be due on a day that is not a Business Day, such payment shall be made on the next Business Day and such extension of time shall in each case be included in the computation of the interest payable on such Loan; provided that, with

respect to a Libor Rate Loan, if the next Business Day shall fall in the succeeding calendar month, such payment shall be made on the preceding Business Day and the relevant Interest Period shall be adjusted accordingly.

- (f) The Borrowers shall pay principal, interest, and all other amounts payable hereunder, or under any other Loan Document, without any deduction whatsoever, including any deduction for any setoff or counterclaim.
- 2.6 <u>Statement of Account</u>. The Lender shall maintain, in accordance with its customary procedures, a loan account ("<u>Loan Account</u>") in the name of the Borrowers in which shall be recorded, among other things, the date and amount of each Advance made by the Lender and the date and amount of each payment in respect thereof; provided, however, the failure by the Lender to record the date and amount of any Advance shall not adversely affect the Lender.
- 2.7 Letters of Credit. Subject to the terms and conditions hereof, the Issuer shall (a) issue or cause the issuance of commercial documentary and standby letters of credit ("Letters of Credit") on behalf of the Borrowers; provided, however, that the Issuer will not be required to issue or cause to be issued any Letters of Credit to the extent that the face amount of such Letters of Credit would then cause the Revolving Exposure to exceed the Maximum Borrowing Amount. The maximum amount of the Letters of Credit outstanding shall not exceed \$1,000,000 in the aggregate at any time. The Borrowers shall immediately reimburse the Lender or the Issuer for all disbursements or payments related to Letters of Credit or the amounts of such disbursements or payments shall be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations on the next Business Day.
- 2.8 <u>Issuance of Letters of Credit.</u> The Borrower Representative may request the Issuer to issue or cause the issuance of a Letter of Credit by delivering to the Issuer at the Payment Office the Issuer's form of letter of credit application completed to the satisfaction of the Issuer; and, such other certificates, documents and other papers and information as the Issuer may reasonably request no later than 11:00 A.M. at least three (3) Business Days' prior to the date of such proposed issuance. Each Letter of Credit shall have an expiry date not later than the earlier to occur of (a) 364 days from the date of issuance and (b) 30 days prior to the Facility Termination Date. The Issuer shall have absolute discretion whether to issue any Letter of Credit.

2.9 Requirements For Issuance of Letters of Credit.

- (a) In connection with the issuance of any Letter of Credit, the Borrowers shall indemnify, save and hold the Lender and the Issuer harmless from any loss, cost, expense or liability, including payments made by the Lender or the Issuer and expenses and reasonable attorneys' fees incurred by the Lender or the Issuer arising out of, or in connection with, any Letter of Credit to be issued or created for any Borrower. The Borrowers shall be bound by the Lender's or the Issuer's regulations and good faith interpretations of any Letter of Credit issued or created to the Loan Account, although this interpretation may be different from its own; and, neither the Lender, nor the Issuer nor any of their correspondents shall be liable for any error, negligence, or mistakes, whether of omission or commission, in following any Borrower's instructions or those contained in any Letter of Credit or of any modifications, amendments or supplements thereto or in issuing or paying any Letter of Credit.
- (b) The Borrowers shall authorize and direct the Issuer to deliver to the Lender all instruments, documents, and other writings and property received by the Issuer pursuant to the Letter of Credit and to accept and rely upon the Lender's instructions and agreements with respect to all matters arising in connection with the Letter of Credit, the application therefor or any acceptance therefor.

(c) Immediately upon the request of the Lender, (i) after the occurrence of a Default or an Event of Default, or (ii) if any Letter of Credit remains outstanding after five (5) Business Days prior to the Facility Termination Date, in each such case, the Borrowers will deposit and maintain in an account with the Lender in cash, as cash collateral, in an amount equal to one hundred five percent (105%) of the amount of outstanding Letters of Credit. In each case, the Borrowers hereby irrevocably authorize the Lender, in its discretion, on the Borrowers' behalf and in any Borrower's name, to open such an account and to make and maintain deposits in such account or in an account opened by the Borrowers, in the amounts required to be made by the Borrowers, out of the proceeds of Accounts or other Collateral, from an Advance, or out of any other funds of the Borrowers coming into the Lender's possession at any time. The Lender will invest such cash collateral (*less* applicable Reserves) in such short-term money-market items as to which the Lender and the Borrowers mutually agree and the net return on such investments shall be credited to such account and constitute additional cash collateral. The Borrowers may not withdraw amounts credited to any such account except upon payment and performance in full of all Obligations and termination of this Agreement.

2.10 Reserved.

- 2.11 <u>Additional Payments</u>. Any sums reasonably expended by the Lender due to any Loan Party's failure to perform or comply with its obligations under any Loan Document, including the Loan Parties' obligations under Sections 2.7, 4.2, 4.4, 4.12, 4.13, 4.15, 6.7 and 15.9, may be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations.
- 2.12 <u>Use of Proceeds.</u> The Borrowers shall apply the proceeds of Advances and the Loans (a) to repay existing Indebtedness owed to ACF Finco I, LP, (b) to make Permitted Acquisitions and Permitted Stock Repurchases, (c) to pay fees and expenses relating to the transaction contemplated by this Agreement, (d) for general corporate purposes and (e) to provide for working capital needs. The Borrowers shall not request any Loan or Letter of Credit, and the Borrowers shall not use, and shall ensure that its Subsidiaries and their respective directors, officers, employees and agents shall not use, the proceeds of any Loan or Letter of Credit (A) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws, (B) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person, or in any Sanctioned Country, or (C) in any manner that would result in the violation of any Sanctions applicable to any party hereto.

ARTICLE 3 INTEREST AND FEES

3.1 **Interest.**

- (a) Interest on the Loans shall be payable (i) with respect to Base Rate Loans, in arrears on the first (1st) day of each calendar month and (ii) with respect to LIBOR Loans, on the last day of the Interest Period applicable to the borrowing of which such LIBOR Loan is a part; provided that, if an Interest Period exceeds three months, the Interest must also be paid every three months, commencing with three months from the beginning of such Interest Period. Interest charges shall be computed on the actual principal amount of Loans outstanding during the calendar month.
- (b) Base Rate Loans shall bear interest for each day at a rate per annum equal to the Base Rate <u>plus</u> the Applicable Base Rate Margin, and Libor Rate Loans shall bear interest during each applicable Interest Period at a rate per annum equal to the Libor Rate <u>plus</u> the Applicable Libor Rate Margin.

- (c) Whenever, subsequent to the date of this Agreement, the Base Rate is increased or decreased, the interest rate for Base Rate Loans shall be similarly changed without notice or demand of any kind by an amount equal to the amount of such change in the Base Rate during the time such change or changes remain in effect. The Libor Rate shall be adjusted with respect to Libor Rate Loans without notice or demand of any kind on the effective date of any change in the Eurocurrency Reserve Percentage as of such effective date.
- (d) In the event that the Base Rate, Libor Rate or any other index that may be used to calculate interest due under the terms of this Agreement is determined to be less than zero, such index shall be deemed to be zero for purposes of this Agreement.
- (e) Upon and after the occurrence and during the continuance of an Event of Default, the Obligations (including all Letter of Credit Fees) shall bear interest at the highest applicable rate set forth in this Agreement <u>plus</u> two percent (2%) per annum.

3.2 Letter of Credit Fees.

- (a) <u>Trade or Commercial Documentary Letters of Credit</u>. With respect to each Letter of Credit that shall be a trade or commercial documentary letter of credit and the drafts thereunder, the Borrowers agree to pay to the Lender issuance, amendment, renewal, negotiation, draw, acceptance, telex, courier, postage and similar transactional fees as are customarily charged by the Lender in respect of the issuance and administration of similar letters of credit under its fee schedule as in effect from time to time.
- (b) <u>Standby Letter of Credit</u>. The Borrowers shall pay (i) to the Lender, fees for each Letter of Credit, for the period from and excluding the date of issuance of same to and including the date of expiration or termination, equal to the average daily face amount of each outstanding Letter of Credit <u>multiplied by</u> the Applicable Letter of Credit Fee Percentage, such fees to be payable monthly in arrears on the first day of each calendar month and on the Facility Termination Date, and (ii) to the Issuer, for its own account, any and all fees and expenses as agreed upon by the Issuer and the Borrowers in connection with any Letter of Credit, including in connection with the opening, amendment or renewal of any such Letter of Credit and any acceptances created thereunder and shall reimburse the Lender for any and all fees and expenses, if any, paid by the Lender to the Issuer (all of the foregoing fees, the "<u>Letter of Credit Fees</u>"). All such charges shall be deemed earned in full on the date when the same are due and payable hereunder and shall not be subject to rebate or proration upon the termination of this Agreement for any reason. Any such charge in effect at the time of a particular transaction shall be the charge for that transaction, notwithstanding any subsequent change in the Issuer's prevailing charges for that type of transaction.
- (c) <u>Fronting Fee.</u> In addition to the other fees set forth in this Section, the Borrowers shall pay to the Issuer a fronting fee, which shall accrue at the rate of 0.15% per annum on the average daily face amount of each outstanding Letter of Credit during the period from and including the issuance of each such Letter of Credit to but excluding the day on which each such Letter of Credit expires or is terminated, such fees to be payable monthly in arrears on the first day of each calendar month and on the Facility Termination Date.
- 3.3 **Unused Facility Fee.** If, for any calendar month during the term of this Agreement, the average daily Revolving Exposure for each day of such calendar month does not equal the Revolving Commitment, then the Borrowers shall pay to the Lender a fee at a rate per annum equal to the Applicable Unused Facility Fee Percentage *multiplied by* the amount by which the Revolving Commitment exceeds such average daily Revolving Exposure, such fees shall be payable to the Lender in arrears on the first

(1st) day of each calendar month after the date hereof until the Facility Termination Date and on the Facility Termination Date.

- 3.4 <u>Administration Fee.</u> The Borrowers shall pay to the Lender an administration fee in an amount equal to \$15,000 on the Closing Date and on each anniversary of the Closing Date.
- 3.5 **Upfront Fee.** The Borrowers shall pay to the Lender an upfront fee equal to Fifty Thousand Dollars (\$50,000) which is due, payable and earned in full on the Closing Date.
- 3.6 <u>Computation of Interest and Fees.</u> With the exception of Base Rate Loans, interest and fees hereunder, including Letter of Credit Fees, shall be computed on the basis of a year of 360 days and for the actual number of days elapsed. With respect to Base Rate Loans, interest shall be computed on the basis of a year having three hundred sixty-five (365) days or three hundred sixty-six (366) days, as the case may be, and calculated for the actual number of days elapsed. If any payment to be made hereunder becomes due and payable on a day other than a Business Day, the due date thereof shall be extended to the next succeeding Business Day and interest thereon shall be payable at the applicable interest rate during such extension.
- 3.7 <u>Maximum Charges</u>. In no event whatsoever shall interest and other charges charged hereunder exceed the highest rate permissible under law. In the event interest and other charges as computed hereunder would otherwise exceed the highest rate permitted under law, such excess amount shall be first applied to any unpaid principal balance owed by the Borrowers, and if the remaining excess amount is greater than the previously unpaid principal balance, the Lender shall promptly refund such excess amount to the Borrowers and the provisions hereof shall be deemed amended to provide for such permissible rate.
- 3.8 <u>Increased Costs.</u> In the event that, (a) the introduction after the Closing Date of any law, treaty, rule or regulation or any change therein after the Closing Date, (b) any change after the Closing Date in the interpretation or administration of any law, treaty, rule or regulation by any central bank or other Governmental Body or (c) the compliance by the Lender or the Issuer with any guideline, request or directive from any central bank or other Governmental Body (whether or not having the force of law) after the Closing Date (for purposes of this Section 3.8, the term "Lender" shall include the Lender and any corporation or bank controlling the Lender and the office or branch where the Lender (as so defined) makes or maintains any Libor Rate Loans), shall:
 - (a) subject the Lender to any tax of any kind whatsoever with respect to any Loan Document or change the basis of taxation of payments to the Lender of principal, fees, interest or any other amount payable under any Loan Documents (except for changes in the rate of tax on the overall net income of the Lender by the jurisdiction in which it maintains its principal office);
 - (b) impose, modify or hold applicable any reserve, special deposit, assessment or similar requirement against assets held by, or deposits in or for the account of, advances or loans by, or other credit extended by, any office of the Lender, including pursuant to Regulation D of the Board of Governors of the Federal Reserve System; or
 - (c) impose on the Lender or the London interbank offered rate market any other condition with respect to any Loan Document;

and the result of any of the foregoing is to increase the cost to the Lender of making, renewing or maintaining its Advances or any Loan hereunder by an amount that the Lender deems to be material or to reduce the amount of any payment (whether of principal, interest or otherwise) in respect of any of the Advances or any Loan by an amount that the Lender deems to be material, then, in any case the

Borrowers shall promptly pay the Lender, upon its demand, such additional amount as will compensate the Lender for such additional cost or such reduction, as the case may be, provided that the foregoing shall not apply to increased costs which are reflected in the Libor Rate. For purposes of this Section 3.8, any rules or directives concerning capital adequacy promulgated by the Bank of International Settlements pursuant to the Dodd-Frank Act or the Basel Committee on Banking Regulations and Supervisory Practices (or any successor or similar authority) under Basel III, and any rules, regulations, orders and directives adopted, promulgated or implemented in connection with any of the foregoing, regardless of the date adopted, issued, promulgated or implemented, are deemed to have been introduced and adopted after the Closing Date. The Lender shall certify the amount of such additional cost or reduced amount to the Borrower Representative, and such certification shall be presumed correct absent manifest error.

3.9 <u>Basis For Determining Interest Rate Inadequate or Unfair; Indemnity; Libor Rate</u> Unlawful.

- (a) In the event that the Lender shall have determined that:
- (i) reasonable means do not exist for ascertaining the Libor Rate applicable pursuant to Section 2.2 for any Interest Period; or
- (ii) Dollar deposits in the relevant amount and for the relevant maturity are not available in the London interbank Libor market, with respect to an outstanding Libor Rate Loan, a proposed Libor Rate Loan, or a proposed conversion of a Base Rate Loan into a Libor Rate Loan,

then the Lender shall give the Borrower Representative prompt notice of such determination. If such notice is given: (A) any such requested Libor Rate Loan shall be made as a Base Rate Loan, unless the Borrower Representative shall notify the Lender no later than 10:00 A.M. two (2) Business Days prior to the date of such proposed borrowing, that its request for such borrowing shall be cancelled or made as an unaffected type of Libor Rate Loan, (B) any Base Rate Loan or Libor Rate Loan which was to have been converted to an affected type of Libor Rate Loan shall be continued as or converted into a Base Rate Loan, or, if the Borrower Representative shall notify the Lender, no later than 10:00 A.M. two (2) Business Days prior to the proposed conversion, shall be maintained as an unaffected type of Libor Rate Loan, and (C) any outstanding affected Libor Rate Loans shall be converted into a Base Rate Loan, or, if the Borrower Representative shall notify the Lender, no later than 10:00 A.M. two (2) Business Days prior to the last Business Day of the then current Interest Period applicable to such affected Libor Rate Loan, shall be converted into an unaffected type of Libor Rate Loan, on the last Business Day of the then current Interest Period for such affected Libor Rate Loans. Until such notice has been withdrawn, the Lender shall have no obligation to make an affected type of Libor Rate Loan or maintain outstanding affected Libor Rate Loans and the Borrowers shall not have the right to convert a Base Rate Loan or an unaffected type of Libor Rate Loan into an affected type of Libor Rate Loan.

(b) The Borrowers shall indemnify the Lender and hold the Lender harmless from and against any and all losses and expenses that the Lender may sustain or incur as a consequence of any prepayment, conversion of, or any default by a Borrower in the payment of the principal of or interest on any Libor Rate Loan or failure by a Borrower to complete a borrowing of, a prepayment of or conversion of or to a Libor Rate Loan after notice thereof has been given, including any interest payable by the Lender to lenders of funds obtained by it in order to make or maintain its Libor Rate Loans hereunder.

- Notwithstanding any other provision hereof, if any applicable law, treaty, regulation or directive or any change therein or in the interpretation or application thereof shall make it unlawful for the Lender (for purposes of this Section 3.9(c), the term "Lender" shall include the Lender or the office or branch where the Lender or any corporation or bank controlling the Lender makes or maintains any Libor Rate Loans) to make or maintain its Libor Rate Loans, the obligation of the Lender to make Libor Rate Loans hereunder shall forthwith be cancelled and the Borrowers shall, if any affected Libor Rate Loans are then outstanding, promptly upon request from the Lender, either pay all such affected Libor Rate Loans or convert such affected Libor Rate Loans into loans of another type. If any such payment or conversion of any Libor Rate Loan is made on a day that is not the last day of the Interest Period applicable to such Libor Rate Loan, the Borrowers shall pay the Lender, upon the Lender's request, such amount or amounts as may be necessary to compensate the Lender for any loss or expense sustained or incurred by the Lender in respect of such Libor Rate Loan as a result of such payment or conversion, including any interest or other amounts payable by the Lender to lenders of funds obtained by the Lender in order to make or maintain such Libor Rate Loan. A certificate as to any additional amounts payable pursuant to the foregoing sentence submitted by the Lender to the Borrower Representative shall be presumed correct absent manifest error.
- Capital Adequacy. In the event that the Lender shall have determined that (a) the 3.10 introduction after the Closing Date of any law, treaty, rule or regulation or any change therein after the Closing Date, (b) any change after the Closing Date in the interpretation or administration of any law, treaty, rule or regulation by any central bank or other Governmental Body or (c) the compliance by the Lender or the Issuer with any guideline, request or directive from any central bank or other Governmental Body (whether or not having the force of law) after the Closing Date (for purposes of this Section 3.10, the term "Lender" shall include the Lender and any corporation or bank controlling the Lender and the office or branch where the Lender (as so defined) makes or maintains any Libor Rate Loans), has or would have the effect of reducing the rate of return on the Lender's capital as a consequence of its obligations hereunder to a level below that which the Lender could have achieved but for such adoption, change or compliance (taking into consideration the Lender's policies with respect to capital adequacy) by an amount deemed by the Lender to be material, then, from time to time, the Borrowers shall pay upon demand to the Lender such additional amount or amounts as will compensate the Lender for such reduction. In determining such amount or amounts, the Lender may use any reasonable averaging or attribution methods. The protection of this Section 3.10 shall be available to the Lender regardless of any possible contention of invalidity or inapplicability with respect to the applicable law, regulation or condition. For purposes of this Section 3.10, any rules or directives concerning capital adequacy promulgated by the Bank of International Settlements pursuant to the Dodd-Frank Act or the Basel Committee on Banking Regulations and Supervisory Practices (or any successor or similar authority) under Basel III, and any rules, regulations, orders and directives adopted, promulgated or implemented in connection with any of the foregoing, regardless of the date adopted, issued, promulgated or implemented, are deemed to have been introduced and adopted after the Closing Date. A certificate of the Lender setting forth such amount or amounts as shall be necessary to compensate the Lender with respect to this Section 3.10 when delivered to the Borrower Representative shall be presumed correct absent manifest error.

ARTICLE 4 COLLATERAL: GENERAL TERMS

4.1 <u>Security Interest in the Collateral</u>. To secure the prompt payment and performance of the Obligations, each Loan Party hereby grants to the Lender, for its benefit, the benefit of the Issuer, and the benefit of each of their respective Affiliates, a continuing security interest in and a pledge of all of its Collateral; <u>provided</u>, <u>however</u>, notwithstanding the foregoing or anything herein to the contrary, the

Collateral shall not include, and no Loan Party shall grant or be deemed to have granted a security interest, collateral assignment or other Lien in, any Excluded Assets. Each Loan Party shall mark its books and records as may be necessary or appropriate to evidence, protect and perfect the Lender's security interest and shall cause its financial statements to reflect such security interest. Each Loan Party shall promptly provide the Lender with written notice of all commercial tort claims, such notice to contain the case title together with the applicable court and a brief description of the claim(s). Upon delivery of each such notice, such Loan Party shall be deemed to hereby grant to the Lender a security interest and Lien in and to such commercial tort claims and all proceeds thereof. In addition, to secure such prompt payment and performance of the Obligations, each Loan Party shall also assign, pledge and grant to the Lender, for its benefit and for the benefit of the Issuer, a mortgage on its Real Property, if any.

- Perfection of Security Interest. Each Loan Party shall take all action that may be 4.2 necessary or desirable, or that the Lender may reasonably request, so as at all times to maintain the validity, perfection, enforceability and priority of the Lender's security interest in the Collateral or to enable the Lender to protect, exercise or enforce its rights hereunder and in the Collateral, including (a) immediately discharging all Liens other than Permitted Encumbrances, (b) using commercially reasonable efforts to obtain such Waivers as the Lender may reasonably request, (c) delivering to the Lender, endorsed or accompanied by such instruments of assignment as the Lender may specify, and stamping or marking, in such manner as the Lender may specify, any and all chattel paper, instruments, letters of credit and advices thereof and documents evidencing or forming a part of the Collateral, (d) entering into lockbox and other custodial arrangements satisfactory to the Lender, and (e) executing and delivering control agreements, instruments of pledge, Mortgages, notices and assignments, in each case in form and substance reasonably satisfactory to the Lender, relating to the creation, validity, perfection, maintenance or continuation of the Lender's security interest in Collateral under the Uniform Commercial Code or other applicable law. By its signature hereto, each Loan Party hereby authorizes the Lender to file against such Loan Party, one or more financing, continuation, or amendment statements pursuant to the Uniform Commercial Code to perfect Liens securing Obligations arising hereunder in form and substance satisfactory to the Lender. All charges, expenses and fees the Lender may incur in doing any of the foregoing, and any local taxes relating thereto, shall be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations, or, at the Lender's option, shall be paid to the Lender immediately upon demand.
- 4.3 <u>Disposition of Collateral</u>. Each Loan Party will safeguard and protect all Collateral for the Lender's general account and shall make no disposition thereof whether by sale, lease or otherwise except as may be otherwise permitted under this Agreement.
- 4.4 Preservation of Collateral. Following the occurrence and during the continuance of an Event of Default, in addition to the rights and remedies set forth in Section 11.1, the Lender may at any time take such steps as the Lender deems necessary to protect the Lender's interest in and to preserve the Collateral, including (a) the hiring of such security guards or the placing of other security protection measures as the Lender may deem appropriate; (b) employing and maintaining at any of any Loan Party's premises a custodian who shall have full authority to do all acts necessary to protect the Lender's interests in the Collateral; (c) leasing warehouse facilities to which the Lender may move all or part of the Collateral: and (d) using any Loan Party's owned or leased lifts, hoists, trucks and other facilities or equipment for handling or removing the Collateral. The Lender shall have, and is hereby granted, a right of ingress and egress to the places where the Collateral is located, and may proceed over and through any of any Loan Party's Real Property. Each Loan Party shall cooperate fully with all of the Lender's efforts to preserve the Collateral as permitted in the foregoing sentence and will take such actions to preserve the Collateral as the Lender may reasonably direct. The Lender is hereby authorized by the Loan Parties and the Issuer, from time to time in the Lender's sole discretion, (a) after the occurrence of a Default or an Event of Default, or (b) at any time that any of the other applicable conditions precedent set forth in

Section 8.2 hereof have not been satisfied, to make Revolving Loans to the Borrowers which the Lender, in its sole discretion, deems necessary or desirable (i) to preserve or protect the Collateral, or any portion thereof, (ii) to enhance the likelihood of, or maximize the amount of, repayment of the Advances and other Obligations, or (iii) to pay any other amount chargeable to the Borrowers pursuant to the terms of this Agreement. All of the Lender's expenses of preserving the Collateral in accordance with the foregoing, including any expenses relating to the bonding of a custodian, shall be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations.

- 4.5 Ownership of Collateral. With respect to the Collateral, at the time the Collateral becomes subject to the Lender's security interest: (a) each Loan Party shall be the sole owner of and fully authorized and able to sell, transfer, pledge and/or grant a first priority security interest in each and every item of its respective Collateral to the Lender, subject to Permitted Encumbrances; and, except for Permitted Encumbrances, the Collateral shall be free and clear of all Liens and encumbrances whatsoever; (b) each document and agreement executed by each Loan Party or delivered to the Lender in connection with this Agreement shall be true and correct in all material respects; (c) all signatures and endorsements of each Loan Party that appear on such documents and agreements shall be genuine and each Loan Party shall have full capacity to execute same; and (d) each Loan Party's Inventory shall be located as set forth on Schedule 4.5 (as such schedule may be updated from time to time) and shall not be removed from such location(s) without the prior written consent of the Lender except with respect to the sale of Inventory in the ordinary course of business and with respect to Inventory in transit from one location identified on Schedule 4.5 (as such schedule may be updated from time to time) to another location identified on Schedule 4.5.
- **Defense of the Lender's Interests.** Until (a) payment and performance in full of all of 4.6 the Obligations and (b) termination of this Agreement, the Lender's interests in the Collateral shall continue in full force and effect. Except for Permitted Dispositions, during such period no Loan Party shall, without the Lender's prior written consent, pledge, sell (except Inventory in the ordinary course of business), assign, transfer, create or suffer to exist a Lien upon or encumber or allow or suffer to be encumbered in any way except for Permitted Encumbrances, and except for sales, assignments, and transfers expressly permitted elsewhere herein, any part of the Collateral. Each Loan Party shall defend the Lender's interests in the Collateral against any and all Persons whatsoever except the holders of Permitted Encumbrances. At any time after an Event of Default has occurred and is continuing, the Lender shall have the right to take possession of the indicia of the Collateral and the Collateral in whatever physical form contained, including: labels, stationery, documents, instruments and advertising materials. If the Lender exercises such right to take possession of the Collateral, the Loan Parties shall, upon demand, assemble it in the best manner possible and make it available to the Lender at a place reasonably convenient to the Lender. In addition, with respect to all Collateral, the Lender and the Issuer shall be entitled to all of the rights and remedies set forth herein and further provided by the Uniform Commercial Code or other applicable law. After the occurrence and during the continuance of an Event of Default, each Loan Party shall, and the Lender may, at its option, instruct all suppliers, carriers, forwarders, warehousers or others receiving or holding cash, checks, Inventory, documents or instruments in which the Lender holds a security interest to deliver same to the Lender and/or subject to the Lender's order and if they shall come into any Loan Party's possession, they, and each of them, shall be held by such Loan Party in trust as the Lender's trustee, and such Loan Party will promptly deliver them to the Lender in their original form together with any necessary endorsement.
- 4.7 **Books and Records.** Each Loan Party shall (a) keep proper books of record and account in which full, true and correct entries will be made in all material respects of all dealings or transactions of or in relation to its business and affairs; (b) set up on its books accruals with respect to all Charges; and (c) on a reasonably current basis set up on its books, from its earnings, allowances against doubtful Accounts, advances and investments and all other proper accruals (including accruals for premiums, if

any, due on required payments and accruals for depreciation, obsolescence, or amortization of properties), which should be set aside from such earnings in connection with its business. All determinations pursuant to this Section 4.7 shall be made in all material respects in accordance with, or as required by, GAAP consistently applied in the opinion of such independent public accountant as shall then be regularly engaged by the Loan Parties.

- 4.8 <u>Financial Disclosure</u>. Each Loan Party hereby irrevocably authorizes and directs all accountants and auditors employed by such Loan Party at any time and promptly after the request of the Lender to exhibit and deliver to the Lender copies of any Loan Party's financial statements (if any exist at or prior to the date of such request), trial balances or other accounting records of any sort in the accountant's or auditor's possession, and to disclose to the Lender any information such accountants may have concerning such Loan Party's financial status and business operations. Each Loan Party hereby authorizes all federal, state and municipal authorities to furnish to the Lender copies of reports or examinations relating to such Loan Party, whether made by such Loan Party or otherwise; however, the Lender will attempt to obtain such information or materials directly from such Loan Party prior to obtaining such information or materials from such accountants or such authorities.
- 4.9 <u>Compliance with Laws.</u> Each Loan Party shall be in compliance in all material respects with all laws, acts, rules, regulations and orders of any Governmental Body with jurisdiction over it or the Collateral or any part thereof or to the operation of such Loan Party's business. The Collateral at all times shall be maintained in accordance with the material requirements of all insurance carriers which provide insurance with respect to the Collateral so that such insurance shall remain in full force and effect. Each Loan Party will maintain in effect and enforce policies and procedures designed to ensure compliance by the Loan Parties, their Subsidiaries and their respective directors, officers, employees and agents with Anti-Corruption Laws and applicable Sanctions.
- **Inspection of Premises; Appraisals.** At all times as the Lender deems necessary and at 4.10 reasonable times, upon reasonable notice and subject to Section 15.14 (unless and Event of Default exists, in which case at any time and without the need for notice), the Lender shall have full access to and the right to audit, check, inspect and make abstracts and copies from each Loan Party's books, records, audits, correspondence and all other papers relating to the Collateral and the operation of each Loan Party's business. In compliance with the first sentence of this Section, the Lender and its agents may enter upon any of each Loan Party's premises at any time during business hours and at any other reasonable time, and from time to time as the Lender deems necessary or desirable, for the purpose of auditing, inspecting and appraising the Collateral and any and all records pertaining thereto and the operation of such Loan Party's business. The Lender shall have the right to conduct such audits, inspections and appraisals at such times as the Lender deems reasonably necessary, in each case, at the Borrowers' expense; provided that, notwithstanding the foregoing or anything else contained herein to the contrary, the Borrowers shall be required to pay for no more than three (3) field exams and one (1) Inventory appraisal per calendar year unless an Event of Default occurs and is continuing, in which case such limitations shall not apply.
- 4.11 <u>Insurance</u>. Each Loan Party shall bear the full risk of any loss of any nature whatsoever with respect to the Collateral. At each Loan Party's own cost and expense in amounts and with carriers reasonably acceptable to the Lender, each Loan Party shall (a) keep all its insurable properties and properties in which each Loan Party has an interest insured against the hazards of fire, flood, sprinkler leakage, those hazards covered by extended coverage insurance and such other hazards, and for such amounts, as is customary in the case of companies engaged in businesses similar to such Loan Party's including business interruption insurance; (b) maintain a bond in such amounts as is customary in the case of companies engaged in businesses similar to such Loan Party insuring against larceny, embezzlement or other criminal misappropriation of insured's officers and employees who may either singly or jointly with

others at any time have access to the assets or funds of such Loan Party either directly or through authority to draw upon such funds or to direct generally the disposition of such assets; (c) maintain public and product liability insurance against claims for personal injury, death or property damage suffered by others; (d) maintain all such worker's compensation or similar insurance as may be required under the laws of any state or jurisdiction in which such Loan Party is engaged in business; and (e) furnish the Lender with (i) a status report with respect to the renewal of all such insurance no later than ten (10) days before the expiration date thereof, (ii) evidence of the maintenance of all such insurance by the renewal thereof no later than the expiration date thereof, and (iii) appropriate loss payable and additional insured endorsements in form and substance reasonably satisfactory to the Lender, naming the Lender as a coinsured and lender loss payee as its interests may appear but only with respect to all insurance coverage covering damage, loss or destruction of Collateral, and providing (A) that all proceeds thereunder covering a loss of or damage to Collateral shall be payable to the Lender, (B) no such insurance shall be affected by any act or neglect of the insured or owner of the property described in such policy, and (C) that such policy and loss payable clauses may not be cancelled, amended or terminated unless at least thirty (30) days' prior written notice is given to the Lender. The Loan Parties shall provide copies of all such insurance policies (including the appropriate lender loss payee and additional insured endorsements) within thirty (30) days after the Lender's request, however, only certificates of such insurance shall be required on the Closing Date. In the event of any loss under any insurance covering Collateral, the carriers named in such insurance policies covering Collateral hereby are directed by the Lender and the applicable Loan Party to make payment for such loss to the Lender and not to such Loan Party and the Lender jointly. If any insurance losses with respect to Collateral are paid by check, draft or other instrument payable to any Loan Party and the Lender jointly, subject to the right to rebuild or replace in other provisions of this Agreement, the Lender may endorse such Loan Party's name thereon and do such other things as the Lender may deem advisable to reduce the same to cash, and the Lender is hereby authorized to adjust and compromise claims under insurance coverage with respect to Collateral. Within ten days after the occurrence of a Material Recovery Event, the Borrower Representative shall furnish to the Lender written notice thereof. Within sixty (60) days after such Material Recovery Event, the Borrower Representative shall notify the Lender of the Borrower Representative's determination as to whether or not to replace, rebuild or restore the affected property (a "Material Recovery Determination Notice"). If the Borrowers decide not to replace, rebuild or restore such property or if the Borrowers have not delivered the Material Recovery Determination Notice within sixty (60) days after such Material Recovery Event, then the proceeds of insurance paid in connection with such Material Recovery Event, when received, shall be applied to the Obligations, in such order as the Lender in its Permitted Discretion shall determine. Notwithstanding anything in this Section 4.11 to the contrary, provided that no Event of Default has occurred and is continuing, if the Borrowers decide to replace, rebuild or restore such property, then any such replacement, rebuilding or restoration must be (a) commenced within six months of the date of the Material Recovery Event, and (b) substantially completed within twelve (12) months of such commencement date, with such casualty insurance proceeds and other net proceeds and other funds available to the Borrowers for replacement, rebuilding or restoration of such property. Any amounts of such insurance proceeds not applied to the costs of replacement or restoration shall be applied to the Obligations, in such order as the Lender in its Permitted Discretion shall determine.

- 4.12 **Failure to Pay Insurance.** If any Loan Party fails to obtain insurance as hereinabove provided, or to keep the same in force, the Lender, if the Lender so elects, may obtain such insurance and pay the premium therefor on behalf of such Loan Party, and such premium shall be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations.
- 4.13 **Payment of Taxes.** Each Loan Party will pay, when due, all taxes, assessments and other Charges lawfully levied or assessed upon such Loan Party or any of the Collateral including real and personal property taxes, assessments and Charges and all franchise, income, employment, social security benefits, withholding, and sales taxes, except those taxes, assessments or Charges to the extent

that any Loan Party has contested or disputed those taxes, assessments or Charges in good faith, by expeditious protest, administrative or judicial appeal or similar proceeding provided that any related tax Lien is stayed and sufficient reserves are established to the reasonable satisfaction of the Lender to protect the Lender's security interest in or Lien on the Collateral. If any tax by any Governmental Body is or may be imposed on or as a result of any transaction between any Loan Party and the Lender or the Issuer which the Lender or the Issuer may be required to withhold or pay or if any taxes, assessments, or other Charges remain unpaid after the date fixed for their payment, or if any claim shall be made which, in the Lender's opinion, may possibly create a valid Lien on the Collateral, the Lender may without notice to the Loan Parties pay the taxes, assessments or other Charges and each Loan Party hereby indemnifies and holds the Lender and the Issuer harmless in respect thereof, except to the extent arising from Lender's gross negligence or willful misconduct. The Lender will not pay any taxes, assessments or Charges to the extent that any Loan Party has contested or disputed those taxes, assessments or Charges in good faith, by expeditious protest, administrative or judicial appeal or similar proceeding provided that any related tax lien is stayed and sufficient reserves are established to the reasonable satisfaction of the Lender to protect the Lender's security interest in or Lien on the Collateral. The amount of any payment by the Lender under this Section 4.13 shall be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations.

4.14 **Payment of Leasehold Obligations.** Each Loan Party shall at all times pay, when and as due, its rental obligations under all leases under which it is a tenant, subject to the right to contest and withhold rental payments to the extent necessary to cure any landlord responsibilities under such leases, and shall otherwise comply, in all material respects, with all other terms of such leases and keep them in full force and effect and, at the Lender's reasonable request will provide evidence of having done so.

4.15 Accounts.

- (a) Nature of Accounts. Each of the Accounts shall be a bona fide and valid account representing a bona fide obligation incurred by the Account Debtor therein named, for a fixed sum as set forth in the invoice relating thereto (provided immaterial or unintentional invoice errors shall not be deemed to be a breach hereof) with respect to an absolute sale or lease and delivery of goods upon stated terms of a Loan Party, or work, labor or services theretofore rendered by a Loan Party as of the date each Account is created. Same shall be due and owing without dispute, setoff or counterclaim except as may be stated on the accounts receivable schedules delivered by the Loan Parties to the Lender.
- (b) <u>Solvency of Account Debtors</u>. Each Account Debtor, to the Loan Parties' knowledge, as of the date each Account is created, is and will be solvent and able to pay all Accounts on which the Account Debtor is obligated in full when due or with respect to such Account Debtor of any Loan Party who are not solvent such Loan Party has set up on its books and in its financial records bad debt reserves adequate to cover the uncollectible portion.
- (c) <u>Locations of Loan Parties</u>. Each Loan Party's state of organization and chief executive office are located at the addresses set forth on <u>Schedule 4.15(c)</u>. Until written notice is given to the Lender by the Borrower Representative of any other office at which any Loan Party keeps its records pertaining to Accounts, all such records shall be kept at such executive office.
- (d) <u>Notification of Assignment of Accounts</u>. At any time following the occurrence and during the continuance of an Event of Default, the Lender shall have the right to send notice of the assignment of, and the Lender's security interest in, the Accounts to any and all Account Debtors or any third party holding or otherwise concerned with any of the Collateral. Thereafter, the Lender shall have the sole right to collect the Accounts, take possession of the Collateral, or both. The Lender's actual collection expenses, including stationery and postage, telephone and

telegraph, secretarial and clerical expenses and the salaries of any collection personnel used for collection, may be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations.

- (e) Power of Lender to Act on Loan Parties' Behalf. The Lender shall have the right, at any time, to receive, endorse, assign and/or deliver in the name of the Lender or any Loan Party any and all checks, drafts and other instruments for the payment of money relating to the Accounts, and each Loan Party hereby waives notice of presentment, protest and nonpayment of any instrument so endorsed. During the term of this Agreement, each Loan Party hereby constitutes the Lender or the Lender's designee as such Loan Party's attorney with power (i) at any time to (A) endorse such Loan Party's name upon any notes, acceptances, checks, drafts, money orders or other evidences of payment or Collateral; and (B) send verifications of Accounts to any Account Debtor; and (ii) following the occurrence and during the continuance of an Event of Default, (A) sign such Loan Party's name on any invoice or bill of lading relating to any of the Accounts, drafts against Account Debtors, assignments and verifications of Accounts; (B) demand payment of the Accounts; (C) enforce payment of the Accounts by legal proceedings or otherwise; (D) exercise all of the Loan Parties' rights and remedies with respect to the collection of the Accounts and any other Collateral; (E) settle, adjust, compromise, extend or renew the Accounts; (F) settle, adjust or compromise any legal proceedings brought to collect Accounts; (G) prepare, file and sign such Loan Party's name on a proof of claim in bankruptcy or similar document against any Account Debtor; (H) prepare, file and sign such Loan Party's name on any notice of Lien, assignment or satisfaction of Lien or similar document in connection with the Accounts; and (I) to do all other acts and things necessary to carry out this Agreement. All acts of said attorney or designee are hereby ratified and approved, and said attorney or designee shall not be liable for any acts of omission or commission nor for any error of judgment or mistake of fact or of law absent gross negligence or willful misconduct; this power being coupled with an interest is irrevocable while any of the Obligations remain unpaid. The Lender shall have the right at any time following the occurrence and during the continuance of an Event of Default, to change the address for delivery of mail addressed to any Loan Party to such address as the Lender may designate and to receive, open and dispose of all mail addressed to any Loan Party.
- (f) No Liability. Neither the Issuer nor the Lender shall, under any circumstances or in any event whatsoever, have any liability for any error or omission or delay of any kind occurring in the settlement, collection or payment of any of the Accounts or any instrument received in payment thereof, or for any damage resulting therefrom, except arising from such Person's gross negligence or willful misconduct. The Lender is authorized and empowered to accept the return of the goods represented by any of the Accounts, without notice to or consent by any Loan Party, all without discharging or in any way affecting any Loan Party's liability hereunder.
- (g) <u>Establishment of Lockboxes, Deposit Accounts</u>. The Borrowers shall maintain with the Lender a Lockbox and any deposit accounts required by the Lender, including the Cash Concentration Account. Each such Lockbox shall be subject to the Master Agreement or lockbox or deposit account agreement that is in form and substance reasonably satisfactory to the Lender. The Borrowers shall notify all of their customers to forward collections of every kind due to the Borrowers to the Lockbox (such notices to be in form and substance satisfactory to the Lender).
- (h) <u>Processing Collections; Cash Concentration Account.</u> All collections from Account Debtors of the Borrowers sent to the Lockbox shall be deposited on a daily basis directly into the Cash Concentration Account. The Cash Concentration Account shall not be subject to any deduction, set off, banker's lien or any other right in favor of any Person. All funds

deposited into the Cash Concentration Account shall be the exclusive property of the Lender and shall be subject to the sole and exclusive control of the Lender and only to such signing authority designated from time to time by the Lender. The Borrowers shall not have control over or any interest in such funds.

Any collections received by the Borrowers shall be deemed held by the Borrowers in trust and as fiduciary for the Lender. The Borrowers agree not to commingle any such collections with any of Borrowers' other funds or property, but to hold such funds separate and apart in trust and as fiduciary for the Lender until deposit is made into the applicable deposit account or the Cash Concentration Account. The Borrowers hereby agree to deposit on a daily basis such directly received collections into the Cash Concentration Account.

- (i) <u>Adjustments</u>. No Loan Party will, without the Lender's consent, compromise or adjust any Account (or extend the time for payment thereof) or accept any returns of merchandise or grant any additional discounts, allowances or credits thereon except for those compromises, adjustments, returns, discounts, credits and allowances as have been heretofore (i) customary in the business or industry of the Borrowers, and (ii) done in the ordinary course of the Borrowers' business.
- (j) Payment of Fees. Each Loan Party shall pay to the Lender on demand all usual and customary reasonable and documented fees and expenses which the Lender incurs in connection with (i) the forwarding of Advance proceeds and (ii) the establishment and maintenance of any accounts as provided for in this Section 4.15. The Lender may, without making demand, charge all such fees and expenses to the Loan Account as a Revolving Loan of a Base Rate Loan and add them to the Obligations.
- 4.16 **Maintenance of Equipment.** Each Loan Party shall maintain its Equipment in good operating condition and repair in substantial accordance with industry standards (reasonable wear and tear excepted) and shall make all necessary replacements of and repairs thereto so that the value and operating efficiency of such Equipment shall be maintained and preserved. No Loan Party shall use or operate the Equipment in violation of any law, statute, ordinance, code, rule or regulation in any material respect.
- 4.17 **Exculpation of Liability.** Nothing herein contained shall be construed to constitute the Lender or the Issuer as any Loan Party's agent for any purpose whatsoever, nor shall the Issuer or the Lender be responsible or liable for any shortage, discrepancy, damage, loss or destruction of any part of the Collateral wherever the same may be located and regardless of the cause thereof, unless the same results from the Lender's or Issuer's gross negligence or willful misconduct. Neither the Issuer nor the Lender, whether by anything herein or in any assignment or otherwise, assume any of any Loan Party's obligations under any contract or agreement assigned to the Issuer or the Lender, and neither the Issuer nor the Lender shall be responsible in any way for the performance by any Loan Party of any of the terms and conditions thereof.

4.18 **Reseved.**

4.19 <u>Financing Statements</u>. Except for (a) the financing statements filed by the Lender, (b) those financing statements permitted to be filed hereunder, and (c) Permitted Encumbrances, no financing statement covering any of the Collateral or any proceeds thereof is on file in any public office.

4.20 Pledged Securities.

(a) <u>Schedule 5.3</u> sets forth, as of the Closing Date, a complete and accurate list of the ownership of the issued and outstanding Equity Interests of each Borrower and their respective

Subsidiaries. Each Loan Party represents and warrants that (i) all Pledged Securities owned by it have been duly authorized, validly issued, are fully paid and non-assessable, (ii) with respect to any certificates delivered to the Lender representing any Pledged Securities, either such certificates are Securities as defined in Article 8 of the Uniform Commercial Code as a result of actions by the issuer or otherwise, or, if such certificates are not Securities as defined in Article 8 of the Uniform Commercial Code, such Loan Party has so informed the Lender so that the Lender may take steps to perfect its security interest therein as a General Intangible and each Loan Party covenants to not cause such certificates to become Securities as defined in Article 8 of the Uniform Commercial Code without the Lender's prior written consent, and (iii) all such Pledged Securities held by a securities intermediary are covered by a control agreement among such Loan Party, the securities intermediary and the Lender pursuant to which the Lender has Control.

- (b) (i) None of the Pledged Securities has been issued or transferred in violation of the securities registration, securities disclosure or similar laws of any jurisdiction to which such issuance or transfer may be subject, (ii) except as set forth on Schedule 5.3, there are existing no options, warrants, calls or commitments of any character whatsoever relating to such Pledged Securities or which obligate the issuer of any Pledged Securities to issue additional Equity Interests, and (iii) no consent, approval, authorization, or other action by, and no giving of notice, filing with, any Governmental Body or any other Person is required for the pledge by such Loan Party of such Pledged Securities pursuant to this Agreement or for the exercise by the Lender of remedies in respect of the Pledged Securities, except as may be required in connection with such disposition by laws affecting the offering and sale of securities generally.
- (i) Each Borrower hereby grants a security interest in and pledges and collaterally assigns all of each Borrower's rights and title to the Pledged Securities. For the better protection of the Lender, the Borrowers shall execute appropriate transfer powers with respect to the Pledged Securities and, concurrently therewith, deposit the Pledged Securities and the aforesaid transfer powers with the Lender promptly upon Lender's request. (ii) Each Borrower authorizes the Lender, at any time after the occurrence and during the continuance of an Event of Default, to transfer the Pledged Securities into the name of the Lender or the Lender's nominee, but the Lender shall be under no duty to do so. Notwithstanding any provision or inference herein or elsewhere to the contrary, unless and until there shall have occurred and be continuing an Event of Default, the Lender shall have no right to vote the Pledged Securities, (iii) The Lender shall at all times have the rights and remedies of a secured party under the Uniform Commercial Code and Ohio law as in effect from time to time, in addition to the rights and remedies of a secured party provided elsewhere within this Agreement, the Notes or any other Loan Document, or otherwise provided in law or equity. Upon the occurrence and during the continuance of an Event of Default hereunder, the Lender, in its Permitted Discretion, may sell, assign, transfer and deliver the Pledged Securities, at any time, or from time to time. No prior notice need be given to the Borrowers or to any other Person in the case of any sale of the Pledged Securities that the Lender reasonably determines to be declining speedily in value or that is customarily sold in any securities exchange, over-the-counter market or other recognized market, but in any other case the Lender shall give the Borrower Representative no fewer than ten days prior notice of either the time and place of any public sale of the Pledged Securities or of the time after which any private sale or other intended disposition thereof is to be made. Each Borrower waives advertisement of any such sale and (except to the extent specifically required by the preceding sentence) waives notice of any kind in respect of any such sale. At any such public sale, the Lender may purchase the Pledged Securities, or any part thereof, free from any right of redemption, all of which rights each Borrower hereby waives and releases. After deducting all expenses, and after paying all claims, if any, secured by liens having precedence over this Agreement, the Lender may apply the net proceeds of each such sale to or toward the payment of

the Obligations, whether or not then due, in such order and by such division as the Lender in its Permitted Discretion may deem advisable. Any excess, to the extent permitted by law, shall be paid to the Borrowers.

- 4.21 <u>Cash Management System.</u> The Borrowers shall establish and maintain, until the payment in full of the Obligations and the termination of the Commitment, the cash management systems described below:
 - (a) <u>Lockbox</u>. On or before the Closing Date, the Borrowers shall (i) establish a Lockbox, which shall be governed by the Master Agreement, and, immediately after the Closing Date, shall request in writing and otherwise take such reasonable steps to ensure that all Account Debtors forward all collections directly to the Lockbox (if the Borrowers neglect or refuse to notify any Account Debtor to remit all such collections to the Lockbox, the Lender shall be entitled to make such notification), (ii) hold for the Lender, all checks, cash and other items of payment received by the Borrowers, and (iii) not commingle any collections with any other funds or property of the Borrowers, but will hold such funds separate and apart and as a fiduciary for the Lender until deposit is made into the Cash Concentration Account.
 - (b) <u>Cash Concentration Account.</u> On or before the Closing Date, the Borrowers shall have established a Cash Concentration Account with the Lender. All collections from sales of Inventory or from Account Debtors sent to the Lockbox shall be deposited directly on a daily basis, and in any event no later than the first Business Day after the date of receipt thereof, into the Cash Concentration Account in the identical form in which such collections were made (except for any necessary endorsements) whether by cash or check. All amounts deposited in the Cash Concentration Account from the Lockbox or any other source shall be under the sole and exclusive control of the Lender. The Borrowers shall have no control over such funds. The Cash Concentration Account shall not be subject to any deduction, set off, banker's lien or any other right in favor of any Person other than the Lender.
 - Operating Account. The Borrower Representative shall maintain, in its name, an Operating Account with the Lender, into which the Lender shall, from time to time, deposit proceeds of the Revolving Loans made to the Borrowers for use by the Loan Parties in accordance with the provisions of this Agreement. Unless otherwise agreed by the Lender and the Borrowers, any Revolving Loan requested by the Borrowers and made under this Agreement shall be deposited into the Operating Account. The Borrowers shall not accumulate or maintain cash in the Operating Account or payroll or other such accounts, as of any date of determination, in excess of checks outstanding against the Controlled Disbursement Account (or Controlled Disbursement Accounts) and other anticipated expenses and other deposit accounts approved by the Lender (such as medical benefit accounts, flexible spending accounts and automated clearing house accounts) as of that date, and amounts necessary to meet minimum balance requirements.
 - (d) <u>Controlled Disbursement Account</u>. The Loan Parties shall maintain, in the name of the Borrower Representative, a Controlled Disbursement Account with the Lender. The Borrowers may maintain more than one Controlled Disbursement Account. The Borrowers shall base its requests for Revolving Loans on, among other things, the daily balance of the Controlled Disbursement Account (or Controlled Disbursement Accounts). The Borrowers shall not, and shall not cause or permit any Loan Party, to maintain cash in any Controlled Disbursement Account, as of any date of determination, in excess of checks outstanding against such account as of that date, and amounts necessary to meet minimum balance requirements.

- (e) <u>Lockbox and Security Accounts</u>. The Lockbox established pursuant to the Master Agreement and the Cash Concentration Account, the Operating Account and the Controlled Disbursement Accounts shall be referred to in this Agreement as "Security Accounts" with all cash, checks and other similar items of payment in such accounts securing payment of the Obligations.
- (f) <u>Costs of Collection</u>. All service charges and costs related to the establishment and maintenance of the Security Accounts shall be the sole responsibility of the Borrowers, whether the same are incurred by the Lender or one of the Loan Parties. The Loan Parties hereby indemnify and hold the Lender harmless from and against any loss or damage with respect to any deposits made in the Security Accounts that are dishonored or returned for any reason, except as provided in the last sentence of this subsection (f). If any deposits are dishonored or returned unpaid for any reason, the Lender, in its Permitted Discretion, may charge the amount thereof against the Cash Concentration Account or any other Security Account or other Deposit Account of one or more of the Loan Parties. The Lender shall not be liable for any loss or damage resulting from any error, omission, failure or negligence on the part of the Lender, except losses or damages resulting from the Lender's gross negligence or willful misconduct, as determined by a final judgment of a court of competent jurisdiction.
- (g) Return of Funds. Upon the payment in full of the Obligations (other than continuing indemnification obligations) and the termination of all the commitments hereunder, (i) the Lender's security interests and other rights in funds in the Security Accounts shall terminate, (ii) all rights to such funds shall revert to the Loan Parties, as applicable, and (iii) Lender will, at the Borrowers' expense, take such steps as the Borrower Representative may reasonably request to evidence the termination of such security interests and to effect the return to the Borrowers such funds.

ARTICLE 5 REPRESENTATIONS AND WARRANTIES

Each Loan Party represents and warrants as follows:

- Agreement and the other Loan Documents to which it is a party and to perform all of its respective obligations hereunder and thereunder, as the case may be. This Agreement and the other Loan Documents to which each Loan Party is a party constitute the legal, valid and binding obligations of such Loan Party, enforceable against it in accordance with their terms, except as such enforceability may be limited by any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights generally. The execution, delivery and performance of this Agreement and of the other Loan Documents by each Loan Party has been approved by all necessary corporate action.
- 5.2 Formation and Qualification; Subsidiaries. Each Loan Party is duly incorporated or organized, as the case may be, and in good standing under the laws of the jurisdictions listed on Schedule 5.2(a) and is qualified to do business and is in good standing in the jurisdictions listed on Schedule 5.2(a) (as such Schedule may be updated from time to time) which constitute all jurisdictions in which qualification and good standing are necessary for such Loan Party to conduct its business and own its property and where the failure to so qualify could reasonably be expected to have a Material Adverse Effect. As of the Closing Date, the only Subsidiaries of each Loan Party are listed on Schedule 5.2(b).
- 5.3 <u>Officers, Directors, Shareholders, Capitalization</u>. The names and titles of all executive officers and directors of each Loan Party, as of the Closing Date, are set forth on Schedule 5.3.

<u>Schedule 5.3</u> also sets forth for each Loan Party, as of the Closing Date, the names of such Loan Party's shareholders and a description of such Person's Equity Interest in such Loan Party (including, if applicable, a listing of the share certificates and the number of shares of capital stock held by such Person). There are no outstanding subscriptions, options, warrants, calls, rights or other agreements or commitments of any nature relating to any Equity Interests in any Loan Party or any Subsidiary thereof, except as set forth on <u>Schedule 5.3</u>.

- 5.4 Governmental Approvals; No Conflicts. The transactions contemplated by this Agreement (a) do not require any consent or approval of, registration or filing with, or any other action by, any Governmental Body, except such as have been obtained or made and are in full force and effect and except for filings necessary to perfect Liens created pursuant to the Loan Documents, (b) will not violate any law applicable to any Loan Party or any of its Subsidiaries, (c) will not violate or result in a default under any indenture, agreement or other instrument binding upon any Loan Party or any of its Subsidiaries, or give rise to a right thereunder to require any payment to be made by any Loan Party or any of its Subsidiaries, and (d) will not result in the creation or imposition of any Lien on any asset of any Loan Party or any of its Subsidiaries, except Liens created pursuant to the Loan Documents.
- 5.5 <u>Tax Returns.</u> Each Loan Party has filed all material federal and state tax returns and other reports such Loan Party is required by law to file and has paid all taxes, assessments, fees and other governmental charges that are due and payable, except taxes that are being contested in good faith by appropriate proceedings and for which such Loan Party has set aside on its books adequate reserves in accordance with GAAP. The provision for taxes on the books of such Loan Party is adequate for all years not closed by applicable statutes, and for its current fiscal year, and no Loan Party has any knowledge of any deficiency or additional assessment in connection therewith not provided for on its books. No tax Liens have been filed and no claims are being asserted with respect to any material taxes.

5.6 Financial Statements.

- The monthly projected statements of income, statements of cash flow, balance (a) sheet and availability, for the twelve-month period commencing July 1, 2016, 2017 and 2018, of the Borrowers and their Subsidiaries prepared on a consolidated and consolidating basis and their projected balance sheets as of the Closing Date, all prepared in a form reasonably satisfactory to the Lender and copies of which were delivered to the Lender (the "Projections"), were prepared by a Financial Officer of the Borrower Representative, are based on underlying assumptions and estimates which provide a reasonable basis for the projections contained therein and reflect the Borrowers' judgment based on present circumstances of the most likely set of conditions and course of action for the projected period, provided that, with respect to projected financial information, the Loan Parties represent only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time delivered and, if such projected financial information was delivered prior to the Closing Date, as of the Closing Date (it being understood that projections are subject to inherent uncertainties and contingencies which may be outside the control of any Loan Party and that no assurance can be given that such projected financial information will be realized).
- (b) Each of (i) the audited consolidated and consolidating balance sheets of the Borrowers and their Subsidiaries and such other Persons described therein as of June 30, 2016, and the related statements of income, changes in stockholders' equity, and changes in cash flow for the period ended on such date, all accompanied by reports thereon containing opinions without qualification by independent certified public accountants and (ii) the consolidated and consolidating balance sheets of the Borrowers and their Subsidiaries and such other Persons described therein as of August 31, 2016, and the related statements of income, changes in

stockholders' equity, and changes in cash flow for the period ended on such date, prepared by an Authorized Officer of the Borrower Representative, copies of which have been delivered to the Lender, have been prepared in accordance with GAAP, consistently applied and present fairly in all material respects the financial condition of the Borrowers and their Subsidiaries at such date and the results of their operations for such period, subject to normal year-end audit adjustments and the absence of footnotes in the case of the statements referred to in clause (ii) above. Since June 30, 2016, there has been no material adverse change in the financial condition of the Borrowers and their Subsidiaries taken as a whole as shown on the consolidated balance sheet as of such date and no change in the aggregate value of machinery, Equipment and Real Property (if any) owned by the Borrowers and their Subsidiaries, except changes in the ordinary course of business, none of which individually or in the aggregate has had, or reasonably could be believed to cause in the future, a Material Adverse Effect.

5.7 <u>Corporate Name</u>. No Loan Party has been known by any other corporate name in the past five (5) years and does not sell Inventory under any other name, nor has any Loan Party been the surviving entity of a merger or consolidation or acquired all or substantially all of the assets of any Person during the preceding five (5) years.

5.8 **O.S.H.A.** and Environmental Compliance.

- (a) Each Loan Party has duly complied with, and (i) its facilities, business, assets, property, and Equipment, and (ii) to its knowledge, its leaseholds are in compliance in all material respects with, the provisions of the Federal Occupational Safety and Health Act, the Environmental Protection Act, RCRA and all other Environmental Laws; and, there have been no outstanding citations, notices or orders of non-compliance issued to any Loan Party or relating to its business, assets, property, leaseholds or Equipment under any such laws, rules or regulations.
- (b) Each Loan Party has been issued all required federal, state and local licenses, certificates or permits relating to all applicable Environmental Laws.
- (c) (i) There are no releases, spills, discharges, leaks or disposals (each, a "Release") of Hazardous Substances at, upon, under or within any Real Property; (ii) there are no underground storage tanks or polychlorinated biphenyls on the Real Property; (iii) to the knowledge of any Loan Party, the Real Property has not ever been used as a treatment, storage or disposal facility of Hazardous Waste and (iv) no Hazardous Substances are present on the Real Property;

except in each instance in paragraphs (a), (b), and (c) where such non-compliance, failure to obtain licenses, certificates or permits or such Releases or substances could not reasonably be expected to have a Material Adverse Effect.

5.9 Solvency; No Litigation, No Violation, ERISA.

(a) After giving effect to the transactions contemplated by this Agreement, the Loan Parties will be solvent, able to pay their debts as they mature, have capital sufficient to carry on their business and all businesses in which they are about to engage, and (i) as of the Closing Date, the fair present saleable value of their assets, calculated on a going concern basis, is in excess of the amount of their liabilities and (ii) subsequent to the Closing Date, the fair saleable value of their assets (calculated on a going concern basis) will be in excess of the amount of their liabilities.

- (b) Except as disclosed in <u>Schedule 5.9(b)</u>, no Loan Party has any pending or threatened litigation, arbitration, actions or proceedings which could reasonably be expected to have a Material Adverse Effect.
- (c) No Loan Party is in violation of any applicable statute, regulation or ordinance in any respect which could reasonably be expected to have a Material Adverse Effect, nor is any Loan Party in violation of any order of any court, Governmental Body or arbitration board or tribunal.
- (d) No Loan Party is a member of any Controlled Group that maintains or contributes to, nor does it maintain or contribute to any Plan or Multiemployer Plan.
- Patents, Trademarks, Copyrights and Licenses. All patents, patent applications, trademarks, trademark applications, service marks, service mark applications, copyrights, copyright applications, design rights, tradenames, assumed names, trade secrets and licenses owned or utilized by any Loan Party are set forth on Schedule 5.10 (as such Schedule may be updated from time to time), are valid and have been duly registered or filed with all appropriate Governmental Bodies and constitute all of the patents, trademarks, service marks, copyrights, design rights, tradenames, assumed names, trade secrets and licenses which are necessary for the operation of its business; there is no objection to or pending challenge to the validity of any such material patent, trademark, copyright, design right, tradename, trade secret or license and no Loan Party is aware of any grounds for any challenge, except as set forth in Schedule 5.10. Each material patent, patent application, patent license, trademark, trademark application, trademark license, service mark, service mark application, service mark license, design right, copyright, copyright application and copyright license owned or held by any Loan Party consists of original material or property developed by such Loan Party or was lawfully acquired by such Loan Party from the proper and lawful owner thereof. Each of such items has been maintained so as to preserve the value thereof from the date of creation or acquisition thereof. With respect to all customized software licensed by any Loan Party, such Loan Party is in possession of all source and object codes related to each piece of software or is the beneficiary of a source code escrow agreement, each such source code escrow agreement being listed on Schedule 5.10.
- 5.11 <u>Licenses and Permits</u>. Each Loan Party (a) is in compliance with and (b) has procured and is now in possession of, all material licenses or permits required by any applicable federal, state or local law or regulation for the operation of its business in each jurisdiction wherein it is now conducting or proposes to conduct business and where the failure to comply with or procure such licenses or permits would reasonably be expected to have a Material Adverse Effect.
- 5.12 **Default of Indebtedness.** No Loan Party is in default in the payment of the principal of or interest on any Indebtedness or under any instrument or agreement under or subject to which any Indebtedness has been issued, in each case in excess of the Threshold Amount, and no event has occurred under the provisions of any such instrument or agreement which with or without the lapse of time or the giving of notice, or both, constitutes or would constitute an event of default thereunder.
- 5.13 No Burdensome Restrictions; No Default. No Loan Party is subject to any restriction or party to any contract or agreement, the compliance with or the performance of which could reasonably be expected to have a Material Adverse Effect. No Loan Party has agreed or consented to cause or permit in the future (upon the happening of a contingency or otherwise) any of its property, whether now owned or hereafter acquired, to be subject to a Lien which is not a Permitted Encumbrance. No Loan Party is in default in the payment or performance of any of its contractual obligations in any material respect and no other material default has occurred thereunder.

- 5.14 **No Labor Disputes.** No Loan Party is involved in any material labor dispute and there are no strikes or walkouts or union organization of any of the Loan Party's employees threatened or in existence and no labor contract is scheduled to expire during the term of this Agreement.
- 5.15 <u>Margin Regulations</u>. No Loan Party is engaged, nor will it engage, principally or as one of its important activities, in the business of extending credit for the purpose of "purchasing" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulation U of the Board of Governors of the Federal Reserve System as now and from time to time hereafter in effect. No part of the proceeds of any Advance or any Loan will be used for "purchasing" or "carrying" "margin stock" as defined in Regulation U of such Board of Governors.
- 5.16 <u>Investment Company Act</u>. No Loan Party is an "investment company" registered or required to be registered under the Investment Company Act of 1940, as amended, nor is it controlled by such a company.
- 5.17 <u>Disclosure</u>. No representation or warranty made by any Loan Party in this Agreement or in any financial statement, report, certificate or any other document furnished in connection herewith contains any untrue statement of fact or omits to state any fact necessary to make the statements herein or therein not materially misleading. There is no fact known to any Loan Party or which reasonably should be known to such Loan Party which such Loan Party has not disclosed to the Lender in writing with respect to the transactions contemplated by this Agreement which could reasonably be expected to have a Material Adverse Effect.
- 5.18 <u>Hedging Contracts</u>. No Loan Party is a party to, nor will it be a party to, any Hedging Contract unless same provides that damages upon termination following an event of default thereunder are payable on a "two-way basis" without regard to fault on the part of either party.
- 5.19 <u>Material Business Agreements</u>. All Material Business Agreements to which any Loan Party is a party or is bound are listed on <u>Schedule 5.19</u>. No Loan Party is in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any Material Business Agreement to which it is a party.
- 5.20 <u>Application of Certain Laws and Regulations</u>. No Loan Party nor any Subsidiary of any Loan Party is subject to any statute, rule or regulation which regulates the incurrence of any Indebtedness, including statutes or regulations relative to common or interstate carriers or to the sale of electricity, gas, steam, water, telephone, telegraph or other public utility services.

5.21 Anti-Terrorism Laws.

- (a) No Loan Party nor any Affiliate of any Loan Party, is in violation in any material respect of any Anti-Terrorism Law or engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.
- (b) No Loan Party, nor any Affiliate of any Loan Party or their respective agents acting or benefiting in any capacity in connection with the Advances, the Loans or other transactions hereunder, is any of the following (each a "Blocked Person"):
 - (i) a Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224;

- (ii) a Person owned or controlled by, or acting for or on behalf of, any Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224;
- (iii) a Person with which the Lender or the Issuer is prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law;
- (iv) a Person that commits, threatens or conspires to commit or supports "terrorism" as defined in Executive Order No. 13224;
- (v) a Person that is named as a "specially designated national" on the most current list published by the U.S. Treasury Department Office of Foreign Asset Control at its official website or any replacement website or other replacement official publication of such list, or
 - (vi) a Person who is affiliated or associated with a Person listed above.

No Loan Party or, to the knowledge of any Loan Party, any of its agents acting or benefiting in any capacity in connection with the Advances, the Loans or other transactions hereunder, (i) conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Blocked Person, or (ii) deals in, or otherwise engages in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224.

5.22 Anti-Corruption Laws and Sanctions. Each Loan Party has implemented and maintains in effect policies and procedures designed to ensure compliance by the Loan Parties, their Subsidiaries and their respective directors, officers, employees and agents with Anti-Corruption Laws and applicable Sanctions, and the Loan Parties, their Subsidiaries and their respective officers and employees and to the knowledge of each Loan Party and its directors and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects and are not knowingly engaged in any activity that would reasonably be expected to result in any Loan Party being designated as a Sanctioned Person. No (a) Loan Party, Subsidiary or any of their respective directors, officers or employees, or (b) to the knowledge of any Loan Party, any agent of any Loan Party or any Subsidiary that will act in any capacity in connection with or benefit from the credit facility established hereby, is a Sanctioned Person. No Loan or Letter of Credit, use of proceeds or other transaction contemplated by this Agreement will violate any Anti-Corruption Law or applicable Sanctions.

ARTICLE 6 AFFIRMATIVE COVENANTS

Each Loan Party shall until payment in full of the Obligations and termination of this Agreement:

6.1 Conduct of Business and Maintenance of Existence and Assets.

- (a) Conduct continuously and operate actively its business according to good business practices;
 - (b) keep in full force and effect its existence; and
- (c) make all such reports and pay all such franchise and other material taxes and license fees and do all such other acts and things as may be lawfully required to maintain its rights, licenses, leases, powers and franchises under the laws of the United States or any political subdivision thereof.

- 6.2 <u>Violations</u>. Immediately notify the Lender in writing of any violation of any law, statute, regulation or ordinance of any Governmental Body, or of any agency thereof, applicable to any Loan Party or the Collateral which could reasonably be expected to have a Material Adverse Effect.
- 6.3 **Fixed Charge Coverage Ratio.**(a) Maintain a Fixed Charge Coverage Ratio (for the Loan Parties on a consolidated basis) of not less than 1.10 to 1.00 calculated as of the last day of the fiscal quarter ending on December 31, 2016, for the period equal to four (4) consecutive fiscal quarters then ending and as of the last day of each fiscal quarter thereafter for the period equal to the four (4) consecutive fiscal quarters then ending. Notwithstanding anything to the foregoing, (a) the Borrowers shall be permitted to add back \$100,000 to EBITDA for the December 31, 2016 calculation of the Fixed Charge Coverage Ratio, and (b) the Borrowers shall be permitted to add back \$50,000 to EBITDA for the March 31, 2017 calculation of the Fixed Charge Coverage Ratio.
- (b) In the event of a breach of the Fixed Charge Coverage Ratio set forth in clause (a) above, the Borrowers shall have the right to cure such breach by receiving a Shareholder Payment in the exact amount necessary to cure such breach calculated as follows: such Shareholder Payment shall be deemed an addition to EBITDA in the most recently ended fiscal quarter for the sole purpose of measuring the Fixed Charge Coverage Ratio, which must equal at least 1.10 to 1.00 after giving effect to such Shareholder Payment. If a Compliance Certificate showing a breach of the covenant set forth above is delivered (or was due to be delivered), any corresponding Shareholder Payment must be made on the date such Compliance Certificate was due. On the date Shareholder Payment is made, the Borrower Representative shall deliver to the Lender (i) written notice of such Shareholder Payment, (ii) a fully-executed Shareholder Subordination Agreement, if applicable, and (iii) a Compliance Certificate demonstrating compliance with the financial covenant after application of the Shareholder Payment. Upon receipt by the Lender of the foregoing items, the Event of Default that would have occurred shall be deemed not to have occurred. The Borrowers shall (1) only have the right to receive three (3) Shareholder Payments during the term of this Agreement and, (2) not be permitted to receive Shareholder Payments in two (2) consecutive fiscal quarters.
- 6.4 Execution of Supplemental Instruments. Execute and deliver, and will cause each Subsidiary to execute and deliver, or cause to be executed and delivered, to the Lender such documents, agreements and instruments, and will take or cause to be taken such further actions (including the filing and recording of financing statements, fixture filings, Mortgages, deeds of trust and other documents and such other actions), which may be required by law or which the Lender may, from time to time, reasonably request to carry out the terms and conditions of this Agreement and the other Loan Documents and to ensure perfection and priority of the Liens created or intended to be created, all at the expense of the Loan Parties.
- 6.5 <u>Payment of Indebtedness</u>. Pay, discharge or otherwise satisfy at or before maturity (subject, where applicable, to specified grace periods and, in the case of the trade payables, to normal payment practices) all its Indebtedness.
- 6.6 <u>Standards of Financial Statements.</u> Cause all financial statements referred to in Sections 9.5, 9.6 and 9.8 to be complete and correct in all material respects (subject, in the case of interim financial statements, to notes and normal year-end audit adjustments) and to be prepared in reasonable detail.
- 6.7 <u>Taxes.</u> Except as provided in Section 4.13, pay, and cause each Subsidiary thereof to pay, when due, all income taxes, assessments and other Charges. If any tax, assessment or other Charge by any Governmental Body creates a Lien on the Collateral which the Lender, in the exercise of its Permitted Discretion, determines is currently enforceable and neither inchoate nor stayed, the Lender may without notice to the Loan Parties pay the taxes, assessments or other Charges except as provided

otherwise in this Agreement. Any such payments shall be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations, or, at the Lender's option, shall be paid to the Lender immediately upon demand.

- 6.8 <u>Deposit Accounts</u>. Not maintain any deposit, investment, brokerage and any other account with any financial institution other than the Lender, unless such account is set forth on <u>Schedule 6.8</u>. Each such account set forth on <u>Schedule 6.8</u> shall be closed within thirty (30) days of the Closing Date. Any lockboxes related to the accounts set forth on <u>Schedule 6.8</u> shall be subject to daily transfer instructions reasonably acceptable to the Lender. Following the Closing Date, no Loan Party shall open a deposit, investment, brokerage and other account with any financial institution unless (a) it has received the prior written consent of the Lender and (b) it delivers to the Lender an account control agreement with respect to such account duly executed by the financial institution.
- 6.9 <u>Interest Rate Protection</u>. Promptly upon the Lender's request, maintain Hedging Contracts with financial institutions acceptable to the Lender on such terms and under such conditions as shall be reasonably acceptable to the Lender.

ARTICLE 7 NEGATIVE COVENANTS

No Loan Party shall until satisfaction in full of the Obligations and termination of this Agreement:

7.1 Merger, Consolidation, Acquisition and Sale of Assets.

- (a) Except for Permitted Acquisitions, enter into any merger, consolidation or other reorganization with or into any other Person or acquire all or a substantial portion of the assets or stock of any Person or permit any other Person to consolidate with or merge with it; provided that any Borrower or a Subsidiary of a Borrower may merge, consolidate or reorganize so long as the Borrower is the surviving party to such transaction.
- (b) Except for Permitted Dispositions, sell, pledge, lease, transfer or otherwise dispose of any of its properties or assets, except in the ordinary course of its business.
- 7.2 <u>Creation of Liens</u>. Create, assign, transfer or suffer to exist any Lien upon or against any of its Collateral, except Permitted Encumbrances.
- 7.3 **Guarantees.** Except for Permitted Guaranties, become liable upon the obligations of any Person by assumption, endorsement or guaranty thereof or otherwise (other than to the Lender or the Issuer) except (a) the endorsement of checks in the ordinary course of business and (b) guarantees made by a Loan Party with respect to the Obligations of another Loan Party.
- 1.4 Investments. Except for Permitted Investments, purchase, hold, acquire or invest in the obligations or stock of, or any other interest in, or make or permit to exist any investment or any other interest in (including any option, warrant or other right to acquire any of the foregoing) any Person, except (a) investments existing on the Closing Date and set forth on Schedule 7.4, (b) obligations issued or guaranteed by the United States of America or any agency thereof, (c) commercial paper with maturities of not more than one hundred eighty (180) days and a published rating of not less than A-1 or P-1 (or the equivalent rating), (d) certificates of time deposit and bankers' acceptances having maturities of not more than one hundred eighty (180) days and repurchase agreements backed by United States government securities of a commercial bank if (i) such bank has a combined capital and surplus of at least \$500,000,000, or (ii) its debt obligations, or those of a holding company of which it is a Subsidiary, are

rated not less than A (or the equivalent rating) by a nationally recognized investment rating agency, or (e) U.S. money market funds (i) rated AAA by Standard & Poors, Inc. or with an equivalent rating from Moody's Investors Service, Inc., or (ii) that invest solely in obligations issued or guaranteed by the United States of America or an agency thereof.

- 7.5 <u>Loans</u>. Except as set forth on <u>Schedule 7.5</u> and for Permitted Investments, make advances, loans or extensions of credit to any Person (other than another Loan Party), including any Subsidiary or Affiliate, except with respect to the extension of commercial trade credit in connection with the sale of Inventory in the ordinary course of its business.
- 7.6 <u>Capital Expenditures</u>. Make or incur any Capital Expenditure or commitments for Capital Expenditures (including capitalized leases) in any fiscal year in an aggregate amount for the Loan Parties on a consolidated basis in excess of \$750,000.

7.7 **Dividends and Distributions.**

- (a) Expect for (i) Permitted Stock Repurchases and (ii) repurchases of Equity Interests pursuant to terms of the Ed Map, Inc. 2015 stock option plan (and any restricted stock award agreement entered into in connection therewith) in an amount not to exceed \$150,000 in any fiscal year in the aggregate, declare, pay or make any dividend on any shares of the common stock or preferred stock or other Equity Interest, as the case may be, of any Loan Party (other than dividends or distributions payable in stock or other Equity Interest, as the case may be, or splitups, or reclassifications of its stock), or apply or otherwise distribute any of its funds, property or assets to the purchase, redemption or other retirement of any common or preferred stock or other Equity Interest, as the case may be, or of any options to purchase or acquire any such shares of common or preferred stock or other Equity Interest, as the case may be, of any Loan Party; or
- (b) Expect (i) for Permitted Stock Repurchases, (ii) pursuant to terms of the Ed Map, Inc. 2015 stock option plan and any restricted stock award agreement entered into in connection therewith, and (iii) the purchase of any Equity Interests of the Borrower Representative by any Person other than a Borrower (as long as such transaction does not result in a Change of Control) enter into or issue, as applicable, any subscriptions, options, warrants, calls, rights or other agreements or commitments of any nature relating to any Equity Interests of any Loan Party.
- 7.8 <u>Indebtedness</u>. Create, incur, assume or suffer to exist any Indebtedness except in respect
 - (a) Indebtedness existing on the Closing Date and set forth on <u>Schedule 7.8</u> (including any extensions, renewals or refinancings thereof);
 - (b) Indebtedness to the Lender and the Issuer under or pursuant to the Loan Documents;
 - (c) Indebtedness incurred to finance or reimburse (within thirty days of the respective purchase) the Borrower for Capital Expenditures permitted under Section 7.6;
 - (d) Indebtedness as permitted under Sections 7.3 and 7.11;
 - (e) Indebtedness arising from Hedging Contracts; and
 - (f) Permitted Indebtedness not otherwise described above.

{6369722:7}

of:

- 7.9 <u>Nature of Business</u>. Substantially change the nature of the business in which it is currently engaged, nor, except as specifically permitted hereby purchase or invest, directly or indirectly, in any assets or property other than in the ordinary course of business for assets or property which are useful in, necessary for and are to be used in its business, as presently conducted.
- 7.10 <u>Transactions with Affiliates.</u> Directly or indirectly, purchase, acquire or lease any property from, or sell, transfer or lease any property to, or otherwise deal with, any Affiliate, except transactions in the ordinary course of business, on an arm's length basis on terms no less favorable than terms which would have been obtainable from a Person other than an Affiliate.
- 7.11 <u>Leases</u>. Enter as lessee into any lease arrangement for real or personal property (unless capitalized and permitted under Section 7.6) if after giving effect thereto, aggregate annual rental payments for all leased property would exceed \$750,000 in any one fiscal year in the aggregate for all Loan Parties.

7.12 Subsidiaries; Partnerships.

- (a) Form any Domestic Subsidiary unless, (i) such Subsidiary expressly becomes a Borrower and becomes jointly and severally liable for the Obligations, (ii) such Borrower pledges 100% of the Equity Interest of such Subsidiary to the Lender, (iii) the Lender shall have received all documents, including organizational documents and legal opinions it may reasonably require in connection therewith and (iv) such Subsidiary grants first priority perfected Liens in its assets to the Lender; provided, however, to the extent such Subsidiary becomes a Borrower, none of such assets which become Collateral shall be included in the Borrowing Base in accordance with the terms of this Agreement until such time as the Lender makes such determination in its Permitted Discretion;
- (b) Form any Foreign Subsidiary unless the Borrower which is the parent entity of such Subsidiary pledges, in accordance with Section 4.20, up to 65% of the applicable percentage the Equity Interest owned by such Borrower of such Subsidiary to the Lender, so that such percentage of such Subsidiary's Equity Interest becomes Pledged Securities; or
- (c) Enter into any partnership, joint venture or similar agreement, except for Permitted Acquisitions and Permitted Investments.
- 7.13 <u>Management Fees.</u> Pay any management, advisory or other related fees to any Person in excess of \$50,000 in any fiscal year.
- 7.14 <u>Fiscal Year and Accounting Changes</u>. Change its fiscal year from a June 30 year-end or make any material change (a) in accounting treatment and reporting practices except as required by GAAP or (b) in tax reporting treatment except as required or permitted by law.
- 7.15 **Pledge of Credit.** Now or hereafter pledge the Lender's credit on any purchase or for any purpose whatsoever.
- 7.16 <u>Amendment of Charter Documents</u>. Amend, modify or waive any material term or provision of its Charter Documents in a manner adverse to the Lender.
- 7.17 **ERISA.** Become part of a Controlled Group that creates, maintains or becomes obligated to contribute to, or create, maintain or become obligated to contribute to any Plan or Multiemployer Plan.

- 7.18 **Prepayment of Indebtedness.** At any time, directly or indirectly, prepay any subordinated Indebtedness in excess of the amounts, if any, permitted by the applicable Subordination Agreement or repurchase, redeem, retire or otherwise acquire any subordinated Indebtedness of any Loan Party.
- 7.19 <u>Modification of Material Business Agreements.</u> Amend, waive or otherwise modify in any material respect the terms of any Material Business Agreement without the prior written consent of the Lender where such amendment or modification could reasonably be expected to have a Material Adverse Effect.
- Anti-Terrorism Laws. At any time, (a) directly or through its Affiliates and agents, conduct any business or engage in any transaction or dealing with any Blocked Person, including the making or receiving of any contribution of funds, goods or services to or for the benefit of any Blocked Person; (b) directly or through its Affiliates and agents, deal in, or otherwise engage in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224; (c) directly or through its Affiliates and agents, engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in Executive Order No. 13224, the USA Patriot Act or any other Anti-Terrorism Law or (d) fail to deliver to the Lender or the Issuer any certification or other evidence requested from time to time by any Lender or the Issuer in its sole judgment, confirming each Loan Party's compliance with this Section 7.20.

ARTICLE 8 CONDITIONS PRECEDENT

- 8.1 <u>Conditions to Initial Loans</u>. The agreement of the Lender and the Issuer, as the case may be, to make the initial Loans and other Advances requested to be made on the Closing Date is subject to the satisfaction, or waiver by the Lender and the Issuer, immediately prior to or concurrently with the making of such Loans and other Advances, of the following conditions precedent, unless waived by the Lender:
 - (a) <u>Loan Documents</u>. The Lender shall have received duly executed (i) Loan Documents and (ii) original stock certificates of each Subsidiary along with duly executed stock powers, if any, all in form and substance satisfactory to the Lender;
 - (b) <u>Collateral and Security</u>. All Collateral items required to be physically delivered to the Lender under the Loan Documents shall have been so delivered, accompanied by any appropriate instruments of transfer (or arrangements satisfactory to the Lender for such delivery shall be in place), and all taxes, fees and other charges then due and payable in connection with the execution, delivery, recording, publishing and filing of such instruments and incurrence of the Obligations and the delivery of the Loan Documents shall have been paid in full;
 - (c) <u>Lien Searches</u>. The Lender shall have received accurate and complete copies of any Lien, pending suit, title and other public record searches required by the Lender;
 - (d) <u>Filings, Registrations and Recordings</u>. Each document (including any Uniform Commercial Code financing statement) required by this Agreement, any related agreement or under law or reasonably requested by the Lender to be filed, registered or recorded in order to create, in favor of the Lender, a perfected security interest in or Lien upon the Collateral shall have been properly filed, registered or recorded in each jurisdiction in which the filing, registration or recordation thereof is so required or requested, and all actions necessary to perfect and protect the Liens of the Lender shall have been taken;

- (e) <u>Corporate Proceedings of the Loan Parties</u>. The Lender shall have received a copy of the resolutions in form and substance reasonably satisfactory to the Lender, of the Board of Directors, Managers or Members, as the case may be, of each Loan Party authorizing (i) the execution, delivery and performance of this Agreement, the Notes, and any related agreements, and (ii) the granting by such Loan Party of the security interests in and Liens upon the Collateral, in each case, certified by an Authorized Officer of such Loan Party as of the Closing Date; and, such certificate shall state that the resolutions thereby certified have not been amended, modified, revoked or rescinded as of the date of such certificate:
- (f) <u>Incumbency Certificates of the Loan Parties</u>. The Lender shall have received a certificate of the Secretary of each Loan Party, dated the Closing Date, as to the incumbency and signature of the Authorized Officers of each Loan Party executing this Agreement, any certificate or other documents to be delivered by it pursuant hereto, together with evidence of the incumbency of such Secretary;
- (g) <u>Charter Documents</u>. The Lender shall have received copies of the Charter Documents of each Loan Party, together with all amendments thereto, certified by the Secretary of State or other appropriate official of such entity's jurisdiction of formation, incorporation or organization, as the case may be (with respect to the formation documents), and by an Authorized Officer of such Loan Party (with respect to the governance documents);
- (h) <u>Good Standing</u>. The Lender shall have received copies of good standing certificates, or similar certifications, as applicable, for the Loan Parties dated not more than ten (10) days prior to the Closing Date, issued by the Secretary of State or other appropriate official of each such entity's jurisdiction of incorporation or organization, as the case may be, and each jurisdiction where the conduct of each entity's business activities or the ownership of each such entity's properties necessitates qualification;
- (i) <u>Legal Opinion</u>. The Lender shall have received the executed legal opinion of Porter Wright Morris & Arthur LLP, in form and substance reasonably satisfactory to the Lender, which shall cover such matters incident to the transactions contemplated by this Agreement, and the other Loan Documents as the Lender may reasonably require and the Loan Parties hereby authorize and direct such counsel to deliver such opinion to the Lender and the Issuer;
- (j) No Litigation. (i) No litigation, investigation or proceeding before or by any arbitrator or Governmental Body shall be continuing or threatened against any Loan Party or against the officers, directors or managers of any Loan Party, (A) in connection with the Loan Documents or any of the transactions contemplated thereby and which, in the reasonable opinion of the Lender, is deemed material or (B) which could, in the reasonable opinion of the Lender, have a Material Adverse Effect; and (ii) no injunction, writ, restraining order or other order of any nature materially adverse to any Loan Party or the conduct of its business or inconsistent with the due consummation of the transactions contemplated by the Loan Documents shall have been issued by any Governmental Body;
- (k) <u>Collateral Examination; Trade References</u>. The Lender shall have (i) completed a Collateral examination and received an Inventory appraisal, the results of which shall be satisfactory in form and substance to the Lender and the Issuer, (ii) reviewed all books and records in connection with the Collateral, and (iii) reviewed various trade references with respect to the Loan Parties, in form and substance satisfactory to the Lender;
- (l) <u>Fees</u>. The Lender shall have received all fees payable to the Lender and the Issuer on or prior to the Closing Date;

- (m) <u>Financial Statements; Projections</u>. The Lender shall have received (i) the financial statements of the Loan Parties set forth in Section 5.6, and such financial statements shall not in the reasonable judgment of the Lender, reflect any material adverse change in the consolidated financial condition of any Loan Party. The Lender shall have received a copy of the Projections described in Section 5.6 which shall be satisfactory in all respects to the Lender;
- (n) <u>Insurance</u>. The Lender shall have received, in form and substance satisfactory to the Lender, evidence that each Loan Party has the insurance required by Section 4.11, listing the Lender as lender loss payee, and additional insured, as applicable;
- (o) <u>Payment Instructions</u>. The Lender shall have received written instructions from the Borrower Representative directing the application of proceeds of the initial Loans and other Advances made pursuant to this Agreement;
- (p) <u>Collection Accounts and Cash Management Systems</u>. The Borrowers shall have delivered to the Lender and, as the case may be, established (i) the cash management system, and executed the Master Agreement, in form and substance reasonably satisfactory to Lender, and (ii) a Cash Concentration Account, Operating Account, Lockbox arrangement, and any other required accounts with the Lender for the collection or servicing of the Accounts and proceeds of the Collateral and evidence satisfactory to the Lender that the Borrowers have directed all Account Debtors to remit payments to the Lockbox, in each case satisfactory to the Lender;
- (q) <u>Customer List</u>. The Borrowers shall have delivered to the Lender a complete list of all Account Debtors of each Borrower, including but not limited to the name, address and contact information of each Account Debtor, in form and detail satisfactory to Lender;
- (r) <u>Advertising Permission Letter</u>. The Borrowers shall have delivered to Lender an advertising permission letter, in form and substance satisfactory to the Borrowers;
- (s) <u>Consents</u>. The Lender shall have received any and all Consents necessary to permit the effectuation of the transactions contemplated by the Loan Documents; and, such Consents and Waivers of such third parties as might assert claims with respect to the Collateral, as the Lender and its counsel shall deem necessary;
- (t) <u>No Adverse Material Change</u>. (i) since June 30, 2016, there shall not have occurred any event, condition or state of facts which could reasonably be expected to have a Material Adverse Effect and (ii) no representations made or information supplied to the Lender shall have been proven to be inaccurate or misleading in any material respect;
- (u) <u>Waivers</u>. The Lender shall have received Waivers satisfactory to the Lender with respect to all premises leased by the Loan Parties or at which Inventory is located;
- (v) <u>Contract Review</u>. The Lender and its counsel shall have reviewed all Material Business Agreements and such agreements shall be satisfactory in all respects to the Lender and its counsel;
- (w) <u>Existing Indebtedness</u>. The Lender shall have received (i) a payoff letter, in form and substance satisfactory to the Lender, pursuant to which any existing Indebtedness that is to be paid by initial Loans hereunder will be paid in full, and (ii) evidence satisfactory to the Lender that all necessary termination statements, satisfaction documents and any other applicable releases in connection with any existing Indebtedness and all other Liens with respect to the Loan

Parties that are not Permitted Encumbrances have been filed or arrangements satisfactory to the Lender have been made for such filing;

- (x) <u>Borrowing Base</u>; <u>Excess Availability</u>. The Lender shall have received an executed Borrowing Base Certificate from the Borrower Representative demonstrating that the Borrowing Base is sufficient in value and amount to support the Advances in the amount requested by the Borrower Representative on the Closing Date. After giving effect to the initial Advances hereunder, Excess Availability equals at least the greater of (i) \$1,250,000 or (ii) 15% of the Borrowing Base;
- (y) <u>Legal and Capital Structure</u>. The Lender shall have reviewed and shall be satisfied with the legal and capital structure of the Borrowers after the consummation of the transactions contemplated herein; and
- (z) Other. All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the transactions contemplated by this Agreement shall be satisfactory in form and substance to the Lender and its counsel.
- 8.2 <u>Conditions to Each Advance</u>. The agreement of the Lender and the Issuer to make any Advance requested to be made on any date (including the initial Loans and Advances), is subject to the satisfaction of the following conditions precedent as of the date such Advance is made:
 - (a) <u>Representations and Warranties</u>. Each of the representations and warranties made by any Loan Party in or pursuant to any Loan Document shall be true and correct in all material respects on and as of such date as if made on and as of such date.
 - (b) <u>No Default</u>. No Event of Default shall have occurred and be continuing on such date, or would exist after giving effect to the Advances requested to be made, on such date; provided, however that, the Lender, in its sole discretion, may continue to make Advances notwithstanding the existence of an Event of Default and that any Advances so made shall not be deemed a waiver of any such Event of Default.
 - (c) <u>Maximum Advances</u>. In the case of any Advances requested to be made, after giving effect thereto, the aggregate Revolving Exposure shall not exceed the Maximum Borrowing Amount.

Each request for an Advance by the Borrower Representative hereunder shall constitute a representation and warranty by the Borrowers as of the date of such Advance that the conditions contained in this subsection shall have been satisfied.

ARTICLE 9 INFORMATION AS TO THE LOAN PARTIES

Each Borrower shall, on behalf of itself and the other Loan Parties, until satisfaction in full of the Obligations and the termination of this Agreement:

9.1 <u>Disclosure of Material Matters</u>. Immediately upon learning thereof, report to the Lender all matters materially affecting the value, enforceability or collectability of any portion of the Collateral including any Lien or claim asserted against the Collateral, any loss, damage or destruction of any material portion of the Collateral, and any Loan Party's reclamation or repossession of, or the return to any Loan Party of, a material amount of goods or material claims or material disputes asserted by any Account Debtor or other obligor.

9.2 Collateral Reporting and Information.

- Borrowing Base. The Borrowers shall deliver to the Lender, as frequently as the Lender may request, but no less frequently than by 5:00 P.M. (Eastern time) on each Wednesday of each calendar week (or the next Business Day if such Wednesday is not a Business Day), a Borrowing Base Certificate (for the period ending on Friday of the week prior to the date such Borrowing Base Certificate is submitted) prepared and certified by a Financial Officer of the Borrower Representative. Such Borrowing Base Certificate shall be updated for all activity impacting the accounts receivable of the Borrowers from the date of the immediately preceding Borrowing Base Certificate to the date of such Borrowing Base Certificate; provided that the exclusions from Eligible Accounts shall only be calculated and updated on a monthly basis as described in subsection (b)(i) below. The amount of Eligible Inventory and the determination as to which accounts receivable constitute Eligible Accounts to be included on each Borrowing Base Certificate shall, absent a request from the Lender that such amounts be calculated more frequently, be the amount that is calculated and updated monthly pursuant to subsections (b) and (c) below. The Borrower Representative shall also deliver to Lender, within twenty-five (25) days after the end of each calendar month, a Borrowing Base Certificate for the calendar month just ended reconciled to the financial statements delivered pursuant to Section 9.6 for such month, prepared by a Financial Officer of the Borrower Representative.
- (b) Accounts Receivable Aging Report. The Borrowers shall deliver to the Lender an accounts receivable aging report, in form satisfactory to the Lender and signed by a Financial Officer of the Borrower Representative, (i) concurrently with the delivery of the Borrowing Base Certificate referenced in subsection (a) above, aged by the original invoice date of accounts receivable of the Borrowers, prepared as of the last day of the preceding month, reconciled to the month-end balance sheet and month-end Borrowing Base Certificate, together with the calculation of the current month-end Eligible Accounts of the Borrowers, (ii) upon the Lender's request, an aging by original invoice date of all existing accounts receivable, specifying the names, current value and dates of invoices for each Account Debtor, and (iii) that includes any other information the Lender shall reasonably request with respect to such accounts receivable and its evaluation of such reports.
- Inventory Report. The Borrowers shall deliver to the Lender a summary of Inventory, in form satisfactory to the Lender and signed by a Financial Officer of the Borrower Representative, concurrently with the delivery of the Borrowing Base Certificate referenced in subsection (a) above, based upon month-end balances reconciled to the month-end balance sheet and the month-end Borrowing Base Certificate, and accompanied by an Inventory certification, in form reasonably acceptable to the Lender and including a calculation of the Eligible Inventory of the Borrowers (the calculation of Eligible Inventory reflecting the then most recent month-end balance). The Borrowers shall deliver to the Lender, after the end of each month, Inventory records, in such detail as the Lender shall deem reasonably necessary to determine the level of Eligible Inventory. The values shown on the Inventory reports shall be at the lower of cost or market value, determined in accordance with the usual cost accounting system of the Borrowers. The Borrowers shall provide such other reports with respect to the Inventory of the Borrowers as the Lender may reasonably request from time to time.
- (d) <u>Accounts Payable Aging Report</u>. The Borrowers shall deliver to the Lender, concurrently with the delivery of the Borrowing Base Certificate referenced in subsection (a) above, in form and detail reasonably satisfactory to Lender, an aging summary of the accounts payable of the Borrowers, dated as of the last day of the preceding month.

- (e) <u>Equipment Report</u>. The Borrowers shall deliver to the Lender, as frequently as the Lender may reasonably request, an itemized schedule describing the kind, type, quality, quantity and book value of the Equipment of the Borrowers.
- (f) <u>Customer List</u>. Subject to Section 15.14, the Borrowers shall deliver to the Lender an updated customer list, concurrently with the delivery of any field audit report and upon request by any field examiner of the Lender, that sets forth all Account Debtors of the Borrowers, including but not limited to the name, address and contact information of each Account Debtor, in form and detail reasonably satisfactory to the Lender.
- (g) <u>Locations of Collateral</u>. The Borrowers shall deliver to the Lender, during each field exam, a replacement <u>Schedule 4.5</u> that sets forth each location (including third party locations) where any Loan Party conducts business or maintains any Accounts, Inventory or Equipment, in form and substance satisfactory to the Lender.
- (h) <u>Financial Information of the Loan Parties</u>. The Borrowers shall deliver to the Lender, within ten days of the written request of the Lender, such other information about the financial condition, properties and operations of any Loan Party as the Lender may from time to time reasonably request, which information shall be submitted in form and detail satisfactory to the Lender and certified by an Authorized Officer of the Borrower Representative.
- (i) <u>Delivery Through Approved Electronic Communication System.</u> Unless otherwise required by the Lender, the Borrowers shall have the option of delivering all documents and other information required to be provided to the Lender pursuant to Section 9.2(a) (Borrowing Base Certificate), Section 9.2(b) (Accounts Receivable Aging Report), Section 9.2(c) (Inventory Report), Section 9.2(d) (Accounts Payable Aging Report) and Section 9.2(e) (Equipment Report) through the Approved Electronic Communication System, and such submissions shall be completed by a Financial Officer of the Borrower Representative or any other Person acceptable to the Lender.
- 9.3 <u>Litigation</u>. Within two (2) Business Days after a Financial Officer becomes aware of the same, notify the Lender in writing of any litigation, suit or administrative proceeding affecting any Loan Party, whether or not the claim is covered by insurance, and of any suit or administrative proceeding, which in any such case could reasonably be expected to have a Material Adverse Effect.
- Material Occurrences. Within two (2) Business Days after a Financial Officer becomes aware of the same, notify the Lender in writing upon the occurrence of (a) any Event of Default or Default; (b) any default under any Material Business Agreement; (c) any event, development or circumstance whereby any financial statements or other reports furnished to the Lender fail in any material respect to present fairly, in accordance with GAAP consistently applied, the financial condition or operating results of the Loan Parties on a consolidated or consolidating basis as of the date of such statements; (d) each and every default by any Loan Party which would reasonably be expected to result in the acceleration of the maturity of any Indebtedness, in excess of the Threshold Amount, including the names and addresses of the holders of such Indebtedness with respect to which there is a default existing or with respect to which the maturity has been or could be accelerated, and the amount of such Indebtedness; and (e) any other development in the business or affairs of any Loan Party which could reasonably be expected to have a Material Adverse Effect; in each case, to the extent permitted by applicable law, describing the nature thereof and the action the Loan Parties propose to take with respect thereto.
- 9.5 <u>Annual Financial Statements</u>. Furnish the Lender within one hundred twenty (120) days after the end of each fiscal year of the Loan Parties, audited financial statements of the Loan Parties

on a consolidated and consolidating basis including statements of income and stockholders' equity and cash flow from the beginning of the current fiscal year to the end of such fiscal year and the balance sheet as at the end of such fiscal year, all prepared in accordance with GAAP applied on a basis consistent with prior practices, and in reasonable detail and reported upon without qualification by an independent certified public accounting firm selected by the Loan Parties and satisfactory to the Lender. In addition, the reports shall be accompanied by a Compliance Certificate and a copy of any management report, letter or similar writing that may have been furnished to the Borrowers by the independent public accountants in respect of the systems, operations, financial condition or properties of the Loan Parties.

- Monthly Financial Statements. Furnish the Lender within thirty (30) days after the end of each fiscal month, an unaudited balance sheet of the Loan Parties on a consolidated and consolidating basis and unaudited statements of income and stockholders' equity and cash flow of the Loan Parties on a consolidated and consolidating basis reflecting results of operations from the beginning of the fiscal year to the end of such month and for such month, prepared on a basis consistent with prior practices and complete and correct in all material respects, subject to normal and recurring year-end adjustments that individually and in the aggregate are not material to the business of the Loan Parties and setting forth in each case in comparative form the figures from the projected annual operating budget delivered pursuant to Section 9.8 covering the current fiscal year. In addition, the financial statements furnished as of the last day of each of the Borrowers' fiscal quarters shall be accompanied by a Compliance Certificate and a copy of any management report, letter or similar writing, if any, that may have been furnished to the Borrowers by the independent public accountants in respect of the systems, operations, financial condition or properties of the Loan Parties.
- Additional Information. Furnish the Lender with such additional information as the Lender shall reasonably request in order to enable the Lender to determine whether the terms, covenants, provisions and conditions of this Agreement and the Notes have been complied with by the Loan Parties including, without the necessity of any request by the Lender, (a) copies of all environmental audits and reviews, (b) at least thirty (30) days prior thereto, notice of any Loan Party's opening of any new place of business, closing of any existing place of business or a change in its legal name, and (c) within two (2) Business Days after a Financial Officer becomes aware of the same, notice of any material labor dispute to which any Loan Party may become a party, any strikes or walkouts relating to any of its plants or other facilities, and the expiration of any labor contract to which any Loan Party is a party or by which any Loan Party is bound.
- Projected Operating Budget, Availability Forecast. Furnish the Lender no later than fifteen (15) days prior to the beginning of each fiscal year of the Loan Parties, commencing with fiscal year 2017 and each fiscal year thereafter during the term of this Agreement, a month by month projected operating budget and cash flows of the Loan Parties on a consolidated and consolidating basis for such fiscal year (including an income statement and statement of cash flows for each calendar month, and a balance sheet and availability projection as at the end of each calendar month), such projections to be accompanied by a certificate signed by a Financial Officer of the Borrower Representative to the effect that such projections and forecasts have been prepared on the basis of sound financial planning practice consistent with past budgets and financial statements and that such officer has no reasonable basis to question the reasonableness of any material assumptions on which such projections and forecasts were prepared, provided that, with respect to projected financial information, the Loan Parties represent only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time delivered and, if such projected financial information was delivered prior to the Closing Date, as of the Closing Date (it being understood that projections are subject to inherent uncertainties and contingencies which may be outside the control of any Loan Party and that no assurance can be given that such projected financial information will be realized).

9.9 Notice of Suits, Adverse Events. Within two (2) Business Days after a Financial Officer becomes aware of the same, furnish the Lender with immediate notice of (a) any lapse or other termination of any Consent issued to any Loan Party by any Governmental Body or any other Person that is material to the operation of any Loan Party's business, (b) any refusal by any Governmental Body or any other Person to renew or extend any such Consent, (c) copies of any periodic or special reports filed by any Loan Party with any Governmental Body or Person, if such reports indicate any material adverse change in the business, operations, affairs or condition of any Loan Party, or if copies thereof are requested by the Lender and/or the Issuer, and (d) copies of any material notices and other communications from any Governmental Body which specifically relate to any Loan Party.

ARTICLE 10 EVENTS OF DEFAULT

The occurrence of any one or more of the following events shall constitute an "Event of Default":

- 10.1 <u>Payment of Obligations</u>. Failure by any Borrower to pay any principal or interest on the Obligations when due, whether at maturity or by reason of acceleration pursuant to the terms of this Agreement, or by required prepayment or failure to pay any other liabilities or make any other payment, fee or charge provided for in any Loan Document;
- 10.2 <u>Misrepresentations</u>. Any representation or warranty made or deemed made by any Loan Party in this Agreement or any related agreement or in any certificate, document or financial or other statement furnished at any time in connection herewith or therewith, as the case may be, shall prove to have been misleading in any material respect on the date when made or deemed to have been made;
- 10.3 <u>Failure to Furnish Information</u>. Failure by any Loan Party to (a) furnish financial information required to be provided hereunder when due, (b) furnish any additional financial information requested by the Lender within ten (10) days after such information is requested, or (c) permit the inspection of its books or records in accordance with Section 4.10;
- 10.4 <u>Liens Against Assets.</u> Issuance of a notice of Lien, levy, assessment, injunction or attachment against a material portion of any Loan Party's property which is not stayed or lifted within thirty (30) days;
- Breach of Covenants. Failure or neglect of any Loan Party to keep or observe any term, provision, condition, covenant contained in this Agreement or any other Loan Document (other than those in Sections 4.7, 4.9, 4.13, 4.14, 4.16, 4.18, 4.19, 4.20, 6.1, 6.2,6.4, 6.5, 6.6. or 6.7 hereof) or contained in any other agreement or arrangement, now or hereafter entered into, between any Loan Party and Lender; or (b) failure or neglect of any Loan Party to perform, keep or observe any other term, provision, condition, covenant contained in Sections 4.7, 4.9, 4.13, 4.14, 4.16, 4.18, 4.19, 4.20, 6.1, 6.2, 6.4, 6.5, 6.6 or 6.7 hereof and such failure shall continue for fifteen (15) days from the date on which any Loan Party knew or should have know of the occurrence of such failure or neglect.
- 10.6 **Judgment.** Any judgment or judgments are rendered or judgment liens filed against any Loan Party for an aggregate amount in excess of the Threshold Amount which within thirty (30) days of such rendering or filing is not either appealed, satisfied, stayed, discharged of record or bonded;
- 10.7 <u>Insolvency and Related Proceedings</u>. Any Loan Party or any Subsidiary thereof shall (a) apply for, consent to or suffer the appointment of, or the taking of possession by, a receiver, custodian, trustee, liquidator or similar fiduciary of itself or of all or a substantial part of its property, (b) make a general assignment for the benefit of creditors, (c) admit in writing its inability, or be generally unable to pay its debts as they become due or cease operations of its present business, (d) commence a voluntary

case under any state or federal bankruptcy laws (as now or hereafter in effect), (e) be adjudicated a bankrupt or insolvent, (f) file a petition seeking to take advantage of any other law providing for the relief of debtors, (g) acquiesce to, or fail to have dismissed, within forty-five (45) days after such petition is filed, any petition filed against it in any involuntary case under such bankruptcy laws, or (h) take any action for the purpose of effecting any of the foregoing;

- 10.8 Material Adverse Effect. Any Material Adverse Effect occurs;
- 10.9 <u>Loss of Priority Lien</u>. Any Lien created hereunder or provided for hereby or under any related agreement for any reason ceases to be or is not a valid and perfected Lien having a first priority interest:
- 10.10 <u>Breach of Material Business Agreements</u>. A default of the obligations of any Loan Party under any Material Business Agreement to which it is a party shall occur and not be cured within any applicable cure period if such default could reasonably be expected to have a Material Adverse Effect:
- 10.11 <u>Cross Default; Cross Acceleration</u>. Any Loan Party shall (a) default in any payment of principal of or interest on any Indebtedness in excess of the Threshold Amount beyond any period of grace with respect to such payment or (b) default in the observance of any other covenant, term or condition contained in any agreement or instrument pursuant to which such Indebtedness is created, secured or evidenced, if the effect of such default is to permit the acceleration of any such Indebtedness (whether or not such right shall have been waived);
 - 10.12 **Change of Control.** Any Change of Control shall occur;
- 10.13 <u>Invalidity of Loan Documents</u>. Any material provision of any Loan Document shall, for any reason, cease to be valid and binding on any Loan Party, or any Loan Party shall so claim in writing to the Lender;
- 10.14 Loss of Material Intellectual Property. (a) Any Governmental Body shall (i) revoke, terminate, suspend or adversely modify any license, permit, patent, trademark or tradename of any Loan Party, or (ii) commence proceedings to suspend, revoke, terminate or adversely modify any such license, permit, trademark, tradename or patent and such proceedings shall not be dismissed or discharged within sixty (60) days, or (iii) schedule or conduct a hearing on the renewal of any license, permit, trademark, tradename or patent necessary for the continuation of any Loan Party's business and the staff of such Governmental Body issues a report recommending the termination, revocation, suspension or material adverse modification of such license, permit, trademark, tradename or patent; or (b) any agreement which is necessary or material to the operation of any Loan Party's business shall be revoked or terminated and not replaced by a substitute acceptable to the Lender within thirty (30) days after the date of such revocation or termination, and such revocation or termination and non-replacement would reasonably be expected to have a Material Adverse Effect;
- 10.15 <u>Destruction of Collateral</u>. Any portion of the Collateral shall be seized or taken by a Governmental Body, or any Loan Party or the title and rights of any Loan Party shall have become the subject matter of litigation which might, in the reasonable opinion of the Lender, upon final determination, result in material impairment or loss of the security provided by any Loan Document;
- 10.16 **<u>Business Interruption.</u>** The operations of any Loan Party are interrupted at any time for more than seven (7) consecutive days, which interruption would reasonably be expected to have a Material Adverse Effect; or

10.17 <u>Guarantor Repudiation.</u> Any applicable guaranty, if any, of the Obligations shall fail to remain in full force or effect or any action shall be taken to discontinue or to assert the invalidity or unenforceability of such guaranty, or any guarantor of the Obligations shall fail to comply with any of the terms or provisions of the guaranty to which it is a party, or any such guarantor shall deny that it has any further liability under the guaranty to which it is a party, or shall give notice to such effect.

ARTICLE 11 LENDER'S RIGHTS AND REMEDIES AFTER DEFAULT

- Rights and Remedies. Upon the occurrence of (a) an Event of Default pursuant to 11.1 Section 10.7, all Obligations shall be immediately due and payable and this Agreement and the obligation of the Lender and the Issuer to make Advances and maintain Loans shall be deemed terminated; and (b) any other Event of Default and at any time thereafter (such Event of Default not having previously been cured), at the option of the Lender, all Obligations shall be immediately due and payable and the Lender and the Issuer shall have the right to terminate this Agreement and to terminate the obligation of the Lender and the Issuer to make Advances and maintain Loans. Upon the occurrence and during the continuance of any Event of Default, the Lender shall have the right to exercise any and all other rights and remedies provided for herein, under the Uniform Commercial Code and at law or equity generally, including the right to foreclose the security interests granted herein and to realize upon any Collateral by any available judicial procedure and/or to take, to the extent permitted by applicable law, possession of and sell any or all of the Collateral with or without judicial process. The Lender may enter any of any Loan Party's premises or other premises without legal process and without incurring liability to any Loan Party therefor, and the Lender may thereupon, or at any time thereafter, in its discretion without notice or demand, take the Collateral and remove the same to such place as the Lender may deem advisable and the Lender may require the Loan Parties to make the Collateral available to the Lender at a convenient place. With or without having the Collateral at the time or place of sale, the Lender may sell the Collateral, or any part thereof, at public or private sale, at any time or place, in one or more sales, at such price or prices, and upon such terms, either for cash, credit or future delivery, as the Lender may elect. Except as to that part of the Collateral which is perishable or threatens to decline speedily in value or is of a type customarily sold on a recognized market, the Lender shall give the Loan Parties reasonable notification of such sale or sales, it being agreed that in all events written notice mailed to the Loan Parties at least five (5) days prior to such sale or sales is reasonable notification. At any public sale the Lender or the Issuer may bid for and become the purchaser, and the Lender, the Issuer or any other purchaser at any such sale thereafter shall hold the Collateral sold absolutely free from any claim or right of whatsoever kind, including any equity of redemption and such right and equity are hereby expressly waived and released by each Loan Party. In connection with the exercise of the foregoing remedies, the Lender is granted permission to use all of each Loan Party's trademarks, trade styles, trade names, patents, patent applications, licenses, franchises and other proprietary rights which are used in connection with (y) Inventory for the purpose of disposing of such Inventory and (z) Equipment for the purpose of completing the manufacture of unfinished goods. The proceeds realized from the sale of any Collateral shall be applied as follows: first, to the reasonable and documented costs, expenses and attorneys' fees and expenses incurred by the Lender for collection and for acquisition, completion, protection, removal, storage, sale and delivery of the Collateral; second, to interest due upon any of the Obligations and any fees payable under this Agreement; and, third, to the principal of the Obligations. If any deficiency shall arise, the Loan Parties shall remain liable to the Lender and the Issuer therefor.
- 11.2 <u>Lender Discretion</u>. The Lender shall have the right in its sole discretion to determine which rights, Liens, security interests or remedies the Lender may at any time pursue, relinquish, subordinate, or modify or to take any other action with respect thereto and such determination will not in any way modify or affect any of the Lender's or the Issuer's rights hereunder.

- 11.3 <u>Setoff.</u> In addition to any other rights which the Lender or the Issuer may have under applicable law, upon the occurrence of and during the continuance of an Event of Default hereunder, the Lender and the Issuer, including any branch, Subsidiary or Affiliate of the Lender or the Issuer, shall have a right to apply any Loan Party's property, other than an Excluded Asset, held by the Lender or the Issuer, such branch, Subsidiary or Affiliate to reduce the Obligations.
- 11.4 <u>Rights and Remedies not Exclusive</u>. The enumeration of the foregoing rights and remedies is not intended to be exhaustive and the exercise of any right or remedy shall not preclude the exercise of any other right or remedies provided for herein or otherwise provided by law, all of which shall be cumulative and not alternative.
- Appointment of Receiver. Upon the occurrence and during the continuance of an Event of Default and at all times thereafter, the Lender shall be entitled to the immediate appointment of a receiver for all or any part of the Collateral, whether such receivership is incidental to a proposed sale of the Collateral, pursuant to the Uniform Commercial Code or otherwise. Each Loan Party hereby consents to the appointment of such a receiver without notice or bond, to the full extent permitted by applicable statute or law; and waives any and all notices of and defenses to such appointment and agrees not to oppose any application therefor by the Lender, but nothing herein is to be construed to deprive the Lender of any other right, remedy or privilege the Lender may have under law to have a receiver appointed, provided, however, that, the appointment of such receiver shall not impair or in any manner prejudice the rights of the Lender to receive any payments provided for herein. Such receivership shall, at the option of the Lender, continue until full payment of all of the Obligations.

ARTICLE 12 WAIVERS AND JUDICIAL PROCEEDINGS

- 12.1 <u>Waiver of Notice</u>. Each Loan Party hereby waives notice of non-payment of any of the Accounts, demand, presentment, protest and notice thereof with respect to any and all instruments, notice of acceptance hereof, Notice of Loans or advances made, credit extended, Collateral received or delivered, or any other action taken in reliance hereon, and all other demands and notices of any description, except such as are expressly provided for herein.
- 12.2 <u>Delay</u>. No delay or omission on the Lender's or the Issuer's part in exercising any right, remedy or option shall operate as a waiver of such or any other right, remedy or option or of any default.
- Jury Waiver. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION (A) ARISING UNDER THIS AGREEMENT, ANY LOAN DOCUMENT, OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION HEREWITH OR THEREWITH, OR (B) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO THIS AGREEMENT, ANY LOAN DOCUMENT, OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION HEREWITH OR THEREWITH, OR THE TRANSACTIONS RELATED HERETO OR THERETO IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT OR TORT OR OTHERWISE AND EACH PARTY HEREBY CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENTS OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

ARTICLE 13 EFFECTIVE DATE AND TERMINATION

- Term. This Agreement, which shall inure to the benefit of and shall be binding upon the respective successors and permitted assigns of each Loan Party, the Lender and the Issuer, shall become effective on the date hereof and shall continue in full force and effect until the Facility Termination Date unless sooner terminated as herein provided. Without limiting Section 11.1, (a) the Revolving Commitment shall expire on the Facility Termination Date and (b) all unpaid Obligations shall be paid in full by the Borrowers on the Facility Termination Date. The Borrowers may terminate this Agreement with at least thirty (30) Business Days' prior written notice thereof to the Lender, upon (a) the payment in full of all outstanding Loans, together with accrued and unpaid interest thereon, (b) the cancellation and return of all outstanding Letters of Credit (or alternatively, with respect to each such Letter of Credit, the furnishing to the Lender of a cash deposit as required by Section 2.9(c), (c) the payment in full of all reimbursable expenses and other Obligations together with accrued and unpaid interest thereon, and (d) the payment in full of any amount due under Section 2.11.
- **Termination.** The termination of this Agreement shall not affect any Loan Party's, the Lender's or the Issuer's rights, or any of the Obligations having their inception prior to the effective date of such termination, and the provisions hereof shall continue to be fully operative until all transactions entered into, rights or interests created or Obligations (other than contingent or unliquidated Obligations) have been fully disposed of, concluded or liquidated. The security interests, Liens and rights granted to the Lender and the Issuer hereunder and the financing statements filed hereunder shall continue in full force and effect, notwithstanding the termination of this Agreement or the fact that the Loan Account may from time to time be temporarily in a zero or credit position, until all of such Obligations of each Loan Party have been paid or performed in full after the termination of this Agreement or each Loan Party has furnished the Lender and the Issuer with an indemnification reasonably satisfactory to the Lender and the Issuer with respect thereto. Accordingly, each Loan Party waives any rights which it may have under the Uniform Commercial Code to demand the filing of termination statements with respect to the Collateral, and the Lender shall not be required to send such termination statements to each Loan Party, or to file them with any filing office, unless and until this Agreement shall have been terminated in accordance with its terms and all such Obligations paid in full in immediately available funds. All representations, warranties, covenants, waivers and agreements contained herein shall survive the termination hereof until all such Obligations are paid or performed in full. Without limitation, all indemnification obligations contained herein shall survive the termination hereof and payment in full of the Obligations.

ARTICLE 14 THE BORROWER REPRESENTATIVE

Appointment; Nature of Relationship. The Borrower Representative is hereby appointed by each of the Borrowers as its contractual representative hereunder and under each other Loan Document, and each of the Borrowers irrevocably authorizes the Borrower Representative to act as the contractual representative of such Borrower with the rights and duties set forth herein and in the other Loan Documents. The Borrower Representative agrees to act as such contractual representative. Additionally, the Borrowers hereby appoint the Borrower Representative as their agent to receive all of the proceeds of the Loans in its Operating Account, at which time the Borrower Representative shall promptly disburse such Loans to the appropriate Borrower, provided that, in the case of a Revolving Loan, such amount shall not exceed such Borrower's Borrowing Base availability. The Lender and its respective officers, directors, agents or employees, shall not be liable to the Borrower Representative or any Borrower for any action taken or omitted to be taken by the Borrower Representative or the Borrowers pursuant to this Article 14.

Joint and Several Obligations. All Obligations shall be joint and several, and each 14.2 Borrower shall make payment upon the maturity of the Obligations by acceleration or otherwise, and such obligation and liability on the part of each Borrower shall in no way be affected by any extensions, renewals and forbearance granted by the Lender or the Issuer to any Borrower, failure of the Lender or the Issuer to give any Borrower notice of borrowing or any other notice, any failure of the Lender or the Issuer to pursue or preserve its rights against any Borrower, the release by the Lender or the Issuer of any Collateral now or thereafter acquired from any Borrower, and such agreement by each Borrower to pay upon any notice issued pursuant thereto is unconditional and unaffected by prior recourse by the Lender or the Issuer to the other Borrowers or any Collateral for such Borrower's Obligations or the lack thereof. Each Borrower waives all suretyship defenses. Without limiting the generality of the foregoing, each of the Borrowers hereby acknowledges and agrees that any and all actions, inactions or omissions by any one or more, or all, of the Borrowers in connection with, related to or otherwise affecting this Agreement or any of the other Loan Documents are the Obligations of, and inure to and are binding upon, each and all of the Borrowers, jointly and severally. Each covenant, agreement, obligation, representation and warranty of the Borrowers contained herein constitutes the joint and several undertaking of each Borrower. Each Borrower acknowledges that the Obligations of such Borrower undertaken herein might be construed to consist, at least in part, of the guaranty of Obligations of the other Borrowers and, in full recognition of that fact, each Borrower consents and agrees that the Lender may, at any time and from time to time, without notice or demand, whether before or after any actual or purported termination, repudiation or revocation of this Agreement by any Borrower, and without affecting the enforceability or continuing effectiveness hereof as to such Borrower: (a) supplement, restate, modify, amend, increase, decrease, extend, renew or otherwise change the time for payment or the terms of this Agreement or any part thereof, including any increase or decrease of the rate(s) of interest thereon; (b) supplement, restate, modify, amend, increase, decrease or waive, or enter into or give any agreement, approval or consent with respect to, this Agreement or any part thereof, or any of the Loan Documents, or any condition, covenant, default, remedy, right, representation or term thereof or thereunder; (c) accept partial payments; (d) release, reconvey, terminate, waive, abandon, fail to perfect, subordinate, exchange, substitute, transfer or enforce any security or guarantees, and apply any security and direct the order or manner of sale thereof as the Lender, in its sole and absolute discretion may determine; (e) release any Person from any personal liability with respect to this Agreement or any part thereof; (f) settle, release on terms satisfactory to the Lender or by operation of applicable law or otherwise liquidate or enforce any security or guaranty in any manner, consent to the transfer of any security and bid and purchase at any sale; or (g) consent to the merger, change or any other restructuring or termination of the corporate or partnership existence of any Borrower, or any other Person, and correspondingly restructure the Obligations evidenced hereby, and any such merger, change, restructuring or termination shall not affect the liability of any Borrower or the continuing effectiveness hereof, or the enforceability hereof with respect to all or any part of the Obligations evidenced hereby. Each Borrower states and acknowledges that: (w) pursuant to this Agreement, the Borrowers desire to utilize their borrowing potential on a consolidated basis to the same extent possible as if they were merged into a single corporate entity and that this Agreement reflects the establishment of credit facilities which would not otherwise be available to such Borrower if each Borrower were not jointly and severally liable for payment of the Obligations; (x) it has determined that it will benefit specifically and materially from the advances of credit contemplated by this Agreement; (y) it is both a condition precedent to the Obligations of the Lender hereunder and a desire of the Borrowers that each Borrower execute and deliver to the Lender this Agreement; and (z) the Borrowers have requested and bargained for the structure and terms of and security for the advances contemplated by this Agreement. Each Borrower agrees if such Borrower's joint and several liability hereunder, or if any Liens securing such joint and several liability, would, but for the application of this Section 14.2, be unenforceable under applicable law, such joint and several liability and each such Lien shall be valid and enforceable to the maximum extent that would not cause such joint and several liability or such Lien to be unenforceable under applicable law, and such joint and several liability and such Lien shall be deemed to have been automatically amended accordingly at all relevant times. To the extent that any Borrower

shall, under this Agreement as a joint and several obligor, repay any of the Obligations constituting Loans or Advances made to another Borrower hereunder or other Obligations incurred directly and primarily by any other Borrower (an "Accommodation Payment"), then the Borrower making such Accommodation Payment shall be entitled to contribution and indemnification from, and, be reimbursed by, each of the other Borrowers in an amount, for each of such other Borrowers, equal to a fraction of such Accommodation Payment, the numerator of which fraction is such other Borrower's "Allocable Amount" (as defined below) and the denominator of which is the sum of the Allocable Amounts of all of the Borrowers. As of any date of determination, the "Allocable Amount" of each Borrower shall be equal to the maximum amount of liability for Accommodation Payments which could be asserted against such Borrower hereunder without (A) rendering such Borrower "insolvent" within the meaning of Section 101(31) of the Bankruptcy Code, Section 2 of the Uniform Fraudulent Transfer Act ("UFTA") or Section 2 of the Uniform Fraudulent Conveyance Act ("UFCA"), (B) leaving such Borrower with unreasonably small capital or assets, within the meaning of Section 548 of the United States Bankruptcy Code, Section 4 of the UFTA, or (C) leaving such Borrower unable to pay its debts as they become due within the meaning of Section 548 of the United States Bankruptcy Code or Section 4 of the UFTA, or Section 5 of the UFCA. All rights and claims of contribution, indemnification and reimbursement under this Section shall be subordinate in right of payment to the prior payment in full of the Obligations. The provisions of this Section shall, to the extent expressly inconsistent with any provision in any Loan Document, supersede such inconsistent provision.

- 14.3 <u>Notices.</u> Each Borrower shall promptly notify the Borrower Representative of the occurrence of any Default or Event of Default. In the event that the Borrower Representative receives such a notice, the Borrower Representative shall give prompt notice thereof to the Lender. Any notice provided to the Borrower Representative hereunder shall constitute notice to each Borrower on the date received by the Borrower Representative.
- 14.4 Execution of Loan Documents; Borrowing Base Certificate. The Borrowers hereby empower and authorize the Borrower Representative, on behalf of the Borrowers, to execute and deliver to the Lender the Loan Documents and all related agreements, certificates, documents, or instruments as shall be necessary or appropriate to effect the purposes of the Loan Documents, including without limitation, the Borrowing Base Certificates and the Compliance Certificates. Each Borrower agrees that any action taken by the Borrower Representative or the Borrowers in accordance with the terms of this Agreement or the other Loan Documents, and the exercise by the Borrower Representative of its powers set forth therein or herein, together with such other powers that are reasonably incidental thereto, shall be binding upon all of the Borrowers.
- 14.5 <u>Waivers.</u> Each Borrower expressly waives (a) any and all rights of subrogation, reimbursement, indemnity, exoneration, contribution or any other claim which such Borrower may now or hereafter have against the other Borrowers or other Person directly or contingently liable for the Obligations hereunder, or against or with respect to the other Borrowers' property (including, any property which is Collateral for the Obligations), arising from the existence or performance of this Agreement, until termination of this Agreement and repayment in full of the Obligations and (b) any defense it may otherwise have to the payment and performance of the Obligations based on any contention that its liability hereunder and under the Loan Documents is limited and not joint and several. Each Borrower acknowledges and agrees that the foregoing waivers serve as a material inducement to the agreement of the Lender and the Issuer to make the Advances and other Loans, and that the Lender and the Issuer are relying on each specific waiver and all such waivers in entering into this Agreement. The undertakings of each Borrower hereunder secure the Obligations of itself and the other Borrowers.

ARTICLE 15 MISCELLANEOUS

- 15.1 Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Ohio. Any judicial proceeding brought by or against any Loan Party with respect to any of the Obligations, this Agreement or any related agreement may be brought in any court of competent jurisdiction in the State of Ohio, United States of America, and, by execution and delivery of this Agreement, each Loan Party accepts for itself and in connection with its properties, generally and unconditionally, the non-exclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. Each Loan Party hereby waives personal service of any and all process upon it and consents that all such service of process may be made by registered mail (return receipt requested) directed to the Borrower Representative at its address set forth in Section 15.6 and service so made shall be deemed completed five (5) days after the same shall have been so deposited in the mails of the United States of America, or, at the Lender's and/or the Issuer's option, by service upon the Borrower Representative which each Loan Party irrevocably appoints as such Loan Party's agent for the purpose of accepting service within the State of Ohio. Nothing herein shall affect the right to serve process in any manner permitted by law or shall limit the right of the Lender or the Issuer to bring proceedings against any Loan Party in the courts of any other jurisdiction. Each Loan Party waives any objection to jurisdiction and venue of any action instituted hereunder and shall not assert any defense based on lack of jurisdiction or venue or based upon forum non conveniens. Any judicial proceeding by any Loan Party against the Lender or the Issuer involving, directly or indirectly, any matter or claim in any way arising out of, related to or connected with this Agreement or any related agreement, shall be brought only in a federal or state court located in the Cuyahoga County, State of Ohio.
- 15.2 Entire Understanding; Amendments. This Agreement and the other Loan Documents contain the entire understanding between each Loan Party, the Lender and the Issuer and supersedes all prior agreements and understandings, if any, relating to the subject matter hereof. Any promises, representations, warranties or guarantees not herein contained and hereinafter made shall have no force and effect unless in writing, signed by each Loan Party's, the Lender's and the Issuer's respective officers. Neither this Agreement nor any portion or provisions hereof may be changed, modified, amended, waived, supplemented, discharged, cancelled or terminated orally or by any course of dealing, or in any manner other than by an agreement in writing, signed by the party to be charged. Each Loan Party acknowledges that it has been advised by counsel in connection with the execution of the Loan Documents and is not relying upon oral representations or statements inconsistent with the terms and provisions of this Agreement.

15.3 Transfers and Assignments.

- (a) <u>The Loan Parties</u>. The Loan Parties may not assign or otherwise transfer any of their rights or Obligations hereunder without the prior written consent of the Lender.
- (b) <u>Transfers and Assignments by the Lender</u>. The Lender shall have the right to assign its rights and obligations under this Agreement and the other Loan Documents and, in any way, in its sole discretion to participate with other financial institutions in Loans or in making Revolving Loans available to the Borrowers; all of the rights, privileges, remedies and options given to the Lender hereunder shall inure to the benefit of the Lender's successor's and assigns; and all the terms, conditions, promises, covenants, provisions and warranties of this Agreement and the other Loan Documents shall inure to the benefit of and shall bind the representatives, successors and assigns of the Lender.

- (c) <u>Transfers and Assignments by the Issuer</u>. The Issuer shall have the right to assign its rights and obligations under this Agreement and, in any way, in its sole discretion to participate with other financial institutions in issuing Letters of Credit; all of the rights, privileges, remedies and options given to the Issuer hereunder shall inure to the benefit of the Issuer's successor's and assigns; and all the terms, conditions, promises, covenants, provisions and warranties of this Agreement shall inure to the benefit of and shall bind the representatives, successors and assigns of the Issuer.
- (d) <u>Maintenance of Register</u>. The Lender, acting solely for this purpose as an agent of the Borrowers, shall maintain at its office in Cleveland, Ohio, a copy of a register for the recordation of the names and addresses of each party to a transfer under this Section 15.3, and the commitments to make Advances and other Loans hereunder of, and principal amounts of the Loans owing to, the Lender and such other Persons pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive, and the Loan Parties and the Lender may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrowers, the Lender or any such Person, at any reasonable time and from time to time upon reasonable prior notice.
- (e) <u>Pledge of Interests</u>. The Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of the Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release the Lender from any of its obligations hereunder or substitute any such pledgee or assignee for the Lender as a party hereto.
- (f) Notes. The Borrowers shall execute and deliver: (i) to the Lender, the transferor and the transferee, any consent or release (of all or a portion of the obligations of the transferor) to be delivered in connection with each assignment or other transfer under this Section 15.3, (ii) if the Lender's entire interest in its commitments to make Advances and other Loans hereunder and all of its Advances and other Loans have been transferred to the transferee, an appropriate replacement note against return of the Notes (marked "replaced") held by the transferor and (iii) if only a portion of the Lender's interest in its commitments to make Advances and other Loans hereunder and such Advances and other Loans have been transferred, replacement notes to each of the transferor and the transferee against return of the original such Notes of the transferor (each marked "replaced") held by the transferor; provided, that, simultaneously with the Borrowers' delivery of new Notes pursuant to this Section 15.3(f), the transferor Lender will deliver to the Borrowers any Note being replaced in whole or in part, and each such Note delivered by the Lender shall be conspicuously marked "replaced" when so delivered.
- Application of Payments. The Lender shall have the continuing and exclusive right to apply or reverse and re-apply any payment and any and all proceeds of Collateral to any portion of the Obligations in such order as the Lender determines in its Permitted Discretion. To the extent that any Loan Party makes a payment or the Lender or the Issuer receives any payment or proceeds of the Collateral for any Loan Party's benefit, which are subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to a trustee, debtor in possession, receiver, custodian or any other party under any bankruptcy law, common law or equitable cause, then, to such extent, the Obligations or part thereof intended to be satisfied shall be revived and continue as if such payment or proceeds had not been received by the Lender or the Issuer.
- 15.5 <u>Indemnity</u>. Each Loan Party shall indemnify the Lender, the Issuer and each of their respective officers, directors, attorneys, representatives, Affiliates, employees and agents from and against

any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, reasonable and documented costs, expenses and disbursements of any kind or nature whatsoever (including fees and disbursements of counsel) which may be imposed on, incurred by, or asserted against the Lender or the Issuer in any litigation, proceeding or investigation with respect to any aspect of, or any transaction contemplated by, or referred to in, or any matter related to, the Loan Documents, whether or not the Lender or the Issuer is a party thereto, except losses or damages resulting from the Lender's or the Issuer's gross negligence or willful misconduct.

- 15.6 Notice. Any notice or request hereunder may be given to the Borrower Representative or any Loan Party or to the Lender or the Issuer at their respective addresses set forth below or at such other address as may hereafter be specified in a notice designated as a notice of change of address under this Section. Any notice, request, demand, direction or other communication (for purposes of this Section 15.6 only, a "Notice") to be given to or made upon any party hereto under any provision of this Agreement shall be given or made in writing (which includes by means of electronic transmission (i.e., "e-mail") or facsimile transmission). Any such Notice must be delivered to the applicable parties hereto at the addresses and numbers set forth under their respective names on Section 15.6 or in accordance with any subsequent unrevoked Notice from any such party that is given in accordance with this Section 15.6. Any notice provided to the Borrower Representative shall be deemed to have been given to each other Loan Party. Any Notice shall be effective:
 - (a) In the case of hand-delivery, when delivered;
 - (b) If given by mail, four (4) days after such Notice is deposited with the United States Postal Service, with first-class postage prepaid, return receipt requested;
 - (c) In the case of a facsimile transmission, when sent to the applicable party's facsimile machine's telephone number, if the party sending such Notice receives confirmation of the delivery thereof from its own facsimile machine;
 - (d) In the case of electronic transmission, when actually received;
 - (e) If given by any other means (including by overnight courier), when actually received; and
 - (f) When the Lender gives a Notice to the Borrower Representative or any Loan Party, the Lender shall concurrently send a copy thereof to the Issuer. When the Issuer gives a Notice to the Borrower Representative or any Loan Party, the Issuer shall concurrently send a copy thereof to the Lender.

(i) If to the Lender at: KeyBank National Association

127 Public Square Cleveland, Ohio 44114

Attention: KeyBank Business Capital

Telephone: 216.689-4307 Facsimile: 216.689-8470

Email: andrew d blickensderfer@keybank.com

(ii) If to the Borrower

Representative at: Ed Map, Inc.

296 Harper Street Nelsonville, OH 45764 Attention: Gregory P. Smith Telephone: 740-342-0523 Facsimile: 740-757-9402 Email: gsmith@edmap.com

with a copy to: Porter, Wright, Morris & Arthur LLP

41 South High St. Columbus, OH 43215

Attention: Richard C. McQuown

Telephone: 614-227-1975 Facsimile: 614-227-2100

Email: rmcquown@porterwright.com

- 15.7 <u>Survival</u>. The obligations of the Loan Parties under Sections 2.9, 2.11, 3.8, 3.9, 3.10 and 15.5 shall survive termination of the Loan Documents and payment in full of the Obligations.
- 15.8 <u>Severability</u>. If any part of this Agreement is contrary to, prohibited by, or deemed invalid under applicable laws or regulations, such provision shall be inapplicable and deemed omitted to the extent so contrary, prohibited or invalid, but the remainder hereof shall not be invalidated thereby and shall be given effect so far as possible.
- 15.9 All reasonable and documented costs, expenses, including reasonable Expenses. attorneys' fees (including such allocated costs of in house counsel), and disbursements incurred by the Lender on its behalf or on behalf of the Issuer (a) in all efforts made to enforce payment of any Obligation or effect collection of any Collateral, (b) in connection with the entering into, modification, amendment, administration and enforcement of this Agreement or any consents or waivers hereunder and all related agreements, documents and instruments, (c) in instituting, maintaining, preserving, enforcing and foreclosing on the Lender's security interest in or Lien on any of the Collateral, whether through judicial proceedings or otherwise. (d) in defending or prosecuting any actions or proceedings arising out of or relating to the Lender's or the Issuer's transactions with any Loan Party, or (e) in connection with any advice given to the Lender or the Issuer with respect to its rights and obligations under this Agreement and all related agreements, may be charged to the Loan Account as a Revolving Loan of a Base Rate Loan and added to the Obligations. Expenses being reimbursed by the Borrowers under this Section include such costs and expenses incurred in connection with: (u) appraisals and insurance reviews; (v) field examinations and the preparation of reports based on the fees charged by a third party retained by the Lender or the internally allocated fees for each Person employed by the Lender with respect to each field examination; (w) background checks regarding senior management and/or key investors, as deemed necessary or appropriate in the Permitted Discretion of the Lender; (x) taxes, fees and other charges for (i) lien and title searches and title insurance and (ii) recording any Mortgages, filing financing statements and continuations, and other actions to perfect, protect, and continue the Lender's Liens; (y) sums paid or incurred to take any action required of any Loan Party under the Loan Documents that such Loan Party fails to pay or take; and (z) forwarding loan proceeds, collecting checks and other items of payment, and establishing and maintaining the accounts and lock boxes, and such costs and expenses of preserving and protecting the Collateral.
- 15.10 <u>Injunctive Relief.</u> Each Loan Party recognizes that, in the event any Loan Party fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, any remedy at law

may prove to be inadequate relief to the Lender and/or the Issuer; therefore, the Lender, if the Lender so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving that actual damages are not an adequate remedy.

- 15.11 <u>Consequential Damages.</u> Neither the Lender, nor the Issuer, nor any agent or attorney for any of them, shall be liable to any Loan Party for any special, incidental, consequential or punitive damages arising from any breach of contract, tort or other wrong relating to the establishment, administration or collection of the Obligations.
- 15.12 <u>Counterparts; Electronic Signatures.</u> This Agreement may be executed in any number of and by different parties hereto on separate counterparts, all of which, when so executed, shall be deemed an original, but all such counterparts shall constitute one and the same agreement. Any signature delivered by a party by facsimile or email transmission shall be deemed to be an original signature hereto.
- 15.13 <u>Construction</u>. The parties acknowledge that each party and its counsel have reviewed this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments, schedules or exhibits hereto.

15.14 Confidentiality; Sharing Information.

- (a) The Lender, the Issuer and each transferee of the Lender or the Issuer pursuant to Section 15.3 (a "<u>Transferee</u>") shall hold all non-public information obtained by the Lender, the Issuer or such Transferee confidential, in accordance with the Lender's, the Issuer's and such Transferee's customary procedures and equivalent to the efforts the Lender applies to maintain the confidentiality of its own confidential information for handling confidential information of this nature; provided, however, the Lender, the Issuer and such Transferee may disclose such confidential information (a) to its examiners, affiliates, outside auditors, counsel and other professional advisors, (b) to the Lender, the Issuer and such Transferee and (c) as required or requested by any Governmental Body or representative thereof or pursuant to legal process.
- (b) Each Loan Party acknowledges that from time to time financial advisory, investment banking and other services may be offered or provided to such Loan Party (in connection with this Agreement or otherwise) by the Lender, the Issuer or by one or more Subsidiaries or Affiliates of the Lender or the Issuer and each Loan Party hereby authorizes the Lender and the Issuer to share any information delivered to the Lender or the Issuer by such Loan Party and its Subsidiaries pursuant to this Agreement, or in connection with the decision of the Lender or the Issuer to enter into this Agreement, to any such Subsidiary or Affiliate of the Lender or the Issuer, it being understood that any such Subsidiary or Affiliate of the Lender or the Issuer receiving such information shall be bound by the provisions of Section 15.14 as if it were the Lender or the Issuer, as the case may be, hereunder. Such authorization shall survive the repayment of the other Obligations and the termination of this Agreement.

15.15 **Approved Electronic Communication System.**

(a) Unless otherwise specifically identified therein, each posting to an Approved Electronic Communication System shall be deemed to be a representation and warranty by the Borrowers, the Authorized Officer of any such Borrower submitting the information to the Approved Electronic Communication System and, if such Authorized Officer is not a Financial Officer, the Financial Officer who authorized such Authorized Officer to submit such information, as of the date of such posting, of the accuracy of the information provided with respect thereto, and that each of the representations and warranties contained in this Agreement

and the other Loan Documents are true and correct as if made on and as of the date of such posting.

- (b) Although the Approved Electronic Communication System is secured with generally-applicable security procedures and policies implemented or modified from time to time, the Borrowers and each other Loan Party acknowledge and agree that the distribution of material through an electronic medium is not necessarily secure and that there are confidentiality and other risks associated with such distribution. In consideration for the convenience and other benefits afforded by such distribution and for the other consideration provided hereunder, the receipt and sufficiency of which is hereby acknowledged, the Borrowers and each other Loan Party hereby approves of the use of the Approved Electronic Communication System and understands and assumes the risks of using such forms of communication.
- (c) The Approved Electronic Communication System is provided "as is" and "as available". Neither the Lender nor any of the Lender's Affiliates, officers, directors, attorneys, agents or employees warrant the accuracy, adequacy or completeness of the Approved Electronic Communication System and each expressly disclaims any liability for errors or omissions in the Approved Electronic Communication System. No warranty of any kind, express, implied or statutory, including, without limitation, any warranty of merchantability, fitness for a particular purpose, non-infringement of third party rights or freedom from viruses or other code defects, is made by the Lender (or any of the Lender's Affiliates, officers, directors, attorneys, agents or employees) in connection with the Approved Electronic Communication System.
- (d) The Borrowers and each other Loan Party agree that the Lender may, but shall not be obligated to, store information provided through the Approved Electronic Communication System in accordance with the Lender's generally-applicable document retention procedures and policies in effect from time to time.
- 15.16 <u>CONFESSION OF JUDGMENT.</u> THE LOAN PARTIES HEREBY AUTHORIZE ANY ATTORNEY-AT-LAW TO APPEAR IN ANY COURT OF RECORD IN ANY COUNTY IN THE STATE OF OHIO OR ELSEWHERE WHERE A LOAN PARTY HAS A PLACE OF BUSINESS, SIGNED THIS AGREEMENT OR CAN BE FOUND, AFTER THE LENDER DECLARES A DEFAULT AND ACCELERATES THE BALANCES DUE UNDER THIS AGREEMENT, TO WAIVE THE ISSUANCE OF SERVICE OF PROCESS AND CONFESS JUDGMENT AGAINST THE LOAN PARTIES IN FAVOR OF THE LENDER FOR THE AMOUNTS THEN APPEARING DUE, TOGETHER WITH THE COSTS OF SUIT, AND THEREUPON TO RELEASE ALL ERRORS AND WAIVE ALL RIGHT OF APPEAL AND THE LOAN PARTIES AGREE AND CONSENT THAT THE STAY OF EXECUTION. ATTORNEY CONFESSING JUDGMENT ON BEHALF OF THE LOAN PARTIES HEREUNDER MAY ALSO BE COUNSEL TO THE LENDER OR ANY OF ITS AFFILIATES, WAIVES ANY CONFLICT OF INTEREST WHICH MIGHT OTHERWISE ARISE, AND CONSENTS TO THE LENDER PAYING SUCH CONFESSING ATTORNEY A LEGAL FEE OR ALLOWING SUCH ATTORNEY'S FEES TO BE PAID FROM ANY PROCEEDS OF COLLECTION OF THIS AGREEMENT OR COLLATERAL SECURITY THEREFOR.

[Remainder of Page Intentionally Left Blank]

IN WITNESS THEREOF, the Loan Parties, the Lender and the Issuer have caused this Agreement to be executed and delivered as of the date first written above.

BORROWER:

WARNING -- BY SIGNING THIS PAPER YOU GIVE UP YOUR RIGHT TO NOTICE AND COURT TRIAL. IF YOU DO NOT PAY ON TIME A COURT JUDGMENT MAY BE TAKEN AGAINST YOU WITHOUT YOUR PRIOR KNOWLEDGE AND THE POWERS OF A COURT CAN BE USED TO COLLECT FROM YOU REGARDLESS OF ANY CLAIMS YOU MAY HAVE AGAINST THE CREDITOR WHETHER FOR RETURNED GOODS, FAULTY GOODS, FAILURE ON HIS PART TO COMPLY WITH THE AGREEMENT, OR ANY OTHER CAUSE.

ED MAP, INC., a Delaware corporation

By: ____

Name: Gregory P. Smith Title: Chief Financial Officer

LENDER and ISSUER:

KEYBANK NATIONAL ASSOCIATION,

a national banking association

By: ___

Name: Andrew Blickensderfer

Title: Vice President

IN WITNESS THEREOF, the Loan Parties, the Lender and the Issuer have caused this Agreement to be executed and delivered as of the date first written above.

BORROWER:

WARNING -- BY SIGNING THIS PAPER YOU GIVE UP YOUR RIGHT TO NOTICE AND COURT TRIAL. IF YOU DO NOT PAY ON TIME A COURT JUDGMENT MAY BE TAKEN AGAINST YOU WITHOUT YOUR PRIOR KNOWLEDGE AND THE POWERS OF A COURT CAN BE USED TO COLLECT FROM YOU REGARDLESS OF ANY CLAIMS YOU MAY HAVE AGAINST THE CREDITOR WHETHER FOR RETURNED GOODS, FAULTY GOODS, FAILURE ON HIS PART TO COMPLY WITH THE AGREEMENT, OR ANY OTHER CAUSE.

ED MAP, INC., a Delaware corporation

LENDER and ISSUER:

KEYBANK NATIONAL ASSOCIATION, a national banking association

Name: Andrew Blickensderfer

Title: Vice President

DISCLOSURE SCHEDULES TO CREDIT AND SECURITY AGREEMENT DATED AS OF NOVEMBER 22, 2016

INTRODUCTION:

For purposes of this introduction, the term "Schedules" shall include the following:

Schedule 1.2(a)	Owned Real Property
Schedule 1.2(b)	Liens
Schedule 4.5	Inventory
Schedule 4.15(c)	Loan Parties' States of Organization and Chief Executive Offices
Schedule 5.2 (a)	Incorporation/Organization/Foreign Qualification
Schedule 5.2 (b)	Subsidiaries
Schedule 5.3	Officers, Directors, Shareholders, Capitalization
Schedule 5.9(b)	Litigation
Schedule 5.10	Patents, Trademarks, Copyrights and Licenses
Schedule 5.19	Material Business Agreements
Schedule 6.8	Accounts
Schedule 7.4	Investments
Schedule 7.5	Loans
Schedule 7.8	Indebtedness

Unless otherwise defined in these Schedules, all capitalized terms used herein shall have the meanings ascribed to them in the Credit and Security Agreement by and between Ed Map, Inc. a Delaware C-corporation as Borrower, and KeyBank National Association as Lender and Issuer dated as of November 22, 2016 (the "Agreement").

Matters reflected in these Schedules are not necessarily limited to matters required by the Agreement to be reflected in the Schedules. Such additional matters are set forth for informational purposes and do not necessarily include other matters of a similar nature. In no event shall the listing of such matters in these Schedules be deemed or interpreted to broaden or otherwise amplify the Borrower's representations, warranties, covenants or agreements contained in the Agreement.

The headings contained in these Schedules are for reference purposes only and shall not affect in any way the meaning or interpretation of the Agreement or these Schedules. A disclosure with respect to any one matter contained herein shall be deemed a disclosure with respect to all other matters.

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 96 of 130

Schedule 1.2(a) Owned Real Property

None.

Schedule 1.2(b) Liens

None.

Schedule 4.5 Inventory

294 and 296 Harper Street Nelsonville, Ohio 45764

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 99 of 130

Schedule 4.15(c)
Loan Parties' States of Organization and Chief Executive Offices

Ed Map, Inc.'s state of incorporation is Delaware, and its chief executive office is located at 296 Harper Street, Nelsonville, Ohio 45764.

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 100 of 130

Schedule 5.2(a) Incorporation/Organization/Foreign Qualification

Ed Map, Inc. is duly incorporated and in good standing under the laws of the State of Delaware.

Ed Map, Inc. is qualified to do business and is in good standing in the following jurisdictions: Ohio.

Schedule 5.2(b) Subsidiaries

None.

Schedule 5.3 Officers, Directors, Shareholders, Capitalization

Borrower's Shareholders and Equity Interest: See Appendix I

Pledged Securities: None.

Ed Map, Inc. Officers: Dr. Michael Mark - Chief Executive Officer

Kerry Pigman - President and Chief Operating Officer

Gregory P. Smith - Chief Financial Officer

Ed Map, Inc. Board Members: Dr. Michael Mark

Kerry Stoessel Pigman Lynn Gellerman David Kirkpatrick Jim Dugan Carol Clark

COLUMBUS/1825341v.4

Schedule 5.9(b) Litigation

None.

Schedule 5.10 Patents, Trademarks, Copyrights and Licenses

Trademarks:

Company or Affiliate	<u>Trademark</u>	Country	Application No. and/or Registration No. (as applicable)	Application Filing Date and/or Registration Date (as applicable)
Ed Map, Inc.	CURATE by Ed Map	USA	87194569	10/6/2016
Ed Map, Inc.	ALL IN MODEL	USA	87194572	10/6/2016

Schedule 5.19 Material Business Agreements

None.

Schedule 6.8 Accounts

Fifth Third Bank Account Numbers:

- 7024704764 (operating account)
- 7163946499 (disbursement account)
- 7024704798 (depository account)

Well Fargo (lockbox) Account Number 4699L76012

Schedule 7.4 Investments

None.

Schedule 7.5 Loans

A note receivable in the amount of \$110,438.31 pursuant to a promissory note dated September 30, 2016 from Madison Media Institute.

Schedule 7.8 Indebtedness

Credit card exposure not to exceed \$200,000 with Fifth Third Bank, which will be terminated no later than sixty (60) days after the Closing Date (or such extended time as agreed to by the Lender in its sole discretion).

Appendix I

ED MAP CAP TABLE

Stockholder	Cert No.	Common Stock	Series A Preferred	Series B Preferred	TOTAL	Ownership %	Fully Diluted
Authorized Shares		3,000,000	1,000,000	267,000			
Common			***************************************				
DLSC Ltd. (M Mark)	#1	1,000,000	2		1,000,000	45.50%	40.89%
Kerny Pigman	#2 & #7 & #9	150,000			150,000	6.82%	5.13%
Series B Investor	#5 for C\$. #4 & #5 for B	10,500	İ	20,823	31,323	1.43%	1.28%
Other Investors	#3, #4, #6,#8	142,500			142,500	6.48%	5.83%
Adena Ventures, L.P.	#1 for B. #1, #2, #7 for A		309,669	39,981	349,650	15.91%	14.30%
OCA Venture Partners, L.P.	#2 for B. #3 & #5 for A		232,252	29,986	262,238	11.93%	10.72%
SJF Ventures, L.P.	#3 for B, #4 & #6 for A		232,252	29,986	262,238	11.93%	10.72%
TOTAL		1,303,000	774,173	120,777	2,197,950	100.00%	
						AT CONTROL OF THE PARTY OF THE	
Stock Options Outstanding							
Kerry Pigman		65,000	1	****	65,000		2.66%
Gregory P Smith		25,000	1		25,000		1.02%
Options Granted-Prior Plan		12,500	- 1		12,500		0.51%
Options Granted-2015 Plan (to EE)		100,000	ĺ		100,000		4.09%
Options not Granted-2015 Plan		45,000	i		45,000		1.84%
TOTAL	T	1,550,500	774,173	120,777	2,445,450		100.00%

EXHIBIT A COMPLIANCE CERTIFICATE

This Compliance Certificate (this "<u>Certificate</u>") is given by Ed Map, Inc. (the "<u>Borrower Representative</u>"), pursuant to Section [9.5] [9.6] of that certain Credit and Security Agreement by and among Ed Map, Inc. and KeyBank National Association, a national banking association, dated as of November 22, 2016 (as amended, restated, supplemented or modified from time to time, the "<u>Credit Agreement</u>"). Capitalized terms used but not defined herein shall have the meanings set forth in the Credit Agreement. The undersigned, being a Financial Officer of the Borrower Representative, hereby certifies on behalf of the Borrower Representative and not in his or her individual capacity that:

[She/He] is familiar with the terms of the Credit Agreement, has made, or caused to have been made under the undersigned's supervision, a review in reasonable detail of the transactions and condition of the Borrower during the accounting period covered by the attached financial statements.

Attached are the [audited consolidated and consolidating financial statements of the Borrowers and their Subsidiaries] **[OR]** [unaudited consolidated and consolidating financial statements of the Borrowers and their Subsidiaries]. [The audited financial statements of the Borrowers and their Subsidiaries have been prepared in accordance with GAAP applied on a basis consistent with prior practices, and in reasonable detail.] **[OR]** [The unaudited financial statements of the Borrowers and their Subsidiaries have been prepared on a basis consistent with prior practices and in reasonable detail, subject to normal or customary year-end adjustments and the absence of footnotes.]

Based on an examination sufficient to make an informed statement:

no Default or Event of Default existed at the end of the accounting period covered by the attached financial statements or exists as of the date of this Certificate;

[OR]

one or more Defaults or Events of Default exist. Attached to this Certificate is an addendum specifying each such Event of Default, its nature, when it occurred, whether it is continuing and the steps being taken by the Borrowers with respect to such Default or Event of Default.

Set forth on Exhibit A hereto are calculations of the financial covenants required pursuant to Sections [6.3, 7.6 and 7.11] of the Credit Agreement, which calculations [ARE][ARE NOT] in compliance with the terms of the Credit Agreement.

[SIGNATURE PAGE FOLLOWS]

{6369722:7} A-1

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 112 of 130

IN WITNESS WHEREOF, this	s Certificate is executed and delivered as of	
, 20		
	Ed Map, Inc., a Delaware corporation	
	By:	
	Name:	
	Title:	

{6369722:7} A-2

EXHIBIT A TO COMPLIANCE CERTIFICATE

1. Calcu	ulation o	of the Fixed Charge Coverage Ratio (Se	ction 6.3)	
	(A)	EBITDA ¹	\$; <u>less</u>
	(B)	Capital Expenditures (unfunded)	\$; <u>less</u>
	(C)	Taxes paid in cash	\$; <u>less</u>
	(D)	Dividends and distributions	\$; <u>plus</u>
	(E)	Shareholder Payments	\$;
	(F)	Total of (A) <u>minus</u> (B), (C) and (D) <u>pl</u>	<u>us</u> (E)	\$
	<u>Divide</u>	<u>d By</u>		
	(G)	Interest Expense paid in cash	\$; <u>plus</u>
	(H)	Principal payments on Indebtedness	\$	
	(I)	Total of (G) <u>plus</u> (H)		\$
	Fixed Charge Coverage Ratio (F) / (I)			to 1.00
	Required (a minimum of) Compliant?			1.10 to 1.00
				Yes / No
2. Calcu	ulation o	of Capital Expenditures (Section 7.6)		
		Capital Expenditures num Permitted (per fiscal year)		\$ \$750,000
	Comp	liant?		Yes / No
3. Calcu	ulation (of Leases (Section 7.11)		
		Lease Expenditures		\$
	Maxim	num Permitted (per fiscal year)		\$750,000
	Compl	liant?		Yes / No

{6369722:7} A-3

¹ The Borrowers shall be permitted to add back \$100,000 to EBITDA for the December 31, 2016 calculation of the Fixed Charge Coverage Ratio, and the Borrowers shall be permitted to add back \$50,000 to EBITDA for the March 31, 2017 calculation of the Fixed Charge Coverage Ratio.

EXHIBIT B

FORM OF REVOLVING NOTE

November 22, 2016

Ed Map, Inc., a Delaware corporation (the "Borrower"), promises to pay to the order of KeyBank National Association, a national banking association (the "Lender"), the aggregate unpaid principal amount of the Revolving Loans made by the Lender to the Borrower pursuant to Article 2 of the Agreement (as hereinafter defined), in immediately available funds at the main office of the Lender, together with interest on the unpaid principal amount hereof at the rates and on the dates set forth in the Agreement. The Borrower shall pay the principal of and accrued and unpaid interest on the Revolving Loans in full on the Facility Termination Date and shall make such mandatory repayments as are required to be made under the terms of Article 2 of the Agreement.

The Lender shall, and is hereby authorized to record in accordance with its usual practice, the date and amount of each Revolving Loan and the date and amount of each principal payment hereunder.

This Note is one of the Notes issued pursuant to, and is entitled to the benefits of, the Credit and Security Agreement dated as of November 22, 2016 (which, as it may be amended or modified and in effect from time to time, is herein called the "Agreement"), among the Borrower, the Lender and KeyBank National Association, as the Issuer, to which Agreement reference is hereby made for a statement of the terms and conditions governing this Note, including the terms and conditions under which this Note may be prepaid or its maturity date accelerated. This Note is secured pursuant to the Loan Documents, as more specifically described in the Agreement, and reference is made thereto for a statement of the terms and provisions thereof. Capitalized terms used herein and not otherwise defined herein are used with the meanings attributed to them in the Agreement.

Except as expressly provided in the Agreement, the Borrower expressly waives presentment, demand, protest and notice of any kind.

BORROWER HEREBY AUTHORIZES ANY ATTORNEY-AT-LAW TO APPEAR IN ANY COURT OF RECORD IN ANY COUNTY IN THE STATE OF OHIO OR ELSEWHERE WHERE BORROWER HAS A PLACE OF BUSINESS, SIGNED THIS NOTE OR CAN BE FOUND, AFTER THE LENDER DECLARES A DEFAULT AND ACCELERATES THE BALANCES DUE UNDER THE AGREEMENT, TO WAIVE THE ISSUANCE OF SERVICE OF PROCESS AND CONFESS JUDGMENT AGAINST BORROWER IN FAVOR OF THE LENDER FOR THE AMOUNTS THEN APPEARING DUE, TOGETHER WITH THE COSTS OF SUIT, AND THEREUPON TO RELEASE ALL ERRORS AND WAIVES ALL RIGHT OF APPEAL AND STAYS OF EXECUTION. BORROWER AGREES AND CONSENTS THAT THE ATTORNEY CONFESSING JUDGMENT ON BEHALF OF BORROWER HEREUNDER MAY ALSO BE COUNSEL TO THE LENDER OR ANY OF ITS AFFILIATES, WAIVES ANY CONFLICT OF INTEREST WHICH MIGHT OTHERWISE ARISE, AND CONSENTS TO THE LENDER PAYING SUCH CONFESSING ATTORNEY A LEGAL FEE OR ALLOWING SUCH ATTORNEY'S FEES TO BE PAID FROM ANY PROCEEDS OF COLLECTION OF THIS NOTE OR COLLATERAL SECURITY THEREFOR.

{6369722:7} B-1

WARNING -- BY SIGNING THIS PAPER YOU GIVE UP YOUR RIGHT TO NOTICE AND COURT TRIAL. IF YOU DO NOT PAY ON TIME A COURT JUDGMENT MAY BE TAKEN AGAINST YOU WITHOUT YOUR PRIOR KNOWLEDGE AND THE POWERS OF A COURT CAN BE USED TO COLLECT FROM YOU REGARDLESS OF ANY CLAIMS YOU MAY HAVE AGAINST THE CREDITOR WHETHER FOR RETURNED GOODS, FAULTY GOODS, FAILURE ON HIS PART TO COMPLY WITH THE AGREEMENT, OR ANY OTHER CAUSE.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered as of the date first set forth.

Ed Map, Inc.,

a Delaware corpo	ration	
By:		
Name:		
Title:	•	

{6369722:7} B-2

EXHIBIT C

FORM OF NOTICE OF LOAN

, 20
KeyBank National Association
127 Public Square
Cleveland, Ohio 44114-0616 Attention: Asset Based Lending
Attention. Asset based Lending
Ladies and Gentlemen:
The undersigned,, on behalf of Ed Map, Inc. ("Borrower Representative") refers to the Credit and Security Agreement, dated as of November 22, 2016 ("Credit Agreement", the terms defined therein being used herein as therein defined), among the Borrower Representative, the other Loan Parties, the Issuer, and KeyBank National Association, and hereby gives you notice, pursuant to Section 2.2 of the Credit Agreement that the Borrower Representative, on behalf of the Borrowers, hereby requests a Loan under the Credit Agreement, and in connection therewith sets forth below the information relating to the Loan (the "Proposed Loan") as required by Section 2.2 the Credit Agreement:
(a) The Business Day of the Proposed Loan is, 20
(b) The amount of the Proposed Loan is \$
(c) The Proposed Loan is to be a Base Rate Loan / Libor Rate Loan (Check one).
(d) If the Proposed Loan is a Libor Rate Loan, the Interest Period requested is one month, two months, three months, six months (Check one).
The undersigned hereby certifies that the following statements are true on the date hereof, and will be true on the date of the Proposed Loan:
(i) the representations and warranties contained in each Loan Document are correct before and after giving effect to the Proposed Loan and the application of the proceeds therefrom as though made on and as of such date;
(ii) no event has occurred and is continuing, or would result from such Proposed Loan, or the application of proceeds therefrom, that constitutes a Default or Event of Default; and
(iii) the conditions set forth in Section 2.2 and Article 8 of the Credit Agreement have been satisfied.
Ed Map, Inc.,
a Delaware corporation
By:
Name:
Title:

{6369722:7} C-1

FIRST AMENDMENT TO CREDIT AND SECURITY AGREEMENT

This FIRST AMENDMENT TO CREDIT AND SECURITY AGREEMENT (this "Amendment"), dated as of September 25, 2017, is by and between **ED MAP, INC.**, a Delaware corporation ("Borrower"), and **KEYBANK NATIONAL ASSOCIATION**, a national banking association ("Lender").

BACKGROUND

- A. Borrower and Lender entered into that certain Credit and Security Agreement dated as of November 22, 2016 (as may be amended, modified, extended, or restated from time to time, the "Agreement"), pursuant to which Lender extended certain financing accommodations to Borrower.
- B. The parties hereto have agreed to modify the terms and conditions of the Agreement as more fully set forth herein.
- C. Capitalized terms used but not otherwise defined herein shall have the meanings given to such terms in the Agreement, as amended by this Amendment.

NOW THEREFORE, in consideration of the terms, conditions and covenants set forth below, and other good and valuable consideration, the receipt of which is hereby acknowledged, the parties, intending to be legally bound hereby, promise and agree as follows:

- 1. <u>Amendment to Section 1.2 (General Terms) New Definitions.</u> Section 1.2 of the Agreement is hereby amended by adding the following new definition thereto in proper alphabetical order:
 - "First Amendment Effective Date" shall mean September 25, 2017.
- 2. <u>Amendment to Section 1.2 (General Terms) Existing Definitions.</u> Section 1.2 of the Agreement is hereby amended by deleting the following definitions contained therein in their entirety and replacing them with the following:
 - "Applicable Base Rate Margin" shall mean 2.00%.
 - "Applicable Letter of Credit Fee Percentage" shall mean 4.00%.
 - "Applicable LIBOR Rate Margin" shall mean 4.00%.
 - "Borrowing Base" shall mean, at any time, the sum of:
 - (a) up to 85% of each Borrower's Eligible Accounts at such time, *plus*
 - (b) the least of (i) up to 50% of each Borrower's Eligible Inventory, (ii) the product of 85% *multiplied by* the Net Orderly Liquidation Value *multiplied by* each Borrower's Eligible Inventory, in each case, valued at the lower of cost or market, determined on a first in first out basis, at such time, and (iii) \$3,500,000, *minus*

- (c) the Availability Block, minus
- (d) Reserves.

The Lender may, in its Permitted Discretion, reduce the advance rates set forth above, adjust Reserves or reduce one or more of the other elements used in computing the Borrowing Base; provided, however, the Lender shall endeavor to provide notice to the Borrowers of any such Reserve or reduction, but the failure to give such notice shall not affect its ability to implement any such Reserve or reduction.

"Facility Termination Date" shall mean November 22, 2018.

"<u>Permitted Acquisition</u>" shall mean any Acquisition by any Borrower, to the extent that each of the following conditions shall have been satisfied:

- (a) the assets, business or Person being acquired is engaged in the same business of the Borrowers and their Subsidiaries;
- (b) (i) in the case of a stock acquisition, the target thereof shall be a Domestic Subsidiary of a Borrower and (ii) in the case of an asset acquisition, the assets being acquired shall be located in the United States;
- (c) the Borrower Representative shall have delivered to the Lender (i) as soon as available (but no later than five (5) Business Days after the consummation of such acquisition), final executed counterparts of the material agreements, documents and instruments pursuant to which such acquisition is to be consummated (including any management, non-compete, employment and option agreements) and any schedules to such agreements, documents and instruments, (ii) to the extent required under the related acquisition agreement, all consents and approvals from applicable governmental authorities and other Persons required to consummate such acquisition and (iii) if reasonably requested by the Lender, environmental assessments reasonably satisfactory to the Lender;
- (d) the Borrowers and their Subsidiaries (including any new Subsidiary) shall execute and deliver the agreements, instruments and other documents required by Sections 4.2, 4.20, and 7.12, and upon the consummation of such acquisition to the extent required by Section 7.12, the target thereof shall be a Borrower hereunder;
- (e) there shall not exist on the date on which the acquisition is consummated, both before and after giving effect to the acquisition, a Default or Event of Default;
- (f) such acquired assets and/or Equity Interests shall be acquired free and clear of all Liens (except for Permitted Liens);
- (g) if the Accounts and Inventory acquired in connection with such acquisition are proposed to be included in the determination of the Borrowing Base, the Lender shall have conducted an audit and field examination of such

Accounts and Inventory, the results of which shall be reasonably satisfactory to the Lender;

- (h) the total consideration paid or payable (including all transaction costs and all Indebtedness incurred, assumed and/or reflected on a consolidated balance sheet of the Borrowers and their Subsidiaries after giving effect to such Acquisition) shall not exceed for all Acquisitions consummated during the term of this Agreement, \$1,250,000 <u>less</u> the amount paid for Permitted Stock Repurchases during the term of this Agreement;
- (i) as soon as available, but not less than five (5) days prior to such acquisition, the Borrower Representative shall have provided the Lender (i) notice of such acquisition and (ii) a copy of all business and financial information reasonably requested by the Lender including pro forma financial statements, statements of cash flow, and availability projections;
- (j) the Fixed Charge Coverage Ratio shall equal or exceed 1.10 to 1.00 (i) as of the last day of the fiscal month immediately preceding such Acquisition (calculated for the twelve (12) consecutive fiscal months ending on such calculation date) and (ii) as of the last date of each of the twelve (12) fiscal months following such Acquisition, in each case, on a pro forma basis after giving effect to such Acquisition as if it had been made during the prior fiscal month; and
- (k) Excess Availability on the proposed date of such Acquisition (after giving effect to such Acquisition) and on a pro-forma basis for each of the thirty (30) days following such Acquisition shall exceed \$1,250,000.
- "<u>Permitted Stock Repurchase</u>" shall mean repurchases or redemptions by the Borrower Representative of its preferred stock, to the extent that each of the following conditions shall have been satisfied:
- (a) there shall not exist on the date on which the repurchase or redemption is consummated, both before and after giving effect to the repurchase or redemption, a Default or Event of Default;
- (b) the total consideration paid or payable (including all transaction costs) shall not exceed for all such repurchases and redemptions consummated during the term of this Agreement, \$1,250,000 <u>less</u> the amount paid for Permitted Acquisitions during the term of this Agreement;
- (c) as soon as possible, but not less than five (5) days prior to such repurchase or redemption, the Borrower Representative shall have provided the Lender with notice of the same;
- (d) the Fixed Charge Coverage Ratio shall equal or exceed 1.10 to 1.00 (i) as of the last day of the fiscal month immediately preceding such repurchase or redemption (calculated for the twelve (12) consecutive fiscal months ending on such calculation date) and (ii) as of the last date of each of the twelve (12) fiscal months following such repurchase or redemption, in

each case, on a pro forma basis after giving effect to such repurchase or redemption as if it had been made during the prior fiscal month; and

(e) Excess Availability on the proposed date of such repurchase or redemption (after giving effect to such repurchase or redemption) and on a pro-forma basis for each of the thirty (30) days following such repurchase or redemption shall exceed \$1,250,000.

"Revolving Commitment" shall mean the commitment of the Lender to make Revolving Loans and issue Letters of Credit, as such commitment may be reduced pursuant to the terms of this Agreement. As of the First Amendment Effective Date, the amount of the Lender's Revolving Commitment is \$8,000,000.

3. <u>Amendment to Section 6.3 (Fixed Charge Coverage Ratio)</u>. Section 6.3 of the Agreement is hereby amended by deleting such section in its entirety and replacing it with the following:

6.3 Financial Covenants.

- (a) <u>Fixed Charge Coverage Ratio.</u> Maintain a Fixed Charge Coverage Ratio (for the Loan Parties on a consolidated basis) of not less than (i) 2.25 to 1.00 calculated as of the last day of the fiscal quarter ending on June 30, 2018, for the period equal to three (3) consecutive fiscal quarters then ending, and (ii) 1.10 to 1.00 calculated as of the last day of the fiscal quarter ending September 30, 2018, for the period equal to four (4) consecutive fiscal quarters then ending and as of the last day of each fiscal quarter thereafter for the period equal to the four (4) consecutive fiscal quarters then ending.
- (b) Minimum EBITDA. Maintain a minimum EBITDA, calculated using the financial results reported in accordance with GAAP and applied on a basis consistent with the financial statements delivered to the Lender prior to the date hereof, of not less than (i) (\$245,000) calculated as of the last day of the fiscal quarter ending September 30, 2017, for the period equal to one (1) consecutive fiscal quarter then ending, (ii) (\$330,000) calculated as of the last day of the fiscal quarter ending December 31, 2017, for the period equal to two (2) consecutive fiscal quarters then ending, and (iii) (\$295,000) calculated as of the last day of the fiscal quarter ending March 31, 2018, for the period equal to three (3) consecutive fiscal quarters then ending.
- (c) <u>Minimum Excess Availability</u>. Commencing on the First Amendment Effective Date and at all times thereafter, in each case, calculated on a trailing three (3) Business Day average, maintain Excess Availability of greater than \$1,250,000.
- 4. <u>Amendment to Section 7.6 (Capital Expenditures).</u> Section 7.6 of the Agreement is hereby amended by deleting such section in its entirety and replacing it with the following:
 - 7.6 <u>Capital Expenditures</u>. Make or incur any Capital Expenditure or commitments for Capital Expenditures (including capitalized leases) in an aggregate amount for the Loan Parties on a consolidated basis in excess of (a) \$130,000, calculated as of the last day of the fiscal quarter ending September 30, 2017, for the

period equal to one (1) consecutive fiscal quarter then ending, (b) \$180,000, calculated as of the last day of the fiscal quarter ending December 31, 2017, for the period equal to two (2) consecutive fiscal quarters then ending, (c) \$220,000, calculated as of the last day of the fiscal quarter ending March 31, 2018, for the period equal to three (3) consecutive fiscal quarters then ending, (d) \$285,000, calculated as of the last day of the fiscal quarter ending June 30, 2018, for the period equal to four (4) consecutive fiscal quarters then ending, and (e) \$350,000, calculated as of the last day of the fiscal year ending June 30, 2019, for the period equal to four consecutive fiscal quarters then ending and as of the last day of each fiscal year thereafter for the period equal to the four (4) consecutive fiscal quarters then ending.

- 5. Default Waiver. Borrower acknowledges and agrees that, as of the date hereof, Borrower is in default of Section 6.3(a) (Fixed Charge Coverage Ratio) with respect to Borrower's failure to maintain the required Fixed Charge Coverage Ratio for the period ending March 31, 2017 (the "Existing Default"). Upon the effectiveness of this Amendment and in specific reliance on the covenants, representations and warranties made herein, Lender hereby waives the Existing Default; provided that such waiver shall be conditioned on and subject to Borrower satisfying the conditions to effectiveness contained in this Amendment. Further, such waiver shall in no way constitute a waiver of any other Event of Default which may have occurred but which is not specifically referenced as an "Existing Default" nor shall it obligate Lender to provide any further waiver of any other Event of Default. Except as otherwise waived hereunder, Lender reserves the right to invoke fully any and all of its respective rights, remedies, powers and privileges under the Agreement and all other Loan Documents at any time in the exercise of Lender's sole and absolute discretion. This waiver is a onetime waiver related specifically to the Existing Default and shall not constitute a course of conduct of imply a future waiver of any provision of the Agreement.
- 6. <u>Fees and Expenses</u>. To induce Lender to enter into this Amendment, Borrower hereby agrees to pay Lender an amendment fee in the amount of \$10,000 (the "<u>Amendment Fee</u>"), which shall be payable and earned in full upon the execution and delivery of this Amendment. In addition to the Amendment Fee, Borrower shall reimburse Lender for all out-of-pocket costs and expenses incurred in connection with this Amendment, including, without limitation, reasonable attorneys' fees.
- 7. <u>Conditions Precedent.</u> In addition to all of the other conditions and agreements set forth herein, the effectiveness of this Amendment is subject to the following conditions precedent:
 - (a) Lender shall have received this Amendment executed and delivered by all the parties hereto;
 - (b) Borrower shall have paid any and all out-of-pocket costs, fees and expenses of Lender (including the Amendment Fee and reasonable attorney fees) in connection with this Amendment; and
 - (c) Lender shall have received such other and further documentation as Lender may reasonably deem necessary or appropriate to accomplish the terms set forth herein, each in form and substance reasonably satisfactory to Lender.
- 8. <u>Representations and Warranties</u>. Borrower hereby represents and warrants to Lender that (a) Borrower has the legal power and authority to execute and deliver this Amendment, (b) the officials executing this Amendment have been duly authorized to execute and deliver the same and bind Borrower with respect to the provisions hereof, (c) the execution and delivery hereof by Borrower and

the performance and observance by Borrower of the provisions hereof do not violate or conflict with the organizational agreements of Borrower or any law applicable to Borrower or result in a breach of any provisions of or constitute a default under any Material Business Agreement or any other agreement, instrument or document binding upon or enforceable against Borrower, (d) this Amendment constitutes a valid and binding obligation upon Borrower in every respect except as limited by bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium or similar federal or state laws or judicial decisions relating to the rights of creditors, (e) no event or condition which has or could reasonably be expected to have a Material Adverse Effect as to Borrower has occurred from the Closing Date to the date hereof, and (f) no Default or Event of Default is outstanding under the Agreement.

- 9. Governing Law; Use of Terms Etc. Except as previously amended or as herein specifically amended, directly or by reference, all of the terms and conditions set forth in the Agreement are confirmed and ratified, and shall remain as originally written. This Amendment shall be construed in accordance with the laws of the State of Ohio, without regard to principles of conflict of laws. The Agreement and all other Loan Documents shall remain in full force and effect in all respects as if the unpaid balance of the principal outstanding, together with interest accrued thereon, had originally been payable and secured as provided for therein, as amended from time to time and as modified by this Amendment. Nothing herein shall affect or impair any rights and powers which Lender may have under the Agreement and any and all related Loan Documents.
- 10. No Set Offs Etc. Borrower hereby declares that Borrower has no set offs, counterclaims, defenses or other causes of action against Lender arising out of the Agreement, any Loan Document or any related documents, and to the extent any such set offs, counterclaims, defenses or other causes of action may exist, whether known or unknown, such items are hereby waived by Borrower.
- 11. <u>Confirmation of Security Interests</u>. Borrower confirms and agrees that all prior security interests and liens granted to Lender in all existing and future assets of Borrower remain unimpaired and in full force and effect and shall continue to cover and secure all Obligations. Borrower further confirms and represents that all of the collateral of Borrower remains free and clear of all liens other than those in favor of Lender or as otherwise permitted in the Agreement. Nothing contained herein is intended to in any way impair or limit the validity, priority or extent of Lender's security interest in and liens upon the collateral of Borrower.
- 12. <u>Obligations Absolute</u>. Borrower covenants and agrees (a) to pay the balance of any principal, together with all accrued interest, as specified above in connection with any promissory note executed and evidencing any indebtedness incurred in connection with the Agreement, as modified by this Amendment pursuant to the terms set forth therein, and (b) to perform and observe covenants, agreements, stipulations and conditions on its part to be performed hereunder or under the Agreement and all other documents executed in connection herewith or thereof.
- 13. BORROWER AND LOAN GUARANTOR EACH HEREBY RELEASE, Release. WAIVE AND FOREVER RELINQUISH ALL CLAIMS, DEMANDS, OBLIGATIONS, LIABILITIES AND CAUSES OF ACTION OF WHATEVER KIND OR NATURE, WHETHER KNOWN OR UNKNOWN, INCLUDING ANY SO-CALLED "LENDER LIABILITY" CLAIMS OR DEFENSES WHICH IT HAS, MAY HAVE, OR MIGHT ASSERT NOW OR IN THE FUTURE AGAINST LENDER AND/OR ITS RESPECTIVE OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, ATTORNEYS, ACCOUNTANTS, CONSULTANTS, SUCCESSORS, AND ASSIGNS (INDIVIDUALLY, EACH A "RELEASEE" AND COLLECTIVELY, THE "RELEASEES"), DIRECTLY OR INDIRECTLY, ARISING OUT OF, BASED UPON, OR IN ANY MANNER CONNECTED WITH (A) ANY TRANSACTION, EVENT, CIRCUMSTANCE, ACTION, FAILURE

TO ACT, OR OCCURRENCE OF ANY SORT OR TYPE, WHETHER KNOWN OR UNKNOWN, WHICH OCCURRED, EXISTED, OR WAS TAKEN OR PERMITTED PRIOR TO THE EXECUTION OF THIS AMENDMENT WITH RESPECT TO THE OBLIGATIONS, THE AGREEMENT, THE OTHER DOCUMENTS, OR THE ADMINISTRATION THEREOF, (B) ANY DISCUSSIONS, COMMITMENTS, NEGOTIATIONS, CONVERSATIONS, OR COMMUNICATIONS WITH RESPECT TO THE OBLIGATIONS OR (C) ANY THING OR MATTER RELATED TO ANY OF THE FOREGOING PRIOR TO THE EXECUTION OF THIS AMENDMENT. THE INCLUSION OF THIS PARAGRAPH IN THIS AMENDMENT AND THE EXECUTION OF THIS AMENDMENT BY LENDER DOES NOT CONSTITUTE AN ACKNOWLEDGMENT OR ADMISSION BY LENDER OF LIABILITY FOR ANY MATTER, OR A PRECEDENT UPON WHICH ANY LIABILITY MAY BE ASSERTED.

- 14. <u>Non-Waiver</u>. This Amendment does not obligate Lender to agree to any other modification of the Agreement nor does it constitute a course of conduct or dealing on behalf of Lender or a waiver of any other rights or remedies of Lender. No omission or delay by Lender in exercising any right or power under the Agreement, this Amendment or any related instruments, agreements or documents will impair such right or power or be construed to be a waiver of any Default or Event of Default or an acquiescence therein, and any single or partial exercise of any such right or power will not preclude other or further exercise thereof or the exercise of any other right, and no waiver will be valid unless in writing and then only to the extent specified.
- 15. <u>Incorporation</u>. This Amendment is incorporated by reference into, and made part of, the Agreement which, except as expressly modified herein, remains in full force and effect in accordance with its terms.
- 16. <u>No Modification</u>. No modification of this Amendment or of any agreement referred to herein shall be binding or enforceable unless in writing and signed on behalf of the party against whom enforcement is sought.
- 17. <u>Headings</u>. The headings of any section or paragraph of this Amendment are for convenience only and shall not be used to interpret any provision of this Amendment.
- 18. <u>Successors and Assigns</u>. This Amendment will be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.
- 19. <u>Severability</u>. The provisions of this Amendment are to be deemed severable, and the invalidity or unenforceability of any provision shall not affect or impair the remaining provisions which shall continue in full force and effect.
- 20. <u>Counterparts, Electronic Signature</u>. This Amendment may be executed in any number of counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement. Signature by facsimile, email or other electronic method shall have the same force and effect as an original signature hereto.
- 21. <u>Jury Waiver</u>. THE PARTIES HERETO HEREBY WAIVE ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING RELATING TO THIS AMENDMENT, ANY OF THE LOAN DOCUMENTS, ANY DOCUMENT DELIVERED HEREUNDER OR IN CONNECTION HEREWITH, OR ANY TRANSACTION ARISING FROM OR CONNECTED TO ANY OF THE FOREGOING. THE PARTIES REPRESENT THAT THIS WAIVER IS KNOWINGLY, WILLINGLY AND VOLUNTARILY GIVEN.

22. CONFESSION OF JUDGMENT. BORROWER HEREBY AUTHORIZES ANY ATTORNEY-AT-LAW TO APPEAR IN ANY COURT OF RECORD IN ANY COUNTY IN THE STATE OF OHIO OR ELSEWHERE WHERE BORROWER HAS A PLACE OF BUSINESS, SIGNED THIS AGREEMENT OR CAN BE FOUND, AFTER LENDER DECLARES A DEFAULT AND ACCELERATES THE BALANCES DUE UNDER THIS AGREEMENT, TO WAIVE THE ISSUANCE OF SERVICE OF PROCESS AND CONFESS JUDGMENT AGAINST BORROWER IN FAVOR OF LENDER FOR THE AMOUNTS THEN APPEARING DUE, TOGETHER WITH THE COSTS OF SUIT, AND THEREUPON TO RELEASE ALL ERRORS AND WAIVE ALL RIGHT OF APPEAL AND STAY OF EXECUTION. BORROWER AGREES AND CONSENTS THAT THE ATTORNEY CONFESSING JUDGMENT ON BEHALF OF BORROWER HEREUNDER MAY ALSO BE COUNSEL TO LENDER OR ANY OF ITS AFFILIATES, WAIVES ANY CONFLICT OF INTEREST WHICH MIGHT OTHERWISE ARISE, AND CONSENTS TO LENDER PAYING SUCH CONFESSING ATTORNEY A LEGAL FEE OR ALLOWING SUCH ATTORNEY'S FEES TO BE PAID FROM ANY PROCEEDS OF COLLECTION OF THIS AGREEMENT OR COLLATERAL SECURITY THEREFOR.

[Signature page follows]

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 125 of 130

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered on the date first set forth.

BORROWER:

WARNING -- BY SIGNING THIS PAPER YOU GIVE UP YOUR RIGHT TO NOTICE AND COURT TRIAL. IF YOU DO NOT PAY ON TIME A COURT JUDGMENT MAY BE TAKEN AGAINST YOU WITHOUT YOUR PRIOR KNOWLEDGE AND THE POWERS OF A COURT CAN BE USED TO COLLECT FROM YOU REGARDLESS OF ANY CLAIMS YOU MAY HAVE AGAINST THE CREDITOR WHETHER FOR RETURNED GOODS, FAULTY GOODS, FAILURE ON HIS PART TO COMPLY WITH THE AGREEMENT, OR ANY OTHER CAUSE.

ED MAP, INC.,

a Delaware corporation

By: __

Name: Gregory P. Smith Title: Chief Financial Officer

LENDER and ISSUER:

KEYBANK NATIONAL ASSOCIATION,

a national banking association

Bv:

Name: Andrew Blickensderfer

Title: Vice President

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 126 of 130

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered on the date first set forth.

BORROWER:

WARNING -- BY SIGNING THIS PAPER YOU GIVE UP YOUR RIGHT TO NOTICE AND COURT TRIAL. IF YOU DO NOT PAY ON TIME A COURT JUDGMENT MAY BE TAKEN AGAINST YOU WITHOUT YOUR PRIOR KNOWLEDGE AND THE POWERS OF A COURT CAN BE USED TO COLLECT FROM YOU REGARDLESS OF ANY CLAIMS YOU MAY HAVE AGAINST THE CREDITOR WHETHER FOR RETURNED GOODS, FAULTY GOODS, FAILURE ON HIS PART TO COMPLY WITH THE AGREEMENT, OR ANY OTHER CAUSE.

ED MAP, INC.,

a Delaware corporation

Name: Gregory P. Smith
Title: Chief Financial Officer

LENDER and ISSUER:

KEYBANK NATIONAL ASSOCIATION,

a national banking association

Name: Andrew Blickensderfer

Title: Vice President

	Coco 2:19 bk 55990 Doc 18 Filed (Documer	09/20/18 nt Page	Entered 09/20/ e 127 of 130	18 15:0)7:23 Desc N	Main
11/	CO FINANCING STATEMENT					
	CC FINANCING STATEMENT LLOW INSTRUCTIONS					
A.	NAME & PHONE OF CONTACT AT FILER (optional)					
В.	E-MAIL CONTACT AT FILER (optional)					
C.	SEND ACKNOWLEDGMENT TO: (Name and Address) ——	_				
		I				
۱		I				
					R FILING OFFICE USE	
	DEBTOR'S NAME: Provide only <u>one</u> Debtor name (1a or 1b) (use exact, full in name will not fit in line 1b, leave all of item 1 blank, check here and provide to		modify, or abbreviate any part or information in item 10 of the			
	1a. ORGANIZATION'S NAME					
OR	1b. INDIVIDUAL'S SURNAME	FIRST PERSONA	L NAME	ADDITIO	NAL NAME(S)/INITIAL(S)	SUFFIX
				ADDITIONAL NAIVIE(3)/IINTTIAL(5)		
1c.	MAILING ADDRESS	CITY		STATE	POSTAL CODE	COUNTRY
2 [DEBTOR'S NAME: Provide only one Debtor name (2a or 2b) (use exact, full r	name: do not omit	modify or abbreviate any part	of the Debtor'	s name): if any part of the	Individual Debtor's
			or information in item 10 of the			
	2a. ORGANIZATION'S NAME					
OR	2b. INDIVIDUAL'S SURNAME	FIRST PERSONA	L NAME	ADDITIONAL NAME(S)/INITIAL(S) SUF		
2c.	MAILING ADDRESS	CITY		STATE	POSTAL CODE	COUNTRY
3. 5	SECURED PARTY'S NAME (or NAME of ASSIGNEE of ASSIGNOR SECU	RED PARTY): Pro	vide only one Secured Party n	ame (3a or 3b)	
	3a. ORGANIZATION'S NAME	·	,,	,		
OR	3b. INDIVIDUAL'S SURNAME	FIRST PERSONA	L NAME	ADDITIO	NAL NAME(S)/INITIAL(S)	SUFFIX
3c.	MAILING ADDRESS	CITY		STATE	POSTAL CODE	COUNTRY
4. (COLLATERAL: This financing statement covers the following collateral:					
		(see UCC1Ad, item			red by a Decedent's Perso	
od.	Check only if applicable and check only one box: Public-Finance Transaction Manufactured-Home Transaction	A Debtor is	a Transmitting Utility		f applicable and check <u>only</u> ural Lien Non-UC	
7. <i>P</i>		Consignee/Consign				ensee/Licensor
8. 0	DPTIONAL FILER REFERENCE DATA:					

Exhibit A to UCC Financing Statement

"Excluded Asset" shall mean any of the following:

- (a) a leasehold mortgage on Debtor's leased location located at 294/296 Harper Street, Nelsonville, Ohio 45764;
- (b) depository accounts and security accounts maintained solely for one or more of the following: payroll, medical expenses and payments, pension benefits, employee wages, withholding taxes, social security taxes; sales taxes, or accounts otherwise held exclusively for the benefit of third parties, other than the Debtor or its Affiliates;
- (c) any intent-to-use trademark application to the extent and for so long as creation by Debtor of a Lien therein would result in the loss by such Person of any material rights therein;
- (d) Debtor's rights or interests in or under any license, contract, permit, instrument, security or franchise or any of its rights thereunder, in each case to the extent such a grant would, under the terms of such contract, result in a breach of, or constitute a default under, such contract; provided that rights to payment and other rights under any such contract shall be included in the Collateral to the extent permitted by, or to the extent such terms are deemed ineffective by, Section 9-406, 9-407, 9-408 or 9-409 of the UCC, and provided, however, such property shall be included in Collateral at such time as the restriction terminates; and
- (e) Investment Property to the extent a security interest in such assets would result in adverse tax consequences (including as a result of the operation of Section 956 of the IRC or any similar law or regulation in any applicable jurisdiction) to the Debtor or any Subsidiary as reasonably determined in good faith by the Debtor in writing delivered to Secured Party;

provided, however, that "Excluded Assets" shall not include any Proceeds, products, substitutions or replacements of any Excluded Asset (unless such Proceeds, products, substitutions or replacements would themselves constitute an Excluded Asset under paragraphs (a) through (d); and provided further that any such property or asset that at any time ceases to satisfy the criteria for an Excluded Asset and constitutes Collateral (whether as a result of the Debtor obtaining any necessary consent, any change in applicable law, or otherwise), shall no longer be an Excluded Asset.

CERTIFICATE OF SERVICE

I hereby certify that a copy of the foregoing MOTION OF DEBTOR FOR INTERIM AND FINAL ORDERS AUTHORIZING USE OF CASH COLLATERAL AND PROVIDING ADEQUATE PROTECTION TO KEY BANK was served (i) **electronically** on the date of filing through the court's ECF system on all ECF participants registered in the case at the email addresses registered with the court, and (ii) by **ordinary U.S. Mail** on September 20, 2018 addressed to the Debtor, the Office of the United States Trustee, the twenty largest unsecured creditors, all secured creditors and those other parties set forth on the attached mailing matrix.

/s/ John W. Kennedy John W. Kennedy (0042672)

Case 2:18-bk-55889 Doc 18 Filed 09/20/18 Entered 09/20/18 15:07:23 Desc Main Document Page 130 of 130

ED MAP INC 296 HARPER ST NELSONVILLE OH 45764 MARYANNE WILSBACHER OFFICE OF THE U S TRUSTEE 170 N HIGH ST #200 COLUMBUS OH 4315 NATASSIA TRAVIS AHIMA 233 N MICHIGAN AVE 21ST FL CHICAGO IL 60601

LINDA WROBEL AMERICAN MEDICAL ASSOC 330 N WABASH AVE #39300 CHICAGO IL 60601 DEANNA WEINEL CENGAGE LEARNING 10650 TOEBBEN DR INDEPENDENCE KY 41051 ACCOUNTING DEPT DAWN SIGN PRESS 6130 NANY RIDGE DR SAN DIEGO CA 92121

DALE VIGEN EMC PARADIGM PUBLISHING 9004 SOLUTION CENTER CHICAGO IL 60677-9000 SARA HICKMAN ESCIENCE LABS 1500 W HAMPDEN AVE BLDG 2 SHERIDAN CO 80110 KURT KELLER ELVSEVIER SCIENCE 321 RIVERPORT LN MARYLAND HEIGHTS MO 63043

ELLEN PORTER JOHN WILEY & SONS ONE WILEY DR SOMERSET NJ 08875 SUE JONES KENDALL-HUNT PUBLISHING 405 WESTMARK DR DUBUQUE IA 52004 RICK SCHANHALS MEDTRAK 1847 RUDDIMAN DR NORTH MUSKEGON MI 49445

JUANITA ALESSI MPS (FORMERLY VHPS) 16365 JAMES MADISON HWY GORDONSVILLE VA 22942 STEPHANIE ANDERSON MCGRAW HILL P O BOX 2258 CAROL STREAM IL 60132 SHARON CARTER OXFORD UNIVERSITY PRESS 2001 EVANS RD CARY NC 27513

ANNE TORTELLI PEARSON EDUCATION 20 OLD TAPPAN RD WESTWOOD NJ 07675

STEPHANIE PALKA SHADOW HEALTH INC 201 SE 2ND AVE GAINESVILLE FL 32601-6239 JONATHON NEWCOMB TRAZ CAPITAL PARTNERS LLC 60 RIVERSIDE DR NEW YORK NY 10024

KRISTA HOWELL TAYLOR & FRANCIS 5000 BROKEN SOUND PKWY NW #300 BOCA RATON FL 33487 TRACI BIRD TEST OUT CORPORATION 50 SOUTH MAIN ST PLEASANT GROVE UT 84062 KERI CABLE VITALSOURCE ATTN LEGAL DEPARTMENT 227 FAYETTEVILLE ST #400 RALEIGH NC 27601

KEY BANK NATIONAL ASSOC ATTN LEGAL DEPARTMENT 127 PUBLIC SQUARE CLEVELAND OH 44114 ROBERT J BURNS ESQ KEYBANK NATIONAL ASSOC 127 PUBLIC SQUARE 2ND FLOOR CLEVELAND OH 44114-1306 PAMELA ARNDT OFFICE OF THE U S TRUSTEE 170 N HIGH ST #200 COLUMBUS OH 43215

PHILLIP K STOVALL ALLEN STOVALL NEUMAN FISHER & ASHTON LLP 17 S HIGH ST #1220 COLUMBUS OH 43215 RICHARD K STOVALL ALLEN STOVALL NEUMAN FISHER & ASHTON LLP 17 S HIGH ST #1220 COLUMBUS OH 43215 THOMAS R ALLEN ALLEN STOVALL NEUMAN FISHER & ASHTON LLP 17 S HIGH ST #1220 COLUMBUS OH 43215